

Title	Low-Voltage and Small-Area Design and Implementation of Narrowband and Wideband CMOS Low-Noise Amplifiers
Author(s)	木原, 崇雄
Citation	大阪大学, 2009, 博士論文
Version Type	VoR
URL	https://hdl.handle.net/11094/2408
rights	
Note	

Osaka University Knowledge Archive : OUKA

https://ir.library.osaka-u.ac.jp/

Osaka University

Low-Voltage and Small-Area Design and Implementation of Narrowband and Wideband CMOS Low-Noise Amplifiers (狭帯域・広帯域CMOS低雑音増幅器の 低電圧化および小面積化に関する研究)

> Takao Kihara 木原 崇雄

Division of Electrical, Electronic and Information Engineering, Graduate School of Engineering, Osaka University

February 2009

Abstract

This dissertation addressed low-voltage and small-area design and implementation of narrowband and wideband CMOS low-noise amplifiers (LNAs). It was organized into five chapters and two appendixes. The summaries of each chapter were as follows:

Chapter 1

A background for this work and fundamentals of LNAs were described.

Chapter 2

A 1.0 V, 5 GHz two-stage CMOS LNA with inductive source degeneration was demonstrated. Its design methodology based on analytical expressions was also presented. The two-stage topology consisting of common-source and common-gate stages was more suitable for low-voltage operation than a conventional cascode topology. The complete analytical expressions of the LNA performance were first derived from the small-signal equivalent circuits. The LNA fabricated in a 0.15 μ m fully-depleted silicon-on-insulator (FD-SOI) CMOS process occupied 0.25 mm² and achieved an S_{11} of less than -10 dB, NF of 1.7 dB, voltage gain of 23 dB, and IIP_3 of -6.1 dBm at 5.4 GHz with a power consumption of 8.3 mW. These measurements were consistent with calculations obtained from the derived analytical expressions.

Chapter 3

A 0.5 V, 5 GHz transformer folded-cascode CMOS LNA was demonstrated. The chip area of a conventional folded-cascode LNA was reduced by partially coupling the internal inductor with the load inductor. The effects of the magnetic coupling between these inductors on the LNA performance were also analyzed. The LNA fabricated in a 90 nm digital CMOS process achieved an S_{11} of -14 dB, NF of 3.9 dB, and voltage gain of 16.8 dB at 4.7 GHz with a power consumption of 1.0 mW. The chip area of the presented LNA was 25% (0.21 mm²) smaller than that of the conventional folded-cascode LNA (0.29 mm²).

Chapter 4

A 1.0 V, 3.1–10.6 GHz transformer noise-canceling CMOS LNA based on a common-gate topology was demonstrated. The transformer consisting of the input and shunt-peaking inductors partly canceled the noise originating from the common-gate transistor and load resistor. The combination of the transformer with the output series inductor provided wideband input impedance matching. The LNA designed for ultra-wideband (UWB) applications was fabricated in a 90 nm digital CMOS process. It achieved an S_{11} of less than -10 dB, NF of less than 4.4 dB, and S_{21} of more than 9.3 dB with a power consumption of 2.5 mW and occupied the smallest chip area (0.12 mm²) among previously reported 3.1–10.6 GHz CMOS LNAs.

Chapter 5

The achievements obtained in this work were summarized and this dissertation was concluded.

Acknowledgments

First of all, I would like to express my gratitude to my supervisor, Professor Kenji Taniguchi (Division of Electrical, Electronic and Information Engineering, Graduate School of Engineering, Osaka University). His appropriate and critical advices always lead me to many successful research.

I am very grateful to Professor Tetsuya Yagi (Division of Electrical, Electronic and Information Engineering, Graduate School of Engineering, Osaka University) and Associate Professor Masanori Hashimoto (Department of Information Systems Engineering, Graduate School of Information Science and Technology, Osaka University) for their advices and suggestions on writing this dissertation.

I would like to thank Professor Toshimichi Ito, Professor Masanori Ozaki, Professor Mitsuhiro Katayama, Professor Masahiko Kondo, Professor Takashi Sugino, Professor Toshiaki Suhara, Professor Yusuke Mori, and Professor Seizo Morita (Division of Electrical, Electronic and Information Engineering, Graduate School of Engineering, Osaka University) for their helpful suggestions on the dissertation.

I am highly grateful to Associate Professor Toshimasa Matsuoka (Division of Electrical, Electronic and Information Engineering, Graduate School of Engineering, Osaka University) for his leading my research and valuable advices.

I would like to thank Dr. Guechol Kim (Mokpo National University) for his teaching measurement facilities and discussions on LNA measurements. I would like to express my appreciation to Mr. Hisayasu Sato (Renesas Technology and Osaka University) for his technical suggestions and helpful advices on my research. I would like to thank Dr. Yoshiyuki Shimizu (Texas Instruments Japan Limited), Dr. Sungwoo Cha, Mr. Keisuke Ueda (Renesas Technology), Mr. Yoshihiro Utsurogi (The Chugoku Electric Power Co., Inc.), Mr. Bunsei Murakami, Mr. Masaru Goto (Denso Corporation), Mr. Keiji Nakamura (Toyota Motor Corporation), Mr. Lee Tuck Yang (Singapore), Mr. Hae-Ju Park, Mr. Isao Takobe, and Mr. Fumiaki Yamashita (Osaka University) for their supports and discussions on radio-frequency integrated circuits. I also would like to thank Dr. Yoshinari Kamakura, Dr. Tsukasa Ida (Osaka University), Mr. Tomoyuki Tanaka (Renesas Technology), and Mr. Daisuke Kanemoto (Osaka University) for their valuable comments and help. I would like to thank Ms. Noriko Furuta and Ms. Yuko Nomura (Osaka University) for their various supports.

I appreciate various educational programs and financial supports from the Osaka University Global COE Program, "Center for Electronic Devices Innovation", granted from the Ministry of Education, Culture, Sports, Science and Technology of Japan. Finally, I would like to thank my family for their constant supports and caring.

Contents

1	Intr	oduction	1
	1.1	Background	1
	1.2	Fundamentals of Low-Noise Amplifiers	2
		1.2.1 Low-Noise Amplifier	2
		1.2.2 Performance Metrics	3
	1.3	Outline of This Dissertation	8
	Bibl	ography	8
2	Two	-Stage CMOS LNA 1	1
	2.1	Introduction	1
	2.2	Cascode LNA with Inductive Source Degeneration	2
		2.2.1 Analytical Expressions	2
		2.2.2 Limitations	6
	2.3	Two-Stage LNA with Inductive Source Degeneration	6
		2.3.1 Circuit Topology	6
		2.3.2 Small-Signal Equivalent Circuit	7
		2.3.3 Analytical Expressions	20
	2.4	Design Methodology	21
	2.5	Experimental Results and Discussion	25
		2.5.1 Inductor	25
		2.5.2 S-parameters	27
		2.5.3 NF	27
		2.5.4 Linearity	27
		2.5.5 Comparison	31
	2.6	Conclusion	32
	Bibl	ography	54
3	Trar	sformer Folded-Cascode CMOS LNA 3	37
-	3.1	Introduction	37
	3.2	Circuit Topology	;7
	3.3	Effect of Magnetic Coupling	38
		3.3.1 Input Impedance	;9

3.3.2 3.3.3 Noise 3.3.4 3.4 3.4.1 Transistors 3.4.2 Transformer and Inductor 3.5 3.5.1 3.5.2 NF 3.5.3 3.5.4 3.6 Conclusion Bibliography 4 Transformer Noise-Canceling UWB CMOS LNA 4.14.2Wideband CMOS LNAs 4.2.1 4.2.2 4.3 Transformer Noise-Canceling LNA 4.3.1 Circuit Topology 4.3.2 Noise Cancellation 4.4 4.4.1Noise 4.4.2 4.4.3 4.4.4 4.4.5 4.5 4.5.1 4.5.2 4.5.3 Output Series Inductor 4.6 4.6.1

 42

43

47

47

47

48

48

51

51

51

55

55

57

59 59

60

60

62

63

63

64

67

67

69

74

75

78

78

78

78

80

81

83

83

83

87

87

90

4.7

4.6.2

4.6.3

4.6.4

Bibliography

Conclusion

NF

Α	NF Derivations	95
	A.1 Two-Stage LNA	95
	A.1.1 F_{M_1}	96
	A.1.2 F_{M_2}	97
	A.1.3 F_{R_I} and F_{R_L}	98
	A.2 Transformer Folded-Cascode LNA	99
	A.2.1 F_{M_1}	00
	A.2.2 F_{M_2}	01
	A.2.3 F_{L_I} and F_{L_L}	02
	Bibliography	03
B	Frequency Responses of Y_I and $1/Y_I$	105
	$\mathbf{B}.1 Y_I \dots \dots \dots \dots \dots \dots \dots \dots \dots $	05
	B.2 $1/Y_I$	05
Ac	hievements	07

List of Figures

1.1	Block diagram of a typical wireless receiver.	2
1.2	LNA cascaded with a buffer for measurement.	3
1.3	Calculated S_{11} for $Z_{in} = R + sL + 1/sC$.	4
1.4	Third-order intercept point.	7
1.5	Stability of a two-port network	8
2.1	Schematic of the cascode LNA with inductive source degeneration	12
2.2	Small-signal equivalent circuit of the cascode LNA	13
2.3	Calculated NF versus W_1 with I_d as a parameter	14
2.4	Schematic of the two-stage LNA with inductive source degeneration	17
2.5	(a) Input section of the LNA and (b) the Thevenin's equivalent circuit of the input	
	section.	18
2.6	Small-signal equivalent circuit of the common-source stage	19
2.7	Small-signal equivalent circuit of the common-gate stage	19
2.8	Calculated and simulated voltage gain versus V_{od1} with I_{d1} as a parameter	23
2.9	Calculated and simulated NF and IIP_3 versus V_{od1}	23
2.10	Calculated and simulated NF and IIP_3 versus V_{od2}	24
2.11	Complete schematic of the designed two-stage LNA	26
2.12	Micrograph of the fabricated LNA	26
2.13	Measured and simulated L of the fabricated 3.1 nH inductor	28
2.14	Measured and simulated Q of the fabricated 3.1 nH inductor	28
2.15	Measured and simulated S_{11} and S_{21} of the LNA	29
2.16	Measured and simulated S_{12} and S_{22} of the LNA	29
2.17	Measured and simulated NF of the LNA	30
2.18	Measured IIP_3 of the LNA with the buffer	30
2.19	Measured IIP_3 of the stand-alone buffer	31
3.1	Schematic of the proposed LNA	38
3.2	Small-signal equivalent circuit of the input stage.	40
3.3	Small-signal equivalent circuit of the common-gate stage	40
3.4	Calculated (a) α_M , (b) Z_{in} , and (c) S_{11} with k as a parameter.	41
3.5	Calculated voltage gain with k as a parameter	44

3.6	Calculated voltage gain with k as a parameter for $L_L = L_I = 1/\omega_{n,t}^2(1+k)C_L$.	14
3.7	Mechanisms for noise reduction of i_{nd2} .	45
3.8	Calculated F_{M_2} , F_{R_I} , and F_{R_I} versus k	16
3.9	Calculated and simulated NF versus k	16
3.10	Layout of (a) a partially-coupled transformer and (b) a stacked transformer 4	19
3.11	Simulated voltage gain and NF of LNAs employing the partially-coupled trans-	
	former (solid line) and stacked transformer (dashed line).	19
3.12	Complete schematic of the designed transformer folded-cascode LNA	50
3.13	Micrograph of the proposed LNA (left) and conventional folded-cascode LNA	
	(right)	50
3.14	Measured and simulated S_{11} of the LNAs	52
3.15	Measured and simulated S_{21} of the LNAs	52
3.16	Measured and simulated S_{12} of the LNAs	53
3.17	Measured and simulated S_{22} of the LNAs	53
3.18	Measured and simulated NF of the LNAs	54
3.19	Measured IIP_3 of the proposed LNA with the buffer	54
	5	
4.1	UWB frequency bands	59
4.2	Wideband CMOS LNAs.	52
4.3	Schematic of the proposed LNA	54
4.4	Small-signal equivalent circuit with noise sources.	55
4.5	Mechanisms for noise cancellation of (a) i_{nd} and (b) v_{nR_L} .	56
4.6	Simulated NF and NF_{min} of the LNAs with and without noise cancellation ($k =$	
	0, 1.0)	58
4.7	Simulated noise contributions from M_1 , R_L , and R_p to the LNAs with and with-	
	out noise cancellation ($k = 0, 1.0$)	58
4.8	Calculated (a) F , F_{M_1} , and F_{R_L} ($R_L = 50 \ \Omega$) versus n and (b) NF with R_L as a	
	parameter	70
4.9	Calculated real and imaginary parts of (a) Y_T and (b) Y_{IN} , and (c) S_{11} of the	
	LNAs with and without L_1	72
4.10	Calculated A_v of the proposed LNA with k as a parameter	76
4.11	Calculated A_v and S_{21} of the proposed LNA with $k = 0.9$.	76
4.12	Simulated K and B_1 of the proposed LNA	77
4.13	Simulated group delays with (a) L_p and (b) L_1 as a parameter. \ldots \ldots \ldots	79
4.14	Simulated S_{11} and NF of the LNA with C_c as a parameter	30
4.15	(a) Top view and (b) cross section a–a' of the designed transformer 8	31
4.16	Complete schematic of the designed transformer noise-canceling LNA 8	32
4.17	Micrograph of the fabricated LNA	32
4.18	Measured and simulated S_{11} and S_{21} of the LNA	34
4.19	Measured and simulated S_{12} of the LNA	34
4.20	Measured and simulated group delays of the LNA.	35
4.21	Measured and simulated NF of the LNA	35

4.22	Measured IIP_3 of the LNA at 3 GHz.	86
4.23	Measured IIP_3 and IIP_2 of the LNA	86
4.24	Common-source amplifier with a load inductor.	89
4.25	Simulated S_{21} and NF of the LNAs with and without the common-source amplifier.	89
A.1	Noise equivalent circuit of the input stage	96
A.2	Noise equivalent circuit of the common-gate stage	98
A.3	Noise equivalent circuit of the input stage	100
A.4	Noise equivalent circuit of the common-gate stage.	102
B 1	Calculated (a) Y_t and (b) $1/Y_t$ with k as a parameter	106
2.1		. 00

List of Tables

2.1	Specifications of LNAs for WLAN receivers.	22
2.2	Comparison of the LNA performance at 5.4 GHz.	31
2.3	Measured performance and comparison of 1.0 V, 5 GHz CMOS LNAs	33
3.1	Measured performance and comparison of low-voltage CMOS LNAs	56
4.1	Measured performance and comparison of wideband CMOS LNAs	88

Chapter 1

Introduction

1.1 Background

The high-frequency performance of metal-oxide semiconductor field-effect transistors (MOS-FETs) has improved rapidly, due to advances in complementary metal-oxide-semiconductor (CMOS) process technologies [1]. This allows us to implement radio-frequency integrated circuits (RFICs) for the front-ends of wireless transceivers by using Si CMOS technologies. The Si CMOS technology has two advantages over SiGe or III-V compound semiconductor technologies, conventionally used for RFICs. First, the CMOS technology costs lower than the others, due to fewer masks layers and processing steps. For example, a 0.13 μ m CMOS process costs approximately 20% lower than a 0.13 μ m SiGe BiCMOS process [1, 2]. Second, the CMOS technology allows high integration of RF circuits with digital circuits on one chip, called systemon-a-chip (SoC), which can achieve higher performance and lower chip cost [3,4]. Thus, CMOS implementation of RFICs is indispensable for creating low-cost and high performance wireless communication devices.

However, the scaling of MOSFETs has imposed two stringent requirements on CMOS RFICs.

One is low-voltage operation. As MOSFETs scale down, the allowable supply voltages of ICs decrease to maintain the device reliability. Reference [5] predicts that the supply voltage of low-power digital ICs will decrease to 0.5 V in 2016. Considering the integration with digital circuits, we need to design RF circuits that can operate at the same supply voltage. Most existing RF circuits, which require more than 1.0 V supplies, can not operate under such a low supply voltage. Low-voltage circuit topologies are becoming more important.

The other is small chip area or low cost. The CMOS fabrication cost has been increasing dramatically with the scaling of MOSFETs. Reference [6] shows that a 45 nm (state-of-the-art) CMOS process costs approximately 10 times as much as a 0.13 μ m (most widely used) CMOS process. The chip cost of ICs mainly depends on the chip area. The chip area of digital circuits, which mostly consist of transistors, decreases as CMOS processes scale down, and consequently the chip cost does not increase dramatically. On the other hand, the chip area of RFICs remains almost constant even with the scaling of CMOS processes, because RF circuits require many pas-

sive components such as inductors, capacitors, and transmission lines. They shrink very slowly compared to transistors. Thus, the small-area implementation of RF circuits is another important consideration when using state-of-the-art CMOS technologies.

The goal of this research is to propose and demonstrate low-voltage (1.0 V or less) and small-area (0.25 mm² or less) CMOS low-noise amplifiers (LNAs) for narrowband (5 GHz) and wideband (3.1–10.6 GHz) wireless receivers. Narrowband receivers, which have input band-widths of less than 1 GHz, are employed in cellular phones, global positioning systems (GPS), Bluetooth systems, and wireless local area network (WLAN) systems, etc. On the other hand, wideband receivers are mainly used in ultra-wideband (UWB) systems and can be applied to multiband/multistandard systems. LNAs are essential building blocks for all wireless receivers, while they require relatively high supply voltages and large chip area (i.e., many inductors). This causes the difficulties for creating low-voltage and small-area receiver chips. In addition, the circuit topologies of narrowband LNAs are quite different from those of wideband ones, which means that different topologies suitable for each receiver are required. Low-voltage, small-area and narrowband/wideband CMOS LNAs can therefore contribute to low-voltage and low-cost wireless receivers.

1.2 Fundamentals of Low-Noise Amplifiers

1.2.1 Low-Noise Amplifier

The LNA is the first building block of RF front-ends for wireless receivers. Figure 1.1 shows a block diagram of a typical wireless receiver. The RF front-end, which consists of an LNA, mixer, local oscillator (LO), low-pass filter (LPF), and variable gain amplifier (VGA), amplifies and converts high-frequency signals into low-frequency ones with the desired signal-to-noise ratio (SNR). First, the LNA amplifies signals received by an antenna. Second, the mixer downconverts the high-frequency signals to low-frequency ones by using the LO and then the LPF filters out unwanted high-frequency components in the signals. Finally, the VGA adjusts the signal levels



Figure 1.1: Block diagram of a typical wireless receiver.

for an analog-to-digital converter (ADC). The RF front-end is usually integrated on one chip.

The LNA determines the input bandwidth and noise performance of the wireless receiver and must meet the following requirements:

- Input impedance matching The input impedance of the LNA is set to the characteristic impedance of the transmission line (i.e., $Z_0 = 50 \Omega$) for a small input reflection coefficient and maximum power transfer.
- Sufficient gain The LNA gain must be large enough to reduce the noise contributions from the following building blocks (mixer, LPF, and VGA) to the receiver.
- Low noise performance The noise performance of the receiver is mainly determined by that of the LNA.
- High linearity The maximum achievable linearity of the receiver is limited by the linearity of the LNA.
- Stability The LNA oscillates when it has a negative input or output resistance.

1.2.2 Performance Metrics

LNA performance metrics for input impedance matching, noise, gain, and linearity are described in terms of an LNA with a buffer with a 50 Ω output impedance, as shown in Fig. 1.2. RF devices are generally terminated with a resistance R_L of 50 Ω , and then measured using a signal source with a resistance R_s of 50 Ω . However, the output impedances of LNAs are not designed to 50 Ω but to be high, because the LNAs need to drive capacitive inputs of on-chip mixers or additional amplifiers. This means that, in the measurement of stand-alone LNAs, the 50 Ω termination causes inaccurate measurements, in particular, lower gain measurements. For accurate measurements of LNA performance, the LNAs are usually integrated together with buffers, whose input impedances are capacitive, emulating the practical termination. In addition, the buffers have 50 Ω



Figure 1.2: LNA cascaded with a buffer for measurement.

output impedances, providing output impedance matching. The effects of the buffers on the LNA performance can be ignored or de-embedded, as will be shown below.

Input Impedance Matching

The quality of input impedance matching is evaluated by the input reflection coefficient S_{11} , given by [7]

$$S_{11} = \frac{Z_{in} - R_s}{Z_{in} + R_s},\tag{1.1}$$

where Z_{in} is the input impedance of the LNA. Equation (1.1) indicates that the buffer has little impact on S_{11} . Figure 1.3 shows the frequency characteristic of S_{11} for $Z_{in} = R + sL + 1/sC$, which represents the input impedance of a typical narrowband LNA. An S_{11} of less than -10 dB is generally required for input impedance matching and the frequency range is called the input bandwidth.



Figure 1.3: Calculated S_{11} for $Z_{in} = R + sL + 1/sC$.

Gain

The LNA gain is measured as the forward transmission coefficient S_{21} [7]. The voltage gain of the LNA is defined as the ratio of the output voltage v_{out} to the input one v_{in} :

$$A_v \equiv \frac{v_{out}}{v_{in}}.\tag{1.2}$$

When the LNA is followed by a unity-gain buffer with $Z_{out} = R_L$ as shown in Fig. 1.2, S_{21} can be expressed as

$$S_{21} = \frac{v_L}{v_s/2} = 2\frac{v_{out}}{v_s},\tag{1.3}$$

where v_s represents the signal voltage and v_L the output voltage at R_L . The input voltage of the LNA is given by

$$v_{in} = \frac{Z_{in}}{R_s + Z_{in}} v_s. \tag{1.4}$$

Substituting Eq. (1.4) into Eq. (1.3), we obtain

$$S_{21} = \frac{2Z_{in}}{R_s + Z_{in}} \frac{v_{out}}{v_{in}} = \frac{2Z_{in}}{R_s + Z_{in}} A_v,$$
(1.5)

which corresponds to A_v for $Z_{in} = R_s$. This means that, under input impedance matching condition, the S_{21} of the LNA with the unity-gain buffer equals the voltage gain of the standalone LNA.

Noise Figure

The noise performance of the LNA is evaluated by the noise figure (NF), defined as $10 \log_{10} F$, where F is the ratio of the input SNR_{IN} to the output SNR_{OUT} [8]:

$$F \equiv \frac{SNR_{IN}}{SNR_{OUT}}.$$
(1.6)

Equation (1.6) indicates that NF is a measure of the degradation of the SNR. By introducing the total output noise power $P_{n,OUT}$ and the output noise power due to the source, P_{n,OUT,R_s} , we can express Eq. (1.6) as

$$F = \frac{P_{n,OUT}}{P_{n,OUT,R_s}},\tag{1.7}$$

which allows the NF calculations of the LNA.

The overall F of a cascaded system (i.e., receiver or LNA with a buffer) is given by the Friis formula [9]:

$$F_{all} = 1 + (F_1 - 1) + \frac{F_2 - 1}{G_p},$$
(1.8)

where F_1 and F_2 represent the noise factor of the first and second stages, respectively, and G_p the power gain of the first stage. Equation (1.8) indicates that the noise performance of the first stage is critical to the overall noise performance of the cascaded system and the gain of the first stage reduces the noise contribution from the second stage. In the case of wireless receivers, the first stage represents an LNA, and the second stage the other building block consisting of a mixer, filter, and VGA. In the measurement of the LNA with the buffer, the LNA usually has so large gain that the influence of the buffer on NF measurements is small.

Third-Order Intercept Point

The LNA linearity is evaluated by the third-order intercept point (IP_3), measured from a two tone test [10]. The input-output relationship of the LNA can be approximated by

$$v_{out}(t) \approx \alpha_1 v_{in}(t) + \alpha_2 v_{in}^2(t) + \alpha_3 v_{in}^3(t).$$
 (1.9)

Applying two tones with the same amplitude $A(v_{in}(t) = A \cos \omega_1 t + A \cos \omega_2 t)$ to the LNA, we obtain the output voltage as

$$v_{out}(t) = \left(\alpha_1 + \frac{9}{4}\alpha_3 A^2\right) A \cos \omega_1 t + \left(\alpha_1 + \frac{9}{4}\alpha_3 A^2\right) A \cos \omega_2 t + \frac{3}{4}\alpha_3 A^3 \cos \left(2\omega_1 - \omega_2\right) t + \frac{3}{4}\alpha_3 A^3 \cos \left(2\omega_2 - \omega_1\right) t + \cdots$$
(1.10)

The components at $2\omega_1 - \omega_2$ and $2\omega_2 - \omega_1$ are called third-order intermodulation (IM₃) products and appear in the vicinity of ω_1 and ω_2 for $\omega_1 \simeq \omega_2$. The fundamental component and IM₃ product are plotted versus the input on a logarithmic scale as shown in Fig. 1.4. The IM₃ products increase with the slope of three, whereas the fundamental components increase with the slope of one. The intersection of the two lines is IP₃, and the horizontal and vertical coordinates of IP₃ are called the input IP₃ (IIP₃) and output IP₃ (OIP₃), respectively. At the IP₃, the fundamental components have the same amplitude as the IM₃ products:

$$|\alpha_1|A_{IIP_3} = \frac{3}{4} |\alpha_3|A_{IIP_3}^3, \tag{1.11}$$

where $9\alpha_3 A^2/4$ is ignored against α_1 . The IIP₃ is thus given by

$$A_{IIP_3} = \sqrt{\frac{4}{3} \left| \frac{\alpha_1}{\alpha_3} \right|},\tag{1.12}$$

and the OIP₃ is equal to $\alpha_1 A_{IIP3}$. The IIP₃ expressed in power is as follows:

$$P_{IIP_3} = \frac{A_{IIP_3}^2}{2R_s},\tag{1.13}$$

which we will simply express as IIP_3 .

The A_{IIP_3} of a cascaded system is approximated as [10]

$$\frac{1}{A_{IIP_3}^2} \approx \frac{1}{A_{1,IIP_3}^2} + \frac{A_{v_1}^2}{A_{2,IIP_3}^2},\tag{1.14}$$

where A_{1,IIP_3} and A_{2,IIP_3} represent the A_{IIP_3} of the first and second stages, respectively, and A_{v1} the voltage gain of the first stage. Equation (1.14) indicates that the achievable maximum linearity of the cascaded system is limited by the first stage and the nonlinearity of the second stage become significant when the gain of the first stage is large. In the measurement of the LNA with the buffer, the IIP_3 without the effect of the buffer can be calculated from Eq. (1.14).



Figure 1.4: Third-order intercept point.

Stability

A two-port network like an LNA oscillates when either the input or output port presents a negative resistance. The two-port is unconditionally stable when it meets the following conditions [7]:

$$|\Gamma_s| < 1, \tag{1.15}$$

$$|\Gamma_L| < 1, \tag{1.16}$$

$$|\Gamma_{IN}| = \left| S_{11} + \frac{S_{12}S_{21}\Gamma_L}{1 - S_{22}\Gamma_L} \right| < 1, \tag{1.17}$$

$$\left|\Gamma_{OUT}\right| = \left|S_{22} + \frac{S_{12}S_{21}\Gamma_s}{1 - S_{11}\Gamma_s}\right| < 1,\tag{1.18}$$

where Γ_s , Γ_{IN} , Γ_L , and Γ_{OUT} represent the source, input, load, and output reflection coefficients, respectively, as show in Fig. 1.5. Equations (1.15)–(1.18) state that the real parts of the input and output impedances must be positive [7]:

$$\operatorname{Re}[Z_{IN}] > 0, \tag{1.19}$$

$$\operatorname{Re}[Z_{OUT}] > 0. \tag{1.20}$$

The unconditional stability of the two-port can be evaluted by other parameters: K and B_1 , given by [7]

$$K = \frac{1 - |S_{11}|^2 - |S_{22}|^2 + |\Delta|^2}{2|S_{12}S_{21}|},$$
(1.21)

$$B_1 = 1 + |S_{11}|^2 - |S_{22}|^2 - |\Delta|^2, \qquad (1.22)$$



Figure 1.5: Stability of a two-port network.

respectively, where $\Delta = S_{11}S_{22} - S_{12}S_{21}$. The necessary and sufficient conditions for unconditional stability are K > 1 and $B_1 > 0$.

1.3 Outline of This Dissertation

This dissertation proposes and demonstrates low-voltage and small-area CMOS LNAs for narrowband and wideband wireless receivers. The dissertation is organized as follows.

Chapter 2 presents a 1.0 V two-stage CMOS LNA with inductive source degeneration for 5 GHz applications. Complete analytical expressions of the LNA performance are derived, and then the design methodology based on the derived expressions are presented.

Chapter 3 presents a 0.5 V, 5 GHz area-efficient transformer folded-cascode CMOS LNA, which consumes much smaller chip area than the two-stage CMOS LNA presented in Chapter 2. The transformer reduces the chip area of a conventional folded-cascode CMOS LNA, but affects the LNA performance. The effects of the transformer are analyzed, and a transformer structure that has a small impact on the LNA performance are presented.

Chapter 4 presents a 1.0 V transformer noise-canceling CMOS LNA for fullband UWB (3.1–10.6 GHz) applications. The transformer partly cancels the noise originating from the transistor and load resistor, thereby improving the LNA noise performance without increased chip area and power consumption. The noise cancellation mechanisms are described and a wideband impedance matching technique is also presented.

Chapter 5 concludes the dissertation.

Bibliography

[1] H. S. Bennett, R. Brederlow, J. C. Costa, P. E. Cottrell, W. M. Huang, J. Anthony A. Immorlica, J.-E. Mueller, M. Racanelli, H. Shichijo, C. E. Weitzel, and B. Zhao, "Device and technology evolution for Si-based RF integrated circuits," *IEEE Trans. Electron Devices*, vol. 52, no. 7, pp. 1235–1258, Jul. 2005.

- [2] J. Pekarik, D. Geenberg, B. Jagannathan, R. Groves, J. R. Jones, R. Singh, A. Chinthakindi, X. Wang, M. Breitwisch, D. Coolbaugh, P. Coorell, J. Florky, G. Freeman, and R. Krishnasamy, "RFCMOS technology from 0.25μm to 65nm: The state of the art," in *Proc. IEEE Custom Integrated Circuits Conf.*, San Francisco, CA, Sep. 2004, pp. 217–224.
- [3] M. Hammes, C. Kranz, D. Seippel, J. Kissing, and A. Leyk, "Evolution on SoC integration: GSM baseband-radio in 0.13 μ m CMOS extended by fully integrated power management unit," *IEEE J. Solid-State Circuits*, vol. 43, no. 1, pp. 236–245, Jan. 2008.
- [4] A. Afsahi, J. J. Rael, A. Behzad, H.-M. Chien, M. Pan, S. Au, A. Ojo, C. P. Lee, S. B. Anand, K. Chien, S. Wu, R. Roufoogaran, A. Zolfaghari, J. C. Leete, T. Long, K. A. Carter, M. Nariman, K.-K. Yeung, W. Morton, M. Gonikberg, M. Seth, M. Forbes, J. Pattin, L. Gutierrez, S. Ranganathan, L. Ning, E. Blecker, J. Lin, T. Kwan, R. Zhu, M. Chambers, M. Rofougaran, A. Rofougaran, J. Trachewsky, and P. V. Rooyen, "A low-power single-weight-combiner 802.11abg SoC in 0.13 μm CMOS for embedded applications utilizing an area and power efficient Cartesian phase shifter and mixer circuit," *IEEE J. Solid-State Circuits*, vol. 43, no. 5, pp. 1101–1118, May 2008.
- [5] International Roadmap Committee. (2008) International technology roadmap for semiconductors 2007 edition. [Online]. Available: http://www.itrs.net
- [6] H. Tsuchikawa, M. Takakuwa, and S. Sugatani, "Electron beam direct writing technology combined with silicon shuttle service," in *4th ISMI Symp. on Manufacturing Effectiveness*, Austin, TX, Oct. 2007.
- [7] G. Gonzalez, *Microwave Transistor Amplifiers*, 2nd ed. Upper Saddle River, NJ: Prentice Hall, 1997.
- [8] T. H. Lee, *The Design of CMOS Radio-Frequency Integrated Circuits*, 2nd ed. Cambridge: Cambridge University Press, 2004.
- [9] H. T. Friis, "Noise figures of radio receivers," *Proc. IRE*, vol. 32, no. 7, pp. 419–422, Jul. 1944.
- [10] B. Razavi, *RF Microelectronics*. Upper Saddle River, NJ: Prentice Hall PTR, 1998.

Chapter 2

Two-Stage CMOS LNA

2.1 Introduction

Cascode CMOS LNAs with inductive source degeneration [1–3] are widely used for narrowband wireless receivers, such as GPS [4, 5], cellular phone [6–8], and WLAN receivers [9–11]. The cascode LNAs have achieved both good input impedance matching ($|S_{11}| < -10$ dB) and low noise performance ($NF \simeq 2.0$ dB) with reasonable power consumption (~ 10 mW), due to many design methodologies [1, 12–17] and advances in CMOS processes. However, the LNAs require relatively high supply voltages (>1.0 V) to achieve both good noise and linearity performance.

In addition, the previous design methodologies pay much attention on noise optimization, but little on its linearity. Modern wireless systems, in particular, cellular and WLAN systems impose high linearity requirements on LNAs [18, 19]. Low-voltage circuit topologies and design methodologies considering both noise and linearity are required.

This chapter proposes a two-stage CMOS LNA suitable for low-voltage operation and its design methodology based on the analytical expressions of the gain, noise, and linearity. These expressions are derived from the small-signal equivalent circuits of the LNA. The proposed design methodology is expanded from the previous my work [20], which is applied to the cascode CMOS LNA. The 1.0 V two-stage LNA designed for 5 GHz WLAN applications is implemented in a 0.15 μ m fully-depleted silicon-on-insulator (FD-SOI) CMOS technology [21, 22], and both its performance and design methodology are verified.

This chapter is organized as follows. Section 2.2 overviews the conventional cascode CMOS LNA with inductive source degeneration and describes its simple analytical expressions and limitations. Section 2.3 presents the two-stage CMOS LNA and its small-signal equivalent circuit, and then derives the analytical equations of the gain, noise, and linearity. Section 2.4 presents the design methodology based on these equations. Section 2.5 shows the measurement results of the fabricated LNA, and Section 2.6 concludes the chapter.

2.2 Cascode LNA with Inductive Source Degeneration

Figure 2.1 shows the conventional cascode LNA with inductive source degeneration [1–3]. The inductors L_g and L_s are connected to the gate and source terminals of the common-source transistor M_1 , providing input impedance matching at an operating frequency. The cascode transistor M_2 reduces the Miller effect due to the gate-drain capacitance of M_1 , improving the reverse isolation performance of the LNA. The load inductor L_L resonates with the load parasitic capacitance C_L , providing a high impedance.



Figure 2.1: Schematic of the cascode LNA with inductive source degeneration.

2.2.1 Analytical Expressions

Analytical expressions of the input impedance, gain, noise, and linearity of the cascode LNA are derived from the small-signal equivalent circuit shown in Fig. 2.2, where the resistors, $1/g_{m2}$ and R_s , represent M_2 and the signal source resistance, respectively.

Input Impedance

The source inductor L_s provides a resistive component for the LNA input impedance Z_{in} , and the gate inductor L_g adjusts the resonance frequency of Z_{in} to the desired operating frequency. The small-signal equivalent circuit shown in Fig. 2.2 gives the input impedance:

$$Z_{in} = \omega_{T_1} L_s + j\omega (L_g + L_s) + \frac{1}{j\omega C_{gs1}},$$
(2.1)



Figure 2.2: Small-signal equivalent circuit of the cascode LNA.

where $\omega_{T_1} = g_{m1}/C_{gs1}$ is the unity current gain frequency of M_1 and g_{m1} and C_{gs1} are the transconductance and gate-source capacitance of M_1 , respectively. Selecting L_s such that $L_s = R_s/\omega_{T_1}$ allows input impedance matching around the resonance frequency:

$$\omega_0 = \frac{1}{\sqrt{(L_g + L_s)C_{gs1}}}.$$
(2.2)

Gain

The voltage gain of the LNA is determined by the equivalent resistance of the load LC tank and ω_{T_1} . The output current i_{out} , given by $g_{m1}v_{gs1}$ where $v_{gs1} = v_{in}/sC_{gs1}Z_{in}$, flows into the load LC tank, and the voltage gain at ω_0 is therefore given by

$$A_{v} = \left| \frac{v_{out}}{v_{in}} \right| = \left| \frac{i_{out} R_{L_{L}}}{v_{in}} \right| = \left| \frac{g_{m1} R_{L_{L}}}{j\omega_{0} C_{gs1} Z_{in}} \right| = \frac{g_{m1} R_{L_{L}}}{\omega_{0} C_{gs1} \omega_{T_{1}} L_{s}} = \frac{R_{L_{L}}}{R_{s}} \left(\frac{\omega_{T_{1}}}{\omega_{0}} \right), \quad (2.3)$$

where R_{L_L} is the equivalent resistance of the load LC tank at the resonance frequency of ω_L $(= 1/\sqrt{L_L C_L})$ and is approximated by [23]

$$R_{L_L} \approx \frac{\omega_L^2 L_L^2}{R_{L_L,s}} = Q_{L_L}^2 R_{L_L,s},$$
(2.4)

where Q_{L_L} and $R_{L_{L,s}}$ are the quality factor and parasitic series resistance of L_L , respectively. The resonance frequency of the load LC tank, ω_L , is generally set to ω_0 .



Figure 2.3: Calculated NF versus W_1 with I_d as a parameter.

Noise

The LNA noise performance depends on the gate width of M_1 , W_1 , for a fixed current consumption. The noise factor of the LNA is given by [1,15]

$$F = 1 + \frac{\gamma \chi}{\alpha_1} g_{m1} R_s \left(\frac{\omega_0}{\omega_{T_1}}\right)^2 + \frac{\alpha_1 \delta}{\kappa g_{m1} R_s},$$
(2.5)

$$\chi = 1 - 2|c|\alpha_1 \sqrt{\frac{\delta}{\kappa\gamma} + \frac{\delta\alpha_1^2}{\kappa\gamma}},\tag{2.6}$$

where $\alpha_1 = g_{m1}/g_{d01}$ and g_{d01} is the zero-bias drain conductance of M_1 ; γ and δ are the drain noise current factor ($\gamma = 2/3$ in long-channel MOSFETs [24, 25], but $\gamma > 2/3$ in short-channel MOSFETs [26]) and the induced gate noise current factor ($\delta = 2\gamma$ [27]), respectively, and c is the correlation coefficient between these noise currents ($c \simeq j0.395$ [24]); κ is the Elmore constant ($\kappa = 5$ [27]). Equations (2.5) and (2.6) provide an optimum g_{m1} or W_1 for noise performance. The calculated NF versus W_1 with the drain current I_d as a parameter is shown in Fig. 2.3, where 0.15 μ m FD-SOI CMOS process parameters are used. Increasing I_d leads to a lower NF, and the optimum gate width $W_{1,opt}$ is found for each I_d . The gate width W_1 is generally set to $W_{1,opt}$.

Linearity

The cascode LNA consists of the common-source and cascode transistors, and hence the overall IIP_3 of the LNA can be approximated as that of a cascade system, as shown in Section 1.2.2:

$$\frac{1}{A_{LNA,IIP_3}^2} \approx \frac{1}{A_{1,IIP_3}^2} + \frac{A_{v1}^2}{A_{2,IIP_3}^2},\tag{2.7}$$

where A_{1,IIP_3} and A_{2,IIP_3} represent the IIP_3 of the common-source and cascode transistors in the expression of the voltage amplitude, respectively; A_{v1} represents the voltage gain of the common-source transistor and is given by

$$A_{v1} = \frac{1}{g_{m2}R_s} \left(\frac{\omega_{T_1}}{\omega_0}\right). \tag{2.8}$$

The IIP_3 of each transistor M_i (i = 1, 2) can be derived from the drain current equation, given by [28]:

$$I_{di} = \frac{1}{2} \mu_0 C_{ox} \frac{W_i}{L_i} \frac{V_{odi}^2}{1 + \Theta V_{odi}} \frac{1}{1 - \lambda V_{dsi}},$$
(2.9)

where V_{odi} , defined by $V_{odi} = V_{gsi} - V_{thi}$, is the overdrive voltage of M_i and V_{thi} is the threshold voltage of M_i ; W_i and L_i are the gate width and length of M_i , respectively; λ is the channel-length modulation coefficient; $\Theta = \mu_0/(2v_{sat}L) + \theta$ and v_{sat} is the saturation velocity of the carrier, μ_0 the carrier mobility under low electric field, and θ the mobility reduction parameter. For a signal small $v_{in}(t)$ applied to the LNA, the output current of each transistor is expressed as [19]

$$I_{di}(t) = \frac{c_0(c_1 + c'_i v_i(t))^2}{c_2 + c_3 v_i(t) + c'_i^2 c_4 v_i^2(t)} \quad (i = 1, 2),$$
(2.10)

$$c'_{i} = \frac{1}{j\omega_{0}C_{gs1}Z_{in}} \quad (i=1), \quad -1 \quad (i=2), \tag{2.11}$$

$$c_0 = \frac{1}{2}\mu_0 C_{ox} \frac{W_i}{L_i},\tag{2.12}$$

$$c_1 = V_{odi}, \tag{2.13}$$

$$c_2 = (1 + \Theta V_{odi})(1 - \lambda V_{dsi}), \qquad (2.14)$$

$$c_3 = (1 + \Theta V_{odi})\lambda A_{vi} + c'_i (1 - \lambda V_{dsi})\Theta, \qquad (2.15)$$

$$c_4 = \Theta \lambda A_{vi}, \tag{2.16}$$

where $v_1(t) = v_{in}(t)$ and $v_2(t) = -A_{v1}v_{in}(t)$. Substituting the Taylor expansion coefficients of Eq. (2.10) such as α_1 and α_3 into Eq. (1.12), we obtain

$$A_{i,IIP_3}^2 = \frac{4}{3} \left| \frac{\alpha_1}{\alpha_3} \right| = \frac{4}{3} \left| \frac{c_1 c_2^2 (c_1 c_3 - 2c_i' c_2)}{(-c_i' c_2 + c_1 c_3) [(-c_i' c_2 + c_1 c_3) c_3 - 2c_i' c_1 c_2 c_4]} \right|.$$
 (2.17)

Substituting Eqs. (2.8) and (2.17) into Eq. (2.7) gives the overall IIP_3 of the cascode LNA.

2.2.2 Limitations

Circuit Topology

The cascode LNA is not suitable for low-voltage operation, because it requires a supply voltage of more than two drain-source saturation voltages ($V_{DD} > 2V_{DS,sat}$) to operate the cascode transistor. No cascode transistor allows the LNA to operate at lower supply voltages, but causes poor performance (i.e., a lower gain and higher NF) due to the Miller effect. An alternative circuit to reduce the effect is required.

Small-Signal Equivalent Circuit

The small-signal equivalent circuit shown in Fig. 2.2 ignores the gate-drain capacitance of M_1 , C_{gd1} and input parasitic capacitance C_p originating from a gate inductor, input pad, and electro-static-discharge (ESD) devices. These capacitances cause the following effects on the LNA performance:

- C_{gd1} increases effective C_{gs1} and L_s (i.e., the Miller effect), having an impact on the input impedance matching as well as the noise and gain performance [19].
- C_p reduces Z_{in} , changing the input impedance matching condition [15,29].

The next section will present a low-voltage circuit topology with a small Miller effect and its complete small-signal equivalent circuit including the above parasitic capacitances.

2.3 Two-Stage LNA with Inductive Source Degeneration

This section presents a two-stage LNA with inductive source degeneration and describes its small-signal equivalent circuit and analytical expressions.

2.3.1 Circuit Topology

Figure 2.4 shows the two-stage LNA that consists of the common-source stage with inductive source degeneration and the common-gate stage. The two internal LC tanks, $L_{I1}C_{I1}$ and $L_{I2}C_{I2}$, provide high impedances at the resonance frequencies, and thereby the signal current amplified by the common-source transistor M_1 flows into the common-gate transistor M_2 , which alleviates the Miller effect of M_1 . The common-gate stage converts the current to the output voltage using the load LC tank L_LC_L . The DC-blocking capacitor C_c separates the DC voltages of two stages. The required supply voltage of the two-stage LNA is only more than $V_{DS,sat}$, so that the LNA can operate at lower supply voltages than the cascode LNA.



Figure 2.4: Schematic of the two-stage LNA with inductive source degeneration.

2.3.2 Small-Signal Equivalent Circuit

The input parasitic capacitance C_p varies the LNA input impedance. The input section of the two-stage LNA is shown in Fig. 2.5(a). The input impedance looking into the right hand side of reference plane 1 is given by

$$Z_{in}' = \frac{R_{in}(1 - \omega_0^2 L_g C_p) + j(\omega_0 L_g + X_{in} - \omega_0^2 L_g C_p X_{in})}{1 - \omega_0 C_p X_{in} + j\omega_0 C_p R_{in}},$$
(2.18)

where R_{in} and X_{in} are the resistance and reactance of the input impedance looking into the right hand side of reference plane 2, respectively. On the other hand, the equivalent source impedance looking into the left hand side of reference plane 2, $R_{eq} + j\omega_0 L_{eq}$, is given by

$$R_{eq} = \frac{R_s}{\omega_0^2 C_p^2 R_s^2 + (1 - \omega_0^2 C_p L_g)^2},$$
(2.19)

$$L_{eq} = \frac{L_g - C_p(\omega_0^2 L_g^2 + R_s^2)}{\omega_0^2 C_p^2 R_s^2 + (1 - \omega_0^2 C_p L_g)^2}.$$
(2.20)

Using the above impedances, we can derive the input impedance matching condition as

$$Z_{in}' = R_s \tag{2.21}$$

or

$$R_{eq} = R_{in},\tag{2.22}$$

$$\omega_0 L_{eq} = -X_{in}.\tag{2.23}$$



Figure 2.5: (a) Input section of the LNA and (b) the Thevenin's equivalent circuit of the input section.

The equivalent signal source voltage $v_{s,eq}$ applied to reference plane 2 is also given by

$$v_{s,eq} = \frac{v_s}{1 - \omega_0^2 C_p L_g + j \omega_0 C_p R_s}.$$
(2.24)

From Thevenin's theorem, the input section can be expressed as Fig. 2.5(b).

The gate-drain capacitance of M_1 , C_{gd1} , causes the Miller effect, varying the LNA input impedance. Figure 2.6 shows the small-signal equivalent circuit of the common-source stage including C_{gd1} . For more accurate analysis, the non-quasi-static (NQS) resistance $r_{nqs} = 1/\kappa g_{m1}$ [19] is also included. Using Figure 2.6, we can derive the input impedance $Z_{in} = R_{in} + jX_{in}$ at reference plan 2 as

$$R_{in} = \frac{r_{nqs} + \omega_{T_1} L_s}{\alpha_M},\tag{2.25}$$

$$X_{in} = \frac{\omega_0 L_s - \frac{1}{\omega_0 C_{gs1}}}{\alpha_M},\tag{2.26}$$

$$\alpha_{M} = 1 + \frac{g_{m1}}{Y_{I}/\alpha_{gd1} + j\omega C_{gs1}} + \frac{Y_{I}\left(1 - \omega^{2}L_{s}C_{gs1} + j\omega L_{s}g_{m1}\right)}{Y_{I}/\alpha_{gd1} + j\omega C_{gs1}}$$

$$\approx 1 + \alpha_{gd1}\left(\frac{g_{m1}}{Y_{I}} + 1 - \omega^{2}L_{s}C_{gs1} + j\omega L_{s}g_{m1}\right), \qquad (2.27)$$

$$\alpha_{gd1} = \frac{C_{gd1}}{C_{as1}},\tag{2.28}$$

$$Y_I = \frac{1}{R_I} + g_{m2}, (2.29)$$

where Y_I represents the input admittance of the common-gate stage at node I. In Eq. (2.27), $j\omega C_{gs1}$ is ignored against Y_I/α_{gd1} . The resistance R_I represents the parallel resistance of $R_{L_{II}}$



Figure 2.6: Small-signal equivalent circuit of the common-source stage.



Figure 2.7: Small-signal equivalent circuit of the common-gate stage.

and $R_{L_{I2}}$ (i.e., $R_{L_{I1}}//R_{L_{I2}}$), where $R_{L_{Ii}}$ (*i* =1, 2) is the equivalent resistance of the internal LC tank, $L_{Ii}C_{Ii}$, at the resonance frequency of $\omega_{L_{Ii}} = 1/\sqrt{L_{Ii}C_{Ii}}$ and is approximated by Eq. (2.4). The transconductance g_{mi} of M_i is given by a derivative of Eq. (2.9):

$$g_{mi} = \frac{\partial I_{di}}{\partial V_{odi}} = \frac{1}{2} \mu_0 C_{ox} \frac{W_i}{L_i} \frac{V_{odi}(2 + \Theta V_{odi})}{(1 + \Theta V_{odi})^2} \frac{1}{1 - \lambda V_{dsi}}.$$
 (2.30)

Equations (2.27) and (2.29) show that decreasing g_{m2} leads to a larger α_M , resulting in a lower Z_{in} .

Figure. 2.7 shows the small-signal equivalent circuit of the common-gate stage. The Miller effect due to C_{gd2} is negligible, because the gate terminal of M_2 is connected to the AC ground. The NQS resistance of M_2 is also negligible. The resistance R_L represents the equivalent resistance of the load LC tank L_LC_L , and is given by Eq. (2.4).

2.3.3 Analytical Expressions

The equivalent circuit presented in the previous subsection provides more precise analytical expressions of the LNA performance than the simple equivalent circuit shown in Section 2.2. The analytical equations of the gain, noise, and linearity are derived in this subsection.

Gain

The output current i_{out} flows into the output load R_L , generating the output voltage:

$$v_{out} = -i_{out}R_L. (2.31)$$

From Figs. 2.6 and 2.7, i_{out} is given by

$$i_{out} = \frac{g_{m1}v_{s,eq}}{2R_{in} \cdot j\omega_0 \alpha_M C_{gs1}} \left(\frac{R_I}{R_I + 1/g_{m2}}\right),$$
(2.32)

where input impedance matching is assumed (i.e., $R_{eq} = R_{in}$ and $\omega_0 L_{eq} = -X_{in}$). The voltage gain of the LNA is defined as the ratio of the output voltage to the input voltage at reference plane 1 (see Fig. 2.5(a)):

$$A_{v,LNA} \equiv \left| \frac{v_{out}}{v_{in}} \right| = 2 \left| \frac{v_{out}}{v_s} \right| = 2 \left| \frac{v_{out}}{v_{s,eq} (1 - \omega_0^2 C_p L_g + j\omega_0 C_p R_s)} \right|, \tag{2.33}$$

where $Z'_{in} = R_s$ is assumed. Substituting Eqs. (2.31) and (2.32) into (2.33) gives

$$A_{v,LNA} = \frac{R_L}{R_{in}\alpha_M} \left(\frac{\omega_{T_1}}{\omega_0}\right) \left|\frac{1}{1 - \omega_0^2 C_p L_g + j\omega_0 C_p R_s}\right| \left(\frac{R_I}{R_I + 1/g_{m2}}\right), \quad (2.34)$$

which indicates that a large C_p and small g_{m2} lead to a decrease in $A_{v,LNA}$.

Noise

The noise of M_1 , M_2 , R_I , and R_L contribute to the overall LNA noise. The LNA noise factor can be derived from the small-signal equivalent circuits shown in Figs. 2.6 and 2.7:

$$F_{LNA} = 1 + F_{M_1} + F_{M_2} + F_{R_I} + F_{R_L}, (2.35)$$

$$F_{M_1} = \frac{\gamma_1 \chi_1}{\alpha_1} g_{m1} R_{eq} \left(\frac{\omega_0}{\omega_{T_1}}\right)^2 + \frac{\alpha_1 \delta_1}{\kappa_1 g_{m1} R_{eq}},\tag{2.36}$$

$$F_{M_2} = 4R_{eq}\alpha_M^2 \left(\frac{\omega_0}{\omega_{T_1}}\right)^2 \frac{\gamma_2 \chi_2 g_{m2}}{\alpha_2},\tag{2.37}$$

$$F_{R_I} = 4R_{eq}\alpha_M^2 \left(\frac{\omega_0}{\omega_{T_1}}\right)^2 \frac{1}{R_I},\tag{2.38}$$

$$F_{R_L} = 4R_{eq}\alpha_M^2 \left(\frac{\omega_0}{\omega_{T_1}}\right)^2 \frac{(1+1/g_{m2}R_I)^2}{R_{L_{3,p}}},$$
(2.39)

where F_{M_1} , F_{M_2} , F_{R_I} , and F_{R_L} represent the noise contributions from M_1 , M_2 , R_I , and R_L , respectively, and χ_1 and χ_2 are given by

$$\chi_1 = (1 + \alpha_{gd1})^2 - 2|c|\alpha_1 \sqrt{\frac{\delta_1}{\kappa_1 \gamma_1}} (1 + \alpha_{gd1}) + \frac{\alpha_1^2 \delta_1}{\kappa_1 \gamma_1},$$
(2.40)

$$\chi_2 = \left(\frac{1 + 1/g_{m2}R_I}{1 + g_{m2}R_I}\right)^2 + \left(\frac{\omega_0}{\omega_{T_2}}\right)^2 \frac{\alpha_2^2 \delta_2}{\kappa_2 \gamma_2},$$
(2.41)

respectively. The detailed derivations are summarized in Appendix A.1. Equations (2.36)–(2.39) indicate that increasing ω_{T_1} leads to a lower NF and a lower g_{m2} (higher α_M) results in the increase of F_{M_2} , F_{R_I} , and F_{R_L} .

Linearity

The two-stage LNA consists of the common-source and common-gate stages, and hence the overall IIP_3 of the LNA can be derived in the same way as that of the cascode LNA:

$$\frac{1}{A_{LNA,IIP_3}^2} \approx \frac{1}{A_{1,IIP_3}^2} + \frac{A_{v1}^2}{A_{2,IIP_3}^2},\tag{2.42}$$

$$A_{v1} = \left| \frac{v_i}{v_{in}} \right| = \frac{\frac{1}{g_{m2}} / /R_I}{R_{in} \alpha_M} \left(\frac{\omega_{T_1}}{\omega_0} \right) \left| \frac{1}{1 - \omega_0^2 C_p L_g + j \omega_0 C_p R_s} \right|,$$
(2.43)

$$A_{i,IIP_3}^2 = \frac{4}{3} \left| \frac{c_1 c_2^2 (c_1 c_3 - 2c_i' c_2)}{(-c_i' c_2 + c_1 c_3) [(-c_i' c_2 + c_1 c_3) c_3 - 2c_i' c_1 c_2 c_4]} \right|,$$
(2.44)

$$c'_{i} = \frac{1}{j\omega_{0}\alpha_{M}C_{gs1}R_{in}(1-\omega_{0}^{2}L_{g}C_{p}+j\omega_{0}C_{p}R_{s})} \quad (i=1), -1 \quad (i=2),$$
(2.45)

where A_{1,IIP_3} and A_{2,IIP_3} represent the IIP_3 of the common-source and common-gate stages in the expression of the voltage amplitude, respectively; A_{v1} represents the voltage gain of the common-source stage and v_i the voltage at node I as shown in Fig. 2.6; $c_0 - c_4$ in Eq. (2.44) are given by Eqs. (2.12)–(2.16).

2.4 Design Methodology

This section describes a design methodology of the two-stage LNA that meets the typical specifications of LNAs for WLAN receivers [18], shown in Table 2.1. In this design, 0.15 μ m FD-SOI CMOS process and device parameters are used, and the current consumption and supply voltage are set to 8.0 mA and 1.0 V, respectively. There are thirteen design variables: the bias currents of the two transistors (I_{d1} and I_{d2}), the overdrive voltages (V_{od1} and V_{od2}), the gate widths and lengths (W_1/L_1 and W_2/L_2), gate and source inductances (L_g and L_s), internal inductances (L_{I1} and L_{I2}), and load inductance (L_L). The gate lengths, L_1 and L_2 , are set to the minimum gate length in order to increase ω_{Ti} , resulting in a lower NF [1]. Once I_{di} and V_{odi} are determined,
Frequency [GHz]	5.4
S_{11} [dB]	< -10
Voltage Gain [dB]	>20
NF [dB]	<2.0
IIP_3 [dBm]	> -5

Table 2.1: Specifications of LNAs for WLAN receivers.

 W_i can be calculated from Eq. (2.9). Consequently, we can reduce thirteen design variables to nine ones such as I_{d1} , I_{d2} , V_{od1} , V_{od2} , L_g , L_s , L_{I1} , L_{I2} , and L_L . In what follows, these nine design variables are determined from the derived equations.

Bias Currents of M_1 and M_2

The gain specification determines the distribution of bias currents of M_1 and M_2 , I_{d1} and I_{d2} . Equation (2.34) indicates that the voltage gain of the LNA mainly depends on ω_{T_1} ($\propto I_{di}$) and R_L (i.e., Q_L). The quality factors of on-chip inductors are determined by process technologies and inductor structures ($Q_L \simeq 8$ at 5 GHz in this design). Therefore, I_{d1} is selected to satisfy the gain specification, and then I_{d2} to the rest of the given bias current: $I_{d2} = I_{spec} - I_{d1}$. Figure 2.8 shows the calculated voltage gain versus V_{od1} with I_{d1} as a parameter. Note that g_{m2} is set to infinity in Eq. (2.34). For comparison, simulations for $I_{d1} = 7.0$ mA are also plotted in Fig. 2.8. This and the following simulations of the LNA were carried out using the small-signal and noise FD-SOI MOSFET models shown in [30] and Agilent Advanced Design System (ADS). Figure 2.8 shows that the calculations are comparable to the simulations, and the given specification is satisfied in the range of 4.0–8.0 mA. Taking account of process and temperature variations and the effect of M_2 , we select I_{d1} to 7.0 mA for a voltage gain of 23 dB including a 3 dB margin, and then I_{d2} to 1.0 mA.

Overdrive Voltage of M_1

Increasing overdrive voltage of M_1 , V_{od1} , leads to a lower NF and IIP_3 . Figure 2.9 shows the calculated and simulated NF and IIP_3 versus V_{od1} for $I_{d1} = 7.0$ mA. In the calculations and simulations, M_2 was set to be noiseless, and g_{m2} and A_{2,IIP_3}^2 to infinity in Eqs. (2.37)– (2.39) and (2.42), respectively. Besides, the noise parameters such as γ_i , δ_i , and α_i based on the experimental results [30] were used. Figure 2.9 shows that the calculations are comparable to the simulations and the noise performance improves with increasing V_{od1} , while the linearity deteriorates. The degradation of the linearity can be explained as follows: For input impedance matching ($Z'_{in} = R_s$), the constant current $i_{in} = v_{in}/R_s$ injects into the LNA. In this case, the



Figure 2.8: Calculated and simulated voltage gain versus V_{od1} with I_{d1} as a parameter.



Figure 2.9: Calculated and simulated NF and IIP_3 versus V_{od1} .



Figure 2.10: Calculated and simulated NF and IIP_3 versus V_{od2} .

gate-source voltage of M_1 is approximated by

$$v_{gs1} \approx \frac{i_{in}}{j\omega_0 C_{qs1}} = \frac{v_{in}}{j\omega_0 C_{qs1} R_s}.$$
 (2.46)

For a fixed I_{d1} , the gate width, i.e., the gate-source capacitance of M_1 , C_{gs1} , decreases with increasing V_{od1} , which results in an increase of v_{gs1} , as shown in Eq. (2.46). Although the MOSFET generally has higher linearity with increasing V_{od} (because I_{di} is proportional to V_{odi} when V_{odi} is large), the degradation of linearity due to the increase in v_{gs1} becomes significant in the range of 0.20–0.50 V, causing poor linearity. Taking account of additional noise of the common-gate stage, we set V_{od1} to 0.32 V, which results in $W_1 = 5 \times 24 \ \mu m$ (24 gate fingers, each with a unit of 5 μm width) and provides calculated NF = 1.25 dB and $IIP_3 = 3.8$ dBm.

Overdrive Voltage of M_2

Increasing the overdrive voltage of M_2 , V_{od2} , allows high linearity, but causing a higher NF. Figure 2.10 shows the calculated and simulated NF and IIP_3 versus V_{od2} for $I_{d1} = 7.0$ mA, $V_{od1} = 0.32$ V, and $I_{d2} = 1.0$ mA. Increasing V_{od2} results in better linearity but poor noise performance. The reason for a higher NF is that increasing V_{od2} leads to a decrease in g_{m2} and increase in α_M for a fixed I_{d2} , which results in a larger F_{M_2} , F_{R_I} , and F_{R_L} , as shown in Eqs. (2.37)–(2.39). Figure 2.10 also shows that the difference between the IIP_3 calculations and simulations increases at higher V_{od2} : the calculated IIP_3 becomes higher than the simulated IIP_3 as V_{od2} increases. This difference originates form the simplification of I_{di} (Eq. (2.9)) and the IIP_3 approximation of two nonlinear stages in cascade (Eq. (2.42)). With a few simulations, we can avoid to overestimate the achievable IIP_3 . From Fig. 2.10, we can find $V_{od2} = 0.17$ V, at which the noise and linearity specifications are satisfied on both calculation and simulation. This results in $W_2 = 50 \ \mu$ m and calculated NF = 1.85 dB and $IIP_3 = -4.1$ dBm.

Inductors

The gate and source inductances, L_g and L_s , are determined from the impedance matching conditions. Substituting the determined design variables into Eqs. (2.22) and (2.23) gives $L_g \simeq 3.3$ nH and $L_s \simeq 1.0$ nH. The source inductor L_s is implemented by a 2.5-turn square spiral inductor with a diameter of 115 μ m, a metal width of 8 μ m, and a metal spacing of 2 μ m, while the gate inductor L_g by a bonding wire [29], which has a higher Q than on-chip inductors [19, 31], resulting in a smaller NF.

The inductances in the LC tanks, L_{I1} , L_{I2} , and L_L , are determined from the resonance frequency, given by $\omega = 1/\sqrt{LC}$. Substituting f = 5.4 GHz and C = 300 fF into $L = 1/\omega^2 C$, we obtain an L of 3.1 nH. For these inductors, 3.5-turn square spiral inductors with a diameter of 170 μ m, a metal width of 8 μ m, and a metal spacing of 2 μ m are used.

Figure 2.11 shows the complete schematic of the designed two-stage LNA. For measurements, a unity-gain common-source amplifier with a 50 Ω output resistor is used as a buffer. The 10 pF capacitor provides the AC ground for the gate terminal of M_2 . The bias voltages are generated by current mirror circuits (not shown).

2.5 Experimental Results and Discussion

The designed two-stage LNA was fabricated in a 0.15 μ m FD-SOI CMOS process with a high resistivity substrate (~1 k Ω ·cm), metal-insulator-metal (MIM) capacitors, and five metal layers including a 1.95- μ m thick metal layer. The cut-off frequency of a 0.15 μ m NMOS consisting of 48 gate fingers with a unit of 5 μ m width was approximately 54 GHz for $V_{ds} = 1$ V and $I_d = 7$ mA [21]. A micrograph of the fabricated LNA is shown in Fig. 2.12. The active chip area excluding pads was 0.46 mm × 0.53 mm. The input and output pads were not ESD protected. The stand-alone buffer and inductor for L_{I1} , L_{I2} , and L_L were also fabricated on the same chip. The current consumption of the LNA and buffer were 8.3 mA and 5.8 mA from a 1.0 V supply voltage, respectively. The S-parameters, NF and IIP_3 of the LNA without L_g were measured using on-wafer RF probes. The above characteristics of the LNA with L_g were calculated based on the measurements [30]. This avoids instrumental error originating from a bonding wire L_g .

2.5.1 Inductor

The S-parameters of the fabricated 3.1 nH inductor were measured using an Agilent Technologies HP8722ES network analyzer and then converted into Y-parameters. The inductance and quality



Figure 2.11: Complete schematic of the designed two-stage LNA.



Figure 2.12: Micrograph of the fabricated LNA.

factor, L and Q, can be calculated from the Y-parameters and following equations [32]:

$$L = \operatorname{Im}\left[\frac{1/Y_{11}}{\omega}\right],\tag{2.47}$$

$$Q = \frac{\text{Im}\left[1/Y_{11}\right]}{\text{Re}\left[1/Y_{11}\right]}.$$
(2.48)

Figures 2.13 and 2.14 show the measured L and Q of the inductor, respectively. For comparison, the L and Q achieved using a three-dimensional (3-D) electromagnetic simulator (Ansoft HFSS) are also shown in Figs. 2.13 and 2.14, respectively. The measured L was 3.2 nH and Q was 8.0 at 5.4 GHz.

2.5.2 S-parameters

Figures 2.15 and 2.16 show the measured and simulated S-parameters of the LNA with an ideal gate inductor of 3.3 nH. The S-parameters were measured using the same network analyzer as in the inductor measurements. An S_{11} of less than -10 dB was achieved around 5.4 GHz, where a maximum S_{21} was 23 dB, which met the WLAN specification shown in Tab. 2.1. The agreements between the measurements and simulations are due to the small-signal FD-SOI MOSFET models proposed in [30]. Figure 2.16 shows that S_{12} and S_{22} are -46 dB and -9.9 dB at 5.4 GHz, respectively. The discrepancy between the measured and simulated S_{12} is attributed to measurement limits. The measured S_{12} of the stand-alone buffer (not shown) was -29 dB. The S_{12} of the stand-alone LNA was thus approximately -17 dB.

2.5.3 NF

Figure 2.17 shows the measured and simulated NF. The NF was measured using an Agilent Technologies HP8970B noise figure meter and Maury automated tuner system. The LNA achieved an NF of 1.7 dB at 5.4 GHz, satisfying the noise specification. The measurements agree well with the simulations, due to the noise models proposed in [30].

2.5.4 Linearity

Figures 2.18 and 2.19 show the measured output power of the fundamental tones and thirdorder intermodulation (IM₃) products for two tones (5.4 and 5.41 GHz), applied to the LNA and stand-alone buffer. The two tones were generated by Agilent Technologies HP8671B and E4438C signal generators, and the fundamental and IM₃ tones were measured using an Agilent Technologies E4448A spectrum analyzer. The measured *IIP*₃ of the LNA with the buffer was -18.0 dBm, while that of the stand-alone buffer was 4.8 dBm. The *IIP*₃ of the LNA without the buffer can be calculated from the following equation:

$$\frac{1}{A_{LNA,IIP_3}^2} \approx \frac{1}{A_{LNA+Buf,IIP_3}^2} - \frac{A_{v,LNA}^2}{A_{Buf,IIP_3}^2},$$
(2.49)



Figure 2.13: Measured and simulated L of the fabricated 3.1 nH inductor.



Figure 2.14: Measured and simulated Q of the fabricated 3.1 nH inductor.



Figure 2.15: Measured and simulated S_{11} and S_{21} of the LNA.



Figure 2.16: Measured and simulated S_{12} and S_{22} of the LNA.



Figure 2.17: Measured and simulated NF of the LNA.



Figure 2.18: Measured IIP_3 of the LNA with the buffer.



Figure 2.19: Measured IIP_3 of the stand-alone buffer.

where $A_{v,LNA}$ is the voltage gain of the LNA and A_{LNA,IIP_3} , A_{Buf,IIP_3} , and $A_{LNA+Buf,IIP_3}$ represent the IIP_3 of the LNA, buffer, and LNA cascaded with the buffer in the expression of voltage amplitude, respectively. Substituting $IIP_{3,LNA+Buf} = -18.0$ dBm, $IIP_{3,Buf} = 4.8$ dBm and $A_{v,LNA} = 22.5$ dB into Eq. (2.49) gives $IIP_{3,LNA} = -6.1$ dBm.

2.5.5 Comparison

Table 2.2 shows a comparison of the LNA performance obtained from the measurements, simulations, and calculations. The simulated and calculated voltage gain and NF were consistent with the measured results, which satisfied the specifications. The calculated IIP_3 agreed well with the simulated one, but these results were slightly different from the measured IIP_3 . The difference can be attributed to inaccurate FD-SOI MOS device parameters used in the simulations

	Gain [dB]	<i>NF</i> [dB]	IIP_3 [dBm]
Specification	20	2.0	-5.0
Measurement	23	1.70	-6.1
Simulation	25	1.60	-3.0
Calculation	23	1.85	-4.1

Table 2.2: Comparison of the LNA performance at 5.4 GHz.

and calculations. Although the fabricated LNA did not meet the IIP_3 specification, its linearity can be improved by increasing I_{d2} as shown in Section 2.4.

Table 2.3 shows a summary of the LNA and a comparison with previously reported 1.0 V, 5 GHz CMOS LNAs. The figure of merits for LNAs, FoM_1 and FoM_2 included in Table 2.3, are defined as [36]:

$$FoM_1[\mathbf{m}\mathbf{W}^{-1}] = \frac{Gain[\mathbf{lin}]}{Power[\mathbf{m}\mathbf{W}] \cdot (NF[\mathbf{lin}] - 1)},$$
(2.50)

$$FoM_2[-] = \frac{Gain[\lim] \cdot IIP_3[\mathbf{mW}] \cdot f_0[\mathbf{GHz}]}{Power[\mathbf{mW}] \cdot (NF[\lim] - 1)}.$$
(2.51)

The proposed LNA obtained the best FoM_1 among the other 1.0 V, 5 GHz CMOS LNAs. Although the LNA reported in [34] achieved the lowest NF and best FoM_2 , it adopted an inputoutput differential topology and had difficulty in achieving input impedance matching.

2.6 Conclusion

This chapter has demonstrated a 1.0 V two-stage CMOS LNA and its design methodology based on derived analytical expressions. The presented two-stage topology that consists of commonsource and common-gate stages is more suitable for low-voltage operation than a conventional cascode topology. The analytical expressions show that a higher V_{od1} results in a lower NFand IIP_3 while a higher V_{od2} leads to a higher NF and IIP_3 . The proposed design methodology based on the expressions allows us to efficiently design two-stage LNAs that satisfy target specifications. The 1.0 V, 5.4 GHz LNA implemented with a 0.15 μ m FD-SOI CMOS technology achieved an NF of 1.7 dB, voltage gain of 23 dB, and IIP_3 of -6.1 dBm with a power consumption of 8.3 mW. These measurements were consistent with the calculations obtained from the derived expressions. This ensures the validity of the analytical expressions and design methodology.

Frequency S_{11} NF S_{21} IIP_3 PowerArea FoM_1 [GHz][dB][dB][dB][dBm][mW][mm ²] $[mW^{-1}]$ FOM_1	5.4 -18 1.7 23 -6.1 8.3 0.25 3.55 4.71	5.5 -12.7 2.8 15.0 -5.6 11.1 0.30 0.56 0.85	5.75 N/A 0.9 14.2 0.9 16 0.24 2.78 9.85	5.8 -5.3 2.5 13.2 N/A 22.2 0.54 0.26 N/A
$ \begin{array}{c c} S_{21} & III \\ dB \end{array} $	23 -6	5.0 -5	4.2 0.9	3.2 N/
$\begin{bmatrix} NF \\ [dB] \end{bmatrix} \begin{bmatrix} c \\ c \end{bmatrix}$	1.7	2.8 1	0.9 1	2.5 1
$\begin{bmatrix} S_{11} \\ [dB] \end{bmatrix}$	-18	-12.7	N/A	-5.3
Frequency [GHz]	5.4	5.5	5.75	5.8
CMOS Technology	150 nm FD-SOI	90 nm	180 nm	180 nm
eference	This work	[33]*	[34] [†]	[35]*

Table 2.3: Measured performance and comparison of 1.0 V, 5 GHz CMOS LNAs.

*On-chip gate inductor used † Input-output differential topology

Bibliography

- D. K. Shaeffer and T. H. Lee, "A 1.5 V, 1.5 GHz CMOS low noise amplifier," *IEEE J. Solid-State Circuits*, vol. 32, no. 5, pp. 745–759, May 1997.
- [2] —, "Corrections to "A 1.5-V, 1.5-GHz CMOS low noise amplifier"," *IEEE J. Solid-State Circuits*, vol. 40, no. 6, pp. 1397–1398, Jun. 2005.
- [3] —, "Comment on corrections to "A 1.5-V,. 1.5-GHz CMOS low noise amplifier"," *IEEE J. Solid-State Circuits*, vol. 41, no. 10, pp. 2359–2359, Oct. 2006.
- [4] —, *The Design and Implementation of Low-Power CMOS Radio Receivers*. Dordrecht: Kluwer Acadmic Publishers, 1999.
- [5] G. Gramegna, P. G. Mattos, M. Losi, S. Das, M. Franciotta, N. G. Bellantone, M. Vaiana, V. Mandara, and M. Paparo, "A 56-mW 23-mm² single-chip 180-nm CMOS GPS receiver with 27.2-mW 4.1-mm² radio," *IEEE J. Solid-State Circuits*, vol. 41, no. 3, pp. 540–551, Mar. 2006.
- [6] M. S. J. Steyaert, H. Janssens, B. D. Muer, and M. Borremans, "A 2-V CMOS cellular transceiver front-end," *IEEE J. Solid-State Circuits*, vol. 35, no. 12, pp. 1895–1907, Dec. 2000.
- [7] J. Rogin, I. Kouchev, G. Brenna, D. Tschopp, and Q. Huang, "A 1.5-V 45-mW directconversion WCDMA receiver IC in 0.13-μm CMOS," *IEEE J. Solid-State Circuits*, vol. 38, no. 12, pp. 2239–2248, Dec. 2003.
- [8] B. Bakkaloglu, P. Fontaine, A. N. Mohieldin, S. Peng, S. J. Fang, and F. Dulger, "A 1.5-V multi-mode quad-band RF receiver for GSM/EDGE/CDMA2K in 90-nm digital CMOS process," *IEEE J. Solid-State Circuits*, vol. 41, no. 5, pp. 1149–1159, May 2006.
- [9] T. H. Lee, "5-GHz CMOS wireless LANs," *IEEE Trans. Microw. Theory Tech.*, vol. 50, no. 1, pp. 268–276, 2002.
- [10] M. Zargari, M. Terrovitis, S. H.-M. Jen, B. J. Kaczynski, M. Lee, M. P. Mack, S. S. Mehta, S. Mendis, K. Onodera, H. Samavati, W. W. Si, K. Singh, A. Tabatabaei, D. Weber, D. K. Su, and B. A. Wooley, "A single-chip dual-band tri-mode CMOS transceiver for IEEE 802.11a/b/g wireless LAN," *IEEE J. Solid-State Circuits*, vol. 39, no. 12, pp. 2239–2249, Dec. 2004.
- [11] T. Maeda, H. Yano, S. Hori, N. Matsuno, T. Yamase, T. Tokairin, R. Walkington, N. Yoshida, K. Numata, K. Yanagisawa, Y. Takahashi, M. Fujii, and H. Hida, "Low-powerconsumption direct-conversion CMOS transceiver for multi-standard 5-GHz wireless LAN systems with channel bandwidths of 5-20 MHz," *IEEE J. Solid-State Circuits*, vol. 41, no. 2, pp. 375–383, Feb. 2006.

- [12] P. Andreani and H. Sjoeland, "Noise optimization of an inductively degenerated CMOS low noise amplifier," *IEEE Trans. Circuits Syst. II*, vol. 48, no. 9, pp. 835–841, Sep. 2001.
- [13] J.-S. Goo, H.-T. Ahn, D. J. Ladwig, Z. Yu, T. H. Lee, and R. W. Dutton, "A noise optimization technique for integrated low-noise amplifiers," *IEEE J. Solid-State Circuits*, vol. 37, no. 8, pp. 994–1002, Aug. 2002.
- [14] T.-K. Nguyen, C.-H. Kim, G.-J. Ihm, M.-S. Yang, and S.-G. Lee, "CMOS low-noise amplifier design optimization technique," *IEEE Trans. Microw. Theory Tech.*, vol. 52, no. 5, pp. 1433–1441, May 2004.
- [15] P. Sivonen and A. Parssinen, "Analysis and optimization of packaged inductively degenerated common-source low-noise amplifiers with ESD protection," *IEEE Trans. Microw. Theory Tech.*, vol. 53, no. 4, pp. 1304–1313, Apr. 2005.
- [16] K.-J. Sun, Z.-M. Tsai, K.-Y. Lin, and H. Wang, "A noise optimization formulation for cmos low-noise amplifiers with on-chip low-Q inudctors," *IEEE Trans. Microw. Theory Tech.*, vol. 54, no. 4, pp. 1554–1560, Apr. 2006.
- [17] L. Belostotski and J. W. Haslett, "Noise figure optimization of inductively degenerated CMOS LNAs with integrated gate inductors," *IEEE Trans. Circuits Syst. I*, vol. 53, no. 7, pp. 1409–1422, Jul. 2006.
- [18] M. Brandolini, P. Rossi, D. Manstretta, and F. Svelto, "Toward multistandard mobile terminals-fully integrated receivers requirements and architectures," *IEEE Trans. Microw. Theory Tech.*, vol. 53, no. 3, pp. 1026–1038, Mar. 2005.
- [19] J. Janssens and M. Steyaert, CMOS Cellular Receiver Front-Ends : from Specification to Realization. Boston: Kluwer Academic Publishers, 2002.
- [20] T. Kihara, G. Kim, Y. Shimizu, B. Murakami, K. Ueda, M. Goto, T. Matsuoka, and K. Taniguchi, "Design of CMOS low-noise amplifier considering noise and linearity," *IE-ICE Trans. Electronics (Japanese Edition)*, vol. J89-C, no. 2, pp. 72–75, Feb. 2006.
- [21] F. Ichikawa, Y. Nagatomo, Y. Katakura, M. Itoh, S. Itoh, H. Matsuhashi, T. Ichimori, N. Hirashita, and S. Baba, "Fully depleted SOI process and device technology for digital and RF applications," *Solid-State Electron*, Jun. 2004.
- [22] A. Uchiyama, S. Baba, Y. Nagatomo, and J. Ida, "Fully depleted SOI technology for ultra low power digital and RF applications," in *Proc. IEEE Int. SOI Conf.*, Niagara Falls, NY, Oct. 2006.
- [23] B. Razavi, RF Microelectronics. Upper Saddle River, NJ: Prentice Hall PTR, 1998.
- [24] A. van der Ziel, *Noise in Solid State Devices and Circuits*. Toronto: John Wiley & Sons, 1986.

- [25] A. J. Scholten, L. F. Tiemeijer, R. van Langevelde, R. J. Havens, A. T. A. Z. van Duijnhoven, and V. C. Venezia, "Noise modeling for RF CMOS circuit simulation," *IEEE Trans. Electron Devices*, vol. 50, no. 3, pp. 618–632, Mar. 2003.
- [26] P. P. Jindal, "Compact noise models for MOSFETs," *IEEE Trans. Electron Devices*, vol. 53, no. 9, pp. 2051–2061, Sep. 2006.
- [27] Y. Tsividis, *Operation and Modeling of the Mos Transistor*, 2nd ed. New York: Oxford University Press, 2003.
- [28] B. Razavi, Design of Analog CMOS Integrated Circuits. New York: McGraw-Hill, 2001.
- [29] P. Leroux, J. Janssens, and M. Steyaert, "A 0.8-dB NF ESD-protected 9-mW CMOS LNA operating at 1.23 GHz," *IEEE J. Solid-State Circuits*, vol. 37, no. 6, pp. 760–756, Jun. 2002.
- [30] G. Kim, B. Murakami, M. Goto, T. Kihara, K. Nakamura, Y. Shimizu, T. Matsuoka, and K. Taniguchi, "Small-signal and noise model of FD-SOI MOS devices for low noise amplifier," *Jpn. J. Appl. Phys.*, *Pt.1*, vol. 45, no. 9A, pp. 6872–6877, Sep. 2006.
- [31] T. H. Lee, *The Dsign of CMOS Radio-Frequency Integrated Circuits*, 2nd ed. Cambridge: Cambridge University Press, 2004.
- [32] A. M. Niknejad and R. G. Meyer, "Analysis, design, and optimization of spiral inductors and transformers for Si RF IC's," *IEEE J. Solid-State Circuits*, vol. 33, no. 10, pp. 1470– 1481, Oct. 1998.
- [33] D. Linten, L. Aspemyr, W. Jeamsaksiri, J. Ramos, A. Mercha, S. Jenei, S. Thijs, R. Garcia, H. Jacobsson, P. Wambacq, S. Donnay, and S. Decoutere, "Low-power 5 GHz LNA and VCO in 90 nm RF CMOS," in *Symp. VLSI Circuits Dig. Tech. Papers*, Honolulu, HI, Jun. 2004, pp. 372–375.
- [34] D. J. Cassan and J. R. Long, "A 1-V transformer-feedback low-noise amplifier for 5-GHz wireless LAN in 0.18-μm CMOS," *IEEE J. Solid-State Circuits*, vol. 38, no. 3, pp. 427–435, Mar. 2003.
- [35] T. K. K. Tsang and M. N. El-Gamal, "Gain and frequency controllable sub-1 V 5.8 GHz CMOS LNA," in *Proc. IEEE Int. Symp. Circuits and Systems*, vol. 4, Scottsdale, AZ, May 2002, pp. 795–798.
- [36] R. Brederlow, W. Weber, S. Donnay, P. Wambacq, J. Sauerer, and M. Vertregt, "A mixedsignal design roadmap," *IEEE Des. Test. Comput.*, vol. 18, no. 6, pp. 34–46, Nov.–Dec. 2001.

Chapter 3

Transformer Folded-Cascode CMOS LNA

3.1 Introduction

Although the continuous scaling of CMOS technologies has improved the high-frequency performance of MOSFETs, it has imposed two challenges on CMOS RFICs: low-voltage operation and small chip area. The ITRS [1] predicts that the supply voltages of low-power digital circuits will decrease to 0.5 V in the near future. Reference [2] shows that a 45 nm (state-of-the-art) CMOS process costs approximately 10 times as much as a 0.13 μ m (most widely used) CMOS process. Considering the integration of RF circuits with digital circuits, we need to develop low-voltage and small-area (low-cost) RF circuits.

Folded-cascode CMOS LNAs with inductive source degeneration [3,4] are the most promising candidates for low-voltage and small-area CMOS LNAs. Although a two-stage CMOS LNA, presented in the previous chapter, achieves higher performance with lower power consumption than the folded-cascode LNAs, it requires more inductors (i.e., five inductors). Other reported low-voltage LNAs [5,6] consume much larger chip area and less performance than the above LNAs.

This chapter proposes a 0.5 V, 5 GHz transformer folded-cascode CMOS LNA [7], which has a smaller chip area than the conventional folded-cascode LNA. The transformer consists of the internal and load inductors and reduces the chip area of the LNA, while affecting the LNA performance. This chapter is organized as follows. Section 3.2 describes the circuit topology of the proposed LNA. The effects of the transformer on the LNA performance are analyzed in Section 3.3. Section 3.4 describes the design of the LNA and transformer. Section 3.5 presents the measurements of the LNA fabricated in a 90 nm digital CMOS process, and then Section 3.6 concludes the chapter.

3.2 Circuit Topology

Figure 3.1 shows a schematic of the proposed LNA, based on the conventional folded-cascode LNA with inductive source degeneration. The PMOS transistor M_2 reduces the Miller effect of



Figure 3.1: Schematic of the proposed LNA.

the gate-drain capacitance of the input transistor M_1 , improving the reverse isolation performance of the LNA. It allows less inductors than the NMOS transistor in the two-stage LNA, although it leads to less gain and a larger NF due to a larger parasitic capacitance at node I. The gate and source inductors, L_g and L_s , provide input impedance matching at an operating frequency [8]. The internal inductor L_I , resonating with the parasitic capacitance C_I at node I, provides a high impedance, thereby the signal current amplified by M_1 flows into M_2 . The load inductor L_L also resonates with the parasitic capacitance C_L , resulting in a high impedance. These inductors, L_I and L_L , are magnetically coupled to form a transformer in such a way as to have a positive magnetic coupling with retaining the LNA performance.

The positive magnetic coupling of L_I and L_L is the most effective way to reduce the chip area of the folded-cascode LNA. Increasing the magnetic coupling leads to a smaller L_I and L_L (smaller chip area), as will be shown in the next section. On the contrary, the negative magnetic coupling requires a larger L_I and L_L (larger chip area). The coupling of L_g or L_s and L_I or L_L is also not beneficial for the following reasons:

- 1. L_g is often implemented with a bonding wire.
- 2. L_s is usually small (< 1.0 nH) for input impedance matching.
- 3. The coupling makes the LNA unstable.

3.3 Effect of Magnetic Coupling

The magnetic coupling between L_I and L_L affects the LNA performance in terms of input impedance, gain, and noise. In this section, the effects of the magnetic coupling are analyzed, and the stability of the LNA is also discussed.

3.3.1 Input Impedance

The magnetic coupling changes the frequency response of the LNA input impedance Z_{in} through the gate-drain capacitance of M_1 , C_{gd1} . The small-signal equivalent circuit of the input stage, shown in Fig. 3.2, yields Z_{in} , given by

$$Z_{in} = j\omega L_g + \frac{\omega_{T1}L_s + j\omega L_s + \frac{1}{j\omega C_{gs1}}}{\alpha_M},$$
(3.1)

$$\alpha_M = 1 + \frac{g_{m1}}{Y_I / \alpha_{gd1} + j\omega C_{gs1}} + \frac{Y_I \left(1 - \omega^2 L_s C_{gs1} + j\omega L_s g_{m1}\right)}{Y_I / \alpha_{gd1} + j\omega C_{gs1}}$$

$$\approx 1 + \alpha_{gd1} \left(\frac{g_{m1}}{Y_I} + 1 - \omega^2 L_s C_{gs1} + j\omega L_s g_{m1}\right),$$
(3.2)

where $\omega_{T_1} = g_{m1}/C_{gs1}$ is the unity current gain frequency of M_1 ; $\alpha_{gd1} = C_{gd1}/C_{gs1}$ is the ratio between C_{gd1} and C_{gs1} ; $j\omega C_{gs1}$ is ignored against Y_I/α_{gd1} . The input admittance of the common-gate stage, shown in Fig. 3.3, is given by

$$Y_{I} = \frac{i_{i}}{v_{i}} = g_{m2} + j\omega C_{I} + \frac{1}{j\omega L_{I} + R_{I}} - \frac{j\omega M}{j\omega L_{I} + R_{I}} \frac{g_{m2} - j\omega C_{L} \cdot \frac{j\omega M}{j\omega L_{I} + R_{I}}}{1 - \omega^{2} L_{L} C_{L} - j\omega C_{L} \cdot \frac{(j\omega M)^{2}}{j\omega L_{I} + R_{I}} + j\omega R_{L} C_{L}}$$

$$\approx g_{m2} + j\omega C_{I} + \frac{1}{j\omega L_{I} + R_{I}} - \frac{nk \left(g_{m2} - j\omega nk C_{L}\right)}{1 - \omega^{2} (1 - k^{2}) L_{L} C_{L} + j\omega R_{L} C_{L}},$$
(3.3)

where g_{m2} is the transconductance of M_2 ; R_I , ignored in the last term for simplicity, and R_L are the parasitic resistances of L_I and L_L , respectively; M is the mutual inductance of the transformer, and k and $n = \sqrt{L_L/L_I}$ are the coupling factor and turn ratio, respectively. The frequency responses of Y_I and $1/Y_I$ are described in Appendix B. The calculated real and imaginary parts of α_M and Z_{in} are shown in Figs. 3.4(a) and (b), respectively. Using Re[α_M] and Im[α_M], we can approximate Z_{in} as

$$\operatorname{Re}[Z_{in}] \approx \frac{\omega_{T1} L_s \cdot \operatorname{Re}[\alpha_M] - \frac{\operatorname{Im}[\alpha_M]}{\omega C_{gs1}}}{|\alpha_M|^2}, \qquad (3.4)$$

$$\operatorname{Im}[Z_{in}] \approx \omega L_g - \frac{\omega_{T1} L_s \cdot \operatorname{Im}[\alpha_M] + \frac{\operatorname{Re}[\alpha_M]}{\omega C_{gs1}}}{|\alpha_M|^2}, \qquad (3.5)$$

where ωL_s is ignored against $1/\omega C_{gs1}$. As shown in Fig. 3.4(a), $\operatorname{Re}[\alpha_M]$ increases and $\operatorname{Im}[\alpha_M]$ decreases at low frequencies, which results in an increase of $\operatorname{Re}[Z_{in}]$. For a low k (<0.6), $\operatorname{Re}[Z_{in}]$ becomes a maximum around the frequency at which $\operatorname{Im}[\alpha_M]$ becomes a minimum. Meanwhile, $\operatorname{Im}[Z_{in}]$ with k approaches zero faster than $\operatorname{Im}[Z_{in}]$ without k (k = 0), due to the increase in $\operatorname{Re}[\alpha_M]$ at low frequencies.

The magnetic coupling shifts the input impedance matching region ($|S_{11}| < -10$ dB) toward lower frequencies. Due to good reverse isolation of the folded-cascode topology ($S_{12} \simeq 0$), the



Figure 3.2: Small-signal equivalent circuit of the input stage.



Figure 3.3: Small-signal equivalent circuit of the common-gate stage.

 S_{11} of the LNA can be approximated as Eq. (1.1). For input impedance matching at ω_0 , the following conditions must be satisfied:

$$\operatorname{Re}[Z_{in}] \simeq R_s, \tag{3.6}$$

$$\operatorname{Im}[Z_{in}] \simeq 0, \tag{3.7}$$

$$\operatorname{Im}[\alpha_M] \simeq 0, \tag{3.8}$$

which give the following conditions:

$$\frac{\omega_{T1}L_s}{\operatorname{Re}[\alpha_M(j\omega_{0,\alpha_M})]} \simeq R_s, \tag{3.9}$$

$$\omega_0 \approx \frac{1}{\sqrt{L_g C_{gs1} \cdot \operatorname{Re}[\alpha_M(j\omega_{0,\alpha_M})]}},\tag{3.10}$$

$$\omega_{0,\alpha_M} \approx \frac{1}{\sqrt{L_I(C_I + n^2 k^2 C_L)}},\tag{3.11}$$



Figure 3.4: Calculated (a) α_M , (b) Z_{in} , and (c) S_{11} with k as a parameter.

respectively, where $\operatorname{Re}[\alpha_M(j\omega_{0,\alpha_M})]$ is approximated as

$$\operatorname{Re}[\alpha_M(j\omega_{0,\alpha_M})] \approx 1 + \alpha_{gd1} + \frac{\alpha_{gd1}g_{m1}}{(1-nk)g_{m2}}.$$
(3.12)

Equations (3.9)–(3.12) show that ω_0 and ω_{0,α_M} decrease and $\operatorname{Re}[\alpha_M]$ increases as k increases. Figure 3.4(c) shows the calculated S_{11} with k as a parameter. Input impedance matching is achieved around ω_0 , which decreases with increasing k.

3.3.2 Gain

The magnetic coupling reduces the peak frequency and magnitude of the LNA gain. The common-gate stage acts as a transimpedance, which converts the input signal current i_i to the output voltage v_{out} , as shown in Fig. 3.3. The input current i_i amplified by the first stage is derived from Fig. 3.2:

$$i_i \approx -\frac{g_{m1}}{j\omega_0 C_{gs1} R_s \cdot \operatorname{Re}[\alpha_M]} v_{in}, \qquad (3.13)$$

where input impedance matching is assumed (i.e., $Z_{in} = R_s$) and v_{in} represents the input voltage of the LNA as shown in Fig. 3.2. The transimpedance from node I to the output is given by

$$Z_{T} = \frac{v_{out}}{i_{i}} = \frac{v_{out}}{v_{i}} \frac{v_{i}}{i_{i}}$$

$$= \frac{nk + g_{m2}[R_{L} + j\omega(1 - k^{2})L_{L}]}{1 + j\omega C_{L}[R_{L} + j\omega(1 - k^{2})L_{L}]} \cdot \frac{1}{Y_{I}} = \frac{\frac{nk}{R_{L} + j\omega(1 - k^{2})L_{L}} + g_{m2}}{D}, \quad (3.14)$$

$$D = \left(j\omega C_{I} + \frac{1}{j\omega L_{I} + R_{I}}\right) \left(j\omega C_{L} + \frac{1}{j\omega(1 - k^{2})L_{L} + R_{L}}\right)$$

$$+ g_{m2} \left(j\omega C_{L} + \frac{1 - nk}{j\omega(1 - k^{2})L_{L} + R_{L}}\right) + \frac{j\omega n^{2}k^{2}C_{L}}{j\omega(1 - k^{2})L_{L} + R_{L}}. \quad (3.15)$$

Around $\omega = 1/\sqrt{L_I C_I} = 1/\sqrt{L_L C_L}$, the first term in Eq. (3.15) is approximated by zero:

$$D \approx g_{m2} \left(j\omega C_L + \frac{1 - nk}{j\omega(1 - k^2)L_L} \right) + \frac{n^2 k^2 C_L}{(1 - k^2)L_L},$$
(3.16)

where R_I and R_L are ignored for simplicity. The magnitude of Z_T becomes a maximum when the first term in Eq. (3.16) equals zero. The voltage gain of the LNA and its peak frequency can be therefore expressed as

$$A_{v,LNA} = \left| \frac{v_{out}}{v_{in}} \right| = \left| \frac{i_i Z_T}{v_{in}} \right| = \left| \frac{g_{m1} Z_T}{j \omega_0 C_{gs1} R_s \cdot \operatorname{Re}[\alpha_M]} \right|,$$
(3.17)

$$\omega_p \approx \frac{1}{\sqrt{\frac{1-k^2}{1-nk}L_LC_L}},\tag{3.18}$$

respectively. Equations (3.17) and (3.18) show that the magnitude and peak frequency of the voltage gain decrease with increasing k. Figure 3.5 shows the calculated $A_{v,LNA}$ with k as a parameter. The increase of k shifts the gain peak toward a lower frequency and reduces the gain magnitude. The peak frequency in Fig. 3.5 corresponds well to that calculated from Eq. (3.18).

The LNA with the magnetic coupling sacrifices a maximum voltage gain to achieve the target peak frequency, $\omega_{p,t}$. Equation (3.18) gives the following condition:

$$L_L = L_I = \frac{1}{\omega_{p,t}^2 (1+k)C_L}$$
 (for $n = 1$). (3.19)

A turns ratio of one provides the smallest chip area of the transformer, because L_L and L_I simultaneously decrease with increasing k. Equation (3.19) shows that a smaller L_I and L_L are required to achieve $\omega_{p,t}$ as k increases. Reducing L_I and L_L leads to a smaller chip area, but to a decrease in the parallel impedances of the internal and load LC tanks at the resonance frequencies $(Z_p \approx (\omega_0 L_{I,L})^2/R_{I,L})$, causing a lower voltage gain. Figure 3.6 shows the calculated $A_{v,LNA}$ for $f_{p,t} = 5.0$ GHz where L_L and L_I satisfy Eq. (3.19). A peak frequency of approximately 5.0 GHz can be achieved even for a large k, while the maximum gain decreases with increasing k $(A_{v,LNA} \propto (1 + k)^{-2})$. However, a small coupling factor such as 0.2 is acceptable for the LNA, due to a small gain reduction of 3 dB.

3.3.3 Noise

The transformer reduces the output noise originating from the common-gate transistor and the parasitic resistance of L_L , thereby improving the noise performance. Figure 3.7 conceptually illustrates how the transformer reduces the drain noise current of M_2 , represented by i_{nd2} . The primary (internal) inductor L_I detects i_{nd2} , and then induces a noise voltage to the secondary (load) inductor L_L . The induced noise voltage is correlated and anti-phase to the output noise voltage produced by i_{nd2} flowing through L_L , reducing the output noise caused by M_2 . The other output noise originating from L_L is also reduced by the transformer in the same way.

The magnetic coupling affects the noise contributions from M_2 , L_I , and L_L to the LNA (F_{M_2} , F_{L_I} , and F_{L_L} , respectively), but not that from M_1 (F_{M_1}). The LNA noise factor is given by

$$F = 1 + F_{M_1} + F_{M_2} + F_{L_I} + F_{L_L}, (3.20)$$

$$F_{M_1} \approx \frac{\gamma_1 \chi_1}{\alpha_1} g_{m1} R_s \left(\frac{\omega_0}{\omega_{T_1}}\right)^2 + \frac{\alpha_1 \delta_1}{\kappa_1 g_{m1} R_s},\tag{3.21}$$

$$\chi_1 = (1 + \alpha_{gd1})^2 - 2|c|\alpha_1 \sqrt{\frac{\delta_1}{\kappa_1 \gamma_1}} (1 + \alpha_{gd1}) + \frac{\delta_1 \alpha_1^2}{\kappa_1 \gamma_1}, \qquad (3.22)$$

$$F_{M_2} \approx 4 \left| \frac{Y_I}{Y_0 + Y_I} \right|^2 \frac{\gamma_2}{\alpha_2} g_{m2} R_s \left(\frac{\omega_0}{\omega_{T_1}} \right)^2 |\alpha_M|^2 \left| \frac{(1-k) \left(1 - j \frac{Y_0 + Y_{L_I C_I}}{\omega_0 C_L} \right)}{k + j \frac{g_{m2}(1-k)}{\omega_0 C_L}} \right|^2,$$
(3.23)



Figure 3.5: Calculated voltage gain with k as a parameter.



Figure 3.6: Calculated voltage gain with k as a parameter for $L_L = L_I = 1/\omega_{p,t}^2(1+k)C_L$.



Figure 3.7: Mechanisms for noise reduction of i_{nd2} .

$$F_{L_{I}} \approx 4 \left| \frac{Y_{I}}{Y_{0} + Y_{I}} \right|^{2} R_{I} R_{s} \left(\frac{\omega_{0}}{\omega_{T_{1}}} \right)^{2} |\alpha_{M}|^{2} \left| \frac{(1 - k)g_{m2} - k\left(Y_{0} + j\omega_{0}C_{I}\right)}{k + j\frac{g_{m2}(1 - k)}{\omega_{0}C_{L}}} \right|^{2},$$
(3.24)

$$F_{L_L} \approx 4 \left| \frac{Y_I}{Y_0 + Y_I} \right|^2 R_L R_s \left(\frac{\omega_0}{\omega_{T_1}} \right)^2 |\alpha_M|^2 \left| \frac{(1-k)g_{m2} + Y_0 + Y_{L_I C_I}}{k + j \frac{g_{m2}(1-k)}{\omega_0 C_L}} \right|^2, \quad (3.25)$$

where $\alpha_i = g_{mi}/g_{d0i}$ and g_{d0i} is the zero bias drain conductance of M_i (i = 1, 2); γ_i and δ_i are the drain noise current factor and the induced gate noise current factor, respectively, and c is the correlation coefficient between these noise currents ($\simeq j0.395$ [9]); κ_i is the Elmore constant (=5 [10]); $L_I = L_L = 1/\omega_0^2(1+k)C_L$; Y_0 represents the output admittance of the input stage at node I and is approximated as $j\omega C_{gd1}$; $Y_{L_IC_I} = j\omega C_I + 1/(j\omega L_I + R_I)$. The detailed derivations are summarized in Appendix A.2. Equations (3.21)–(3.25) show that F_{M_1} is independent of kwhile F_{M_2} , F_{L_I} , and F_{L_L} are functions of k. Figure 3.8 shows the calculated F_{M_2} , F_{L_I} , and F_{L_L} versus k. As Eq. (3.23) shows, F_{M_2} approaches zero with increasing k. Meanwhile, F_{L_I} increases and F_{L_L} slightly decreases. This difference originates from the different numerators in Eqs. (3.24) and (3.25), i.e., $-k (Y_0 + j\omega_0 C_I)$ and $Y_0 + Y_{L_IC_I}$.

The noise improvement by the transformer is limited in the folded-cascode topology. The calculated noise figure (NF), defined by $10 \log F$ versus k are shown in Fig. 3.9, where 90 nm CMOS process parameters are used. The NF simulated using Agilent Advanced Design System (ADS) are also plotted. Figure 3.9 shows that the calculated NF is comparable to the simulated NF, and the magnetic coupling reduces the NF by up to 0.08 dB (calculated) or 0.12 dB (simulated) for $g_{m2} = 15$ mS. The amount of noise reduction is relatively small, because the noise of M_1 is the dominant noise source in the LNA ($F_{M_1} \simeq 1.20$ and 1.05 in the calculations and simulations, respectively).



Figure 3.8: Calculated F_{M_2} , F_{R_I} , and F_{R_L} versus k.



Figure 3.9: Calculated and simulated NF versus k.

3.3.4 Stability

A small L_I or large C_L ensures the stability of the LNA. The proposed LNA becomes potentially unstable, because the transformer provides a positive feedback from the output to node I, as shown in Fig. 3.1. For stability, the LNA must satisfy the following condition [11]:

$$\operatorname{Re}[Z_{in}] > 0. \tag{3.26}$$

For k = 1 and low frequencies (the worst case), $\text{Re}[Z_{in}]$ is approximated as (Eq. (3.1) for $Y_I = 1/j\omega L_I$)

$$\operatorname{Re}[Z_{in}] \approx \frac{\omega_{T1}[L_s(1+\alpha_{gd1}) - L_I\alpha_{gd1}]}{(1+\alpha_{gd1})^2 + \omega^2 L_I^2 \alpha_{gd1}^2 g_{m1}^2}.$$
(3.27)

Substituting Eq. (3.27) in Eq. (3.26), we have

$$L_I < L_s \left(1 + \frac{1}{\alpha_{gd1}} \right), \tag{3.28}$$

which shows that a smaller L_I ensures the stability. Using Eq. (3.19), we can rewrite the above condition as

$$C_L > \frac{1}{2\omega_{p,t}^2 L_s \left(1 + \frac{1}{\alpha_{gd1}}\right)}.$$
(3.29)

For example, $C_L > 160$ fF is calculated from $f_{p,t} = 5$ GHz, $L_s = 0.6$ nH, and $\alpha_{gd1} = 0.2$. This capacitance value can be satisfied with the parasitic capacitances of L_L and the input capacitance of the following stage.

3.4 Design

3.4.1 Transistors

The input transistor M_1 is designed to achieve a minimum NF at 5 GHz with a bias current I_{d1} of 1.0 mA at a supply voltage of 0.5 V. Equations (3.21) and (3.22) provide an optimum (for noise performance) gate width for M_1 of $4 \times 40 \ \mu$ m (40 gate fingers, each with a unit of $4 \ \mu$ m width) and a minimum gate length of 100 nm. Although the calculated minimum NF is 3.6 dB for $I_{d1} = 1.0$ mA, increasing I_{d1} leads to a lower NF (i.e., 2.2 dB for $I_{d1} = 2.0$ mA).

The size of the common-gate transistor M_2 is selected as a compromise between noise and linearity performance. For a fixed bias current of 1.0 mA, a small gate width of M_2 provides high linearity as shown in Chapter 2, but leads to a lower g_{m2} , which results in the increase of F_{M_2} , F_{L_1} , and F_{L_L} as shown in Eqs. (3.23)–(3.25). Figure 3.9 shows less NF degradation for $g_{m2} > 15$ mS than for $g_{m2} < 15$ mS at a low k. Thus, g_{m2} is selected to be approximately 15 mS, which results in a gate width of $4 \times 40 \ \mu$ m and gate length of 100 nm.

3.4.2 Transformer and Inductor

A partially-coupled transformer, shown in Fig. 3.10(a), allows us to simultaneously achieve a small chip area $(0.314 \times 0.200 \text{ mm}^2)$ and reduce the magnetic coupling $(k \simeq 0.1)$. On the other hand, a stacked transformer, shown in Fig. 3.10(b), provides a smaller chip area $(0.210 \times 0.200 \text{ mm}^2)$, thereby reducing the cost. However, a large k of the stacked transformer $(k \simeq 0.9)$ leads to poor gain, and does not significantly reduce the NF of the LNA, as shown in Section 3.3.3. Figure 3.11 shows the ADS simulated voltage gain and NF of the LNAs employing the transformers shown in Figs. 3.10(a) and (b). The transformers were designed using a three-dimensional (3-D) electromagnetic simulator (Ansoft HFSS), and L_s and L_g in both the LNAs were adjusted to achieve an S_{11} of less than -10 dB at 5 GHz. The LNA with the stacked transformer had 13 dB lower gain than the LNA with the partially-coupled transformer at 5 GHz. This leads to an increase in the overall NF of the receiver. A larger k also leads to an increase in the voltage swing at node I (Fig. 3.1), causing poor reverse isolation performance (S_{12}) of the LNA with the stacked transformer.

The inductances of the transformer are selected to resonate at a frequency of approximately 5 GHz. The outer diameter of each inductor is 200 μ m, the metal width 7 μ m, and the metal spacing 2 μ m. Electromagnetic simulations resulted in $L_I = L_L = 3.6$ nH and quality factors (Q) of 6.7 at 5 GHz.

The inductances of L_s and L_g are determined by the input impedance matching conditions, derived from Eqs. (3.6)–(3.8): $L_s \simeq 0.8$ nH and $L_g \simeq 4.3$ nH. The outer diameter of L_g is 200 μ m, the metal width 5 μ m, and metal spacing 2 μ m. The simulated Q of L_g was 7.5 at 5 GHz.

3.5 Experimental Results and Discussion

The designed LNA with the partially-coupled transformer shown in Fig. 3.12 was fabricated in a 90 nm digital CMOS process with seven metal layers including a 1.9- μ m thick metal layer and without metal-insulator-metal (MIM) capacitors. For comparison, a conventional folded-cascode LNA with the same device sizes except the magnetic coupling as in Fig. 3.12 was also fabricated on the same chip. A micrograph of the fabricated LNAs is shown in Fig. 3.13. The active chip areas (without the pads) of the proposed and conventional LNAs were $0.39 \times 0.55 \text{ mm}^2$ and $0.52 \times 0.55 \text{ mm}^2$, respectively. The input and output pads were not electrostatic-discharge (ESD) protected. Metal fills consisting of metal 1–6 layers were placed both inside and outside the fabricated transformer and inductors to meet metal density rules in the CMOS process. They were 1.5 μ m by 1.5 μ m squares with a spacing of 0.2 μ m. For the measurements, a unity-gain common-source amplifier with a 50 Ω output resistor was used as a buffer, shown in Fig. 3.12. The S-parameters, noise, and linearity of the LNAs were 1.0 mW and 1.8 mW at a supply voltage of 0.5 V, respectively.



Figure 3.10: Layout of (a) a partially-coupled transformer and (b) a stacked transformer.



Figure 3.11: Simulated voltage gain and NF of LNAs employing the partially-coupled transformer (solid line) and stacked transformer (dashed line).



Figure 3.12: Complete schematic of the designed transformer folded-cascode LNA.



Figure 3.13: Micrograph of the proposed LNA (left) and conventional folded-cascode LNA (right).

3.5.1 S-parameters

Figures 3.14 and 3.15 show the measured and simulated S_{11} and S_{21} of the fabricated LNAs, respectively. The S-parameters were measured using an Agilent Technologies HP8722ES network analyzer. The proposed LNA obtained an S_{11} of less than -10 dB around 5 GHz and a maximum S_{21} of 16.8 dB at 4.7 GHz. The magnetic coupling in the proposed LNA had a small impact on the S_{11} performance, while the measured peak of S_{21} was shifted to a lower frequency than the simulated one, due to the increased magnetic coupling of the fabricated transformer ($k \simeq 0.2$). This frequency shift can be reduced by using a smaller L_I and L_L (3.4 nH).

The discrepancy between the measured and simulated S_{21} is mainly attributed to insufficient accuracy in the simulation of the inductors used. The HFSS simulation models of the transformer and inductors included no metal fills to solve convergence problems and reduce the memory requirement. The metal fills decrease the quality factors of the transformer and inductor [12–14], which results in a lower gain.

Figures 3.16 and 3.17 show the measured and simulated S_{12} and S_{22} of the LNAs with the buffers, respectively. The propose LNA with the buffer achieved an S_{12} of -47 dB at 5.0 GHz, while the stand-alone buffer obtained an S_{12} of -30 dB at 5.0 GHz (not shown). Thus, the S_{12} of the proposed LNA without the buffer was approximately -17 dB. Figure 3.16 also shows that the inductor coupling deteriorates the reverse isolation by a factor of 12 dB at 5 GHz, compared to the conventional LNA. This deterioration is not problematic, because the proposed LNA still has good isolation, due to the folded-cascode topology. Both the LNAs achieved an S_{22} of less than -10 dB around 5.0 GHz as shown in Fig. 3.17.

3.5.2 NF

Figure 3.18 shows the measured and simulated NF of the LNAs. The NF was measured using an Agilent Technologies HP8970B noise figure meter. The proposed LNA obtained a minimum NF of 3.9 dB at 4.7 GHz, while the conventional LNA achieved a minimum NF of 4.1 dB at the same frequency. The difference between the measured minimum NF can be attributed to more input-referred noise of the buffer in the conventional LNA than that in the proposed LNA. The LNAs had different values of S_{21} at 4.7 GHz, resulting in different input-referred noise of the buffer.

3.5.3 Linearity

Figure 3.19 shows the measured output power of the fundamental tone and third-order intermodulation (IM₃) products for two tones (4.999 GHz and 5.000 GHz), applied to the LNA. The two tones were generated by Agilent Technologies HP8671B and E4438C signal generators, and the fundamental and IM₃ tones were measured using an Agilent Technologies E4448A spectrum analyzer. The measured *IIP*₃ of the proposed LNA with the buffer was -18.5 dBm, and that of the stand-alone buffer was -0.25 dBm (not shown). The *IIP*₃ of the LNA without the buffer can be calculated from Eq. (2.49). Substituting *IIP*_{3,LNA+Buf} = -18.5 dBm, *IIP*_{3,Buf} = -0.25 dBm



Figure 3.14: Measured and simulated S_{11} of the LNAs.



Figure 3.15: Measured and simulated S_{21} of the LNAs.



Figure 3.16: Measured and simulated S_{12} of the LNAs.



Figure 3.17: Measured and simulated S_{22} of the LNAs.



Figure 3.18: Measured and simulated NF of the LNAs.



Figure 3.19: Measured IIP_3 of the proposed LNA with the buffer.

and $A_{v,LNA} = 15.8 \text{ dB}$ (at 5.0 GHz) into Eq. (2.49) gives $IIP_{3,LNA} = -14.8 \text{ dBm}$. The IIP_3 of the conventional LNA was also -14.8 dBm.

3.5.4 Comparison

Table 3.1 shows a summary of the LNA performance and a comparison with previously reported low-voltage (~0.6 V) CMOS LNAs for 5 GHz applications. The proposed LNA achieved performance comparable to the conventional folded-cascode LNA, while consuming three fourths of the chip area of the conventional LNA. The figure of merits for the LNAs, FoM_1 and FoM_2 , included in Table 3.1, are defined by Eqs. (2.50) and (2.51), respectively. The proposed LNA obtained the best FoM_1 (4.8 mW⁻¹) with the smallest chip area among the reported low-voltage CMOS LNAs, whereas it achieved a lower FoM_2 than that of the LNA reported in [3], due to a lower IIP_3 .

3.6 Conclusion

This chapter has demonstrated a transformer folded-cascode CMOS LNA, in which the internal and load inductors have been magnetically coupled to reduce the chip area. Circuit analysis showed that the magnetic coupling between these inductors decreases the resonance frequency of the input matching network, the peak frequency and magnitude of the gain, and the noise figure. The partially-coupled transformer reduced the chip area, while having a small impact on the LNA performance. The LNA implemented with a 90 nm CMOS technology occupied 0.21 mm² and achieved an S_{11} of -14 dB, NF of 3.9 dB, and voltage gain of 16.8 dB at 4.7 GHz with a power consumption of 1.0 mW from a 0.5 V supply. The chip size of the proposed LNA was 25% smaller than that of the conventional folded-cascode LNA. It has been demonstrated that the proposed LNA can replace conventional low-voltage CMOS LNAs.

I_2	+	2					
FoN[$-$]	0.74	0.67	2.3]	N/A	N/A	0.22	0.22
FoM_1 $[\mathrm{mW}^{-1}]$	4.8	4.1	2.2	0.2	1.4	1.8	2.3
Area [mm ²]	0.21	0.29	0.30	0.40	0.75	0.54	0.46
Power [mW]	1.0	1.0	1.0	12.5	1.0	0.9	1.68
Supply [V]	0.5	0.5	0.6	0.7	0.4	0.6	0.6
P:m-1 dB [dBm]	-27	-27	-15.8	6-	-22	-27	-25
IIP_3 [dBm]	-14.8	-14.8	-7.25	N/A	N/A	-16	-17.1
S_{21} [dB]	16.8	16.1	9.2	7.0	10.3	9.2	14.1
[dB]	3.9	4.1	3.6	2.9	5.3	4.5	3.65
Frequency [GHz]	4.7	5.0	5.5	5.8	5.1	5.0	5.0
CMOS Technology	90 nm	90 nm	90 nm	180 nm	130 nm	180 nm	180 nm
Reference	This work	Folded-cascode	[3]	[4]	[5]	[9]	[15]

Table 3.1: Measured performance and comparison of low-voltage CMOS LNAs.

Bibliography

- [1] International technology roadmap for semiconductors 2006 update. [Online]. Available: http://www.itrs.net
- [2] H. Tsuchikawa, M. Takakuwa, and S. Sugatani, "Electron beam direct writing technology combined with silicon shuttle service," in *4th ISMI Symp. on Manufacturing Effectiveness*, Austin, TX, Oct. 2007.
- [3] D. Linten, L. Aspemyr, W. Jeamsaksiri, J. Ramos, A. Mercha, S. Jenei, S. Thijs, R. Garcia, H. Jacobsson, P. Wambacq, S. Donnay, and S. Decoutere, "Low-power 5 GHz LNA and VCO in 90 nm RF CMOS," in *Symp. VLSI Circuits Dig. Tech. Papers*, Honolulu, HI, Jun. 2004, pp. 372–375.
- [4] T. K. K. Tsang and M. N. El-Gamal, "Gain and frequency controllable sub-1 V 5.8 GHz CMOS LNA," in *Proc. IEEE Int. Symp. Circuits and Systems*, vol. 4, Scottsdale, AZ, May 2002, pp. 795–798.
- [5] D. Wu, R. Huang, W. Wong, and Y. Wang, "A 0.4-V low noise amplifier using forward body bias technology for 5 GHz application," *IEEE Microw. Wireless Compon. Lett.*, vol. 17, no. 7, pp. 543–545, Jul. 2007.
- [6] H.-H. Hsieh and L.-H. Lu, "A CMOS 5-GHz micro-power LNA," in *IEEE Radio Frequency Integrated Circuits Symp. Dig. Papers*, Long Beach, CA, Jun. 2005, pp. 31–34.
- [7] T. Kihara, H.-J. Park, I. Takobe, F. Yamashita, T. Matsuoka, and K. Taniguchi, "A 0.5 V area-efficient transformer folded-cascode low-noise amplifier in 90 nm CMOS," in *Proc. IEEE Int. Conf. on Integrated Circuit Design and Technology*, Grenoble, France, Jun. 2008, pp. 21–24.
- [8] D. K. Shaeffer and T. H. Lee, "A 1.5 V, 1.5 GHz CMOS low noise amplifier," *IEEE J. Solid-State Circuits*, vol. 32, no. 5, pp. 745–759, May 1997.
- [9] A. van der Ziel, *Noise in Solid State Devices and Circuits*. Toronto: John Wiley & Sons, 1986.
- [10] Y. Tsividis, *Operation and Modeling of the Mos Transistor*, 2nd ed. New York: Oxford University Press, 2003.
- [11] G. Gonzalez, *Microwave Transistor Amplifiers*, 2nd ed. Upper Saddle River, NJ: Prentice Hall, 1997.
- [12] J.-H. Chang, Y.-S. Youn, H.-K. Yu, and C.-K. Kim, "Effects of dummy patterns and substrate on spiral inductors forsub-micron RF ICs," in *IEEE Radio Frequency Integrated Circuits Symp. Dig. Papers*, Seattle, WA, Jun. 2002, pp. 419–422.
- [13] F. Zhang and P. R. Kinget, "Design of components and circuits underneath integrated inductors," *IEEE J. Solid-State Circuits*, vol. 41, no. 10, pp. 2265–2271, Oct. 2006.
- [14] L. Nan, K. Mouthaan, Y.-Z. Xiong, J. Shi, S. C. Rustagi, and B.-L. Ooi, "Experimental characterization of the effect of metal dummy fills on spiral inductors," in *IEEE Radio Frequency Integrated Circuits Symp. Dig. Papers*, Honolulu, HI, Jun. 2007, pp. 307–310.
- [15] H.-H. Hsieh, J.-H. Wang, and L.-H. Lu, "Gain-enhancement techniques for CMOS folded cascode lnas at low-voltage operations," *IEEE Trans. Microw. Theory Tech.*, vol. 56, no. 8, pp. 1807–1816, Aug. 2008.

Chapter 4

Transformer Noise-Canceling UWB CMOS LNA

4.1 Introduction

The ultra-wideband (UWB) technology has attracted much interest in recent years, because of its ability to realize high-speed wireless personal area networks (WPANs), in which electronic devices are required to transfer large amounts of data, such as audio or video files, at a high data transfer rate. UWB frequency bands assigned from 3.1 to 10.6 GHz (Fig. 4.1) are utilized by two different communication systems: multiband orthogonal frequency division multiplexing (MB-OFDM) UWB [1] or single-carrier direct sequence (DS) UWB [2]. The MB-OFDM UWB system using 14 sub-bands, each with a bandwidth of 528 MHz, transmits signals modulated by OFDM in the subband. The data rate is up to 480 Mbps. The DS-UWB system spreads the spectrum over the low band (3.1–4.85 GHz) or high band (6.2–9.7 GHz), and provides a maximum data rate of 1320 Mbps. In either case, wideband low-noise amplifiers (LNAs) are essential for the RF front-ends of UWB receivers.



Figure 4.1: UWB frequency bands

The UWB LNA must meet several stringent requirements: input impedance matching, low noise performance, and sufficient gain across 3.1–10.6 GHz at low power consumption, low supply voltage, and low cost (i.e., small area and requiring no additional layers). In addition, it is desirable to implement the LNA with digital CMOS technologies for the integration of RF front-ends and digital circuits. Although several wideband CMOS LNAs have been proposed in recent years, none of them have simultaneously met all these requirements. An LNA with wideband LC matching networks [3] consumes a large chip area. Although resistive-feedback LNAs [4–6] and common-drain feedback LNAs [7, 8] occupy small chip areas, they require high power consumption and high supply voltages to simultaneously achieve wideband input impedance matching and low noise performance. A reactive-feedback LNAs [9, 10] demands two thick metal layers to form a transformer that provides a reactive feedback. Noise-canceling LNAs [11–14] require additional circuits and power consumption. Distributed LNAs [15, 16] consume much higher power and larger areas than other LNAs.

This chapter proposes a transformer noise-canceling common-gate LNA employing an output series inductor [17]. The proposed LNA is suitable for low-power and low-voltage operation, and achieves $|S_{11}| < -10$ dB, NF < 4.4 dB, and $|S_{21}| > 9.3$ dB across 3.1–10.6 GHz. This chapter is organized as follows. Section 4.2 describes previously proposed wideband CMOS LNAs and their drawbacks. Section 4.3 shows the proposed circuit topology and the noise cancellation mechanisms. The noise, input admittance, gain, stability, and group delay of the proposed LNA are analyzed in Section 4.4. Section 4.5 describes the design methodology for the LNA. Section 4.6 shows the measurements of the LNA implemented in a 90 nm digital CMOS process, and Section 4.7 concludes the chapter.

4.2 Wideband CMOS LNAs

Wideband CMOS LNAs can be generally categorized into two types: common-source (CS) LNAs, and common-gate (CG) LNAs. LC matching networks or feedback techniques allow the CS LNAs to achieve wideband input bandwidth. Feedback techniques applied to CS LNAs can be divided into three categories: resistive feedback, reactive feedback, and common-drain feedback. A gm-boosting or noise-canceling technique is employed to CG LNAs for noise reduction.

4.2.1 Common-Source LNAs

Input Matching Network

Input LC matching networks based on Chebyshev or Butterworth configurations are used for CS LNAs to achieve wideband impedance matching [3]. However, they require several high-Q inductors: four inductors in an LNA employing the input matching network based on the fifth-order Chebyshev filter [3]. The use of many inductors increases the chip area and the parasitic resistances and capacitances, causing noise and gain degradation at high frequencies.

Resistive Feedback

Resistive-feedback LNAs require high power consumption and high supply voltages to simultaneously achieve wideband impedance matching and a low NF. Figure 4.2(a) shows the basic topology of the LNA, whose input impedance is expressed as

$$Z_{in} \approx \frac{1}{sC_p} / \frac{R_f}{1 + A_v},\tag{4.1}$$

where C_p is the input parasitic capacitance, R_f the feedback resistance, and A_v the voltage gain of the LNA. Increasing R_f reduces the NF of the LNA [4], but A_v must also be increased accordingly to obtain the desired input impedance (50 Ω) as shown in Eq. (4.1). This requires high current consumption and a large load resistor and transistor, which results in a reduction of the input bandwidth. Even with a state-of-the-art CMOS technology, the resistive-feedback LNA [5] demanded 12 mW from a 1.8-V supply to achieve NF < 2.6 dB; the input bandwidth was limited to less than 5.0 GHz.

Common-Drain Feedback

A common-drain (CD) stage shown in Fig. 4.2(b) detects the output voltage and feeds part of the voltage back to the input, thereby producing a resistive component for the input impedance:

$$Z_{in} \approx \frac{1}{g_{m3}(1 + g_{m1}R_L)}$$
(4.2)

where g_{m3} is the transconductance of the CD transistor M_3 , and R_L is the load resistance. However, the extra CD and current source transistors increase the noise and input parasitic capacitance. Hence, the topology also has difficulties in achieving wideband impedance matching and low noise performance at simultaneously low power consumption ($|S_{11}| < -10$ dB and NF < 4.3 dB in the frequency range of 0–6 GHz at 3.4 mW [7]).

Reactive Feedback

The input stage of the reactive-feedback LNA [9, 10] is shown in Fig. 4.2(c). The transformer connected to the gate and source of M_1 detects the source current, returning part of it to the gate. Although this topology provides $|S_{11}| < -10$ dB and NF < 3 dB across 3.1–10.6 GHz [9, 10], it requires a transformer with very low parasitic resistances. The input impedance of the LNA is given by [10]

$$Z_{in} \approx \frac{1}{\left(1+\beta\right)\beta} \left(r_{pri} + \frac{1}{g_{m1}}\right),\tag{4.3}$$

where β is the feedback factor and equals k/n, and k and n are the magnetic coupling factor and the turns ratio of the transformer, respectively; r_{pri} is the parasitic resistance of the primary inductor L_p ; g_{m1} is the transconductance of M_1 . For $\beta = 0.19$ [10] and $g_{m1} = 50$ mS, the upper limit on r_{pri} for $S_{11} < -10$ dB is calculated as 1.7 Ω . It is difficult to implement such



Figure 4.2: Wideband CMOS LNAs.

a primary inductor using a lower thin metal layer¹. Increasing β and g_m alleviates the above limitation but leads to an increase in NF [10] and high power consumption, respectively. In fact, the transformer in [9, 10] consisted of two thick metal layers (3–4 μ m). An additional thick metal layer increases the fabricating cost.

4.2.2 Common-Gate LNAs

The CG LNA (Fig. 4.2(d)) is suitable for wideband and low-voltage operation, because it has a low quality factor of the input network and does not require a cascode transistor to alleviate

¹The secondary inductor, stacked on the primary one, must be implemented with a top thick metal layer for low noise performance [10].

the Miller effect from the CG transistor M_1 [18]. The main drawback of the LNA is poor noise performance. The noise factor is given by

$$F \approx 1 + \gamma + \frac{4R_s}{R_L},\tag{4.4}$$

where R_s is the input signal source resistance. The coefficient of the channel thermal noise in the MOSFET, γ , equals 2/3 in long-channel MOSFETs, but exceeds this value in short-channel MOSFETs [19–21]. For $\gamma = 2$, $R_L = 200 \Omega$, and $R_s = 50 \Omega$, the NF is calculated as approximately 6.0 dB, which is unacceptable for wideband LNAs.

Gm-Boosting Technique

A gm-boosting technique [18] reduces the noise factor contributed from M_1 by a factor of (1+A):

$$F \approx 1 + \frac{\gamma}{1+A} + \frac{4R_s}{R_L},\tag{4.5}$$

where A, shown in Fig. 4.2(e), is an inverting gain and assumed to be noiseless. An additional circuit providing A is required for practical use of the technique. A differential capacitor crosscoupling topology [22, 23] provides $A \simeq 1$ ($NF \simeq 4.8$ dB), but requires an external wideband balun. A transformer-coupled topology [24], which provides A > 1, is not suitable for wideband operation, due to a larger effective C_{gs} , which becomes more than four times as large as that of the CG LNA.

Noise-Canceling Technique

A noise-canceling technique based on [25] has also been applied to the CG LNA for noise reduction [11–13]. This technique cancels the noise of the CG transistor using an additional CS one (Fig. 4.2(f)). However, the noise of the CS transistor is not canceled, and hence the NF of this LNA is less than or comparable to that of the CG LNA, even with higher power consumption and a larger chip area.

4.3 Transformer Noise-Canceling LNA

This section presents the circuit topology of the transformer noise-canceling LNA and the noise cancellation mechanisms.

4.3.1 Circuit Topology

Figure 4.3 shows a schematic of the proposed LNA, based on a CG LNA with a shunt-peaking inductor. The main difference between the proposed LNA and the CG LNA is that the input and shunt-peaking inductors, L_p and L_s , are magnetically coupled to form a transformer. A similar



Figure 4.3: Schematic of the proposed LNA.

topology has been reported for narrowband applications [26]. The transformer reduces the noise of M_1 and the load resistor R_L , thereby improving the noise performance without additional circuits or increased power consumption. The transformer also provides a positive feedback, whose mechanism is as follows: An output current generated by a signal voltage flows through L_s , which induces a voltage that is in phase with the signal voltage to L_p . The output series inductor L_1 forms a π network with the parasitic capacitances, C_1 and C_2 , extending not only the gain bandwidth, but also the input bandwidth. The chip area of the proposed LNA is the same as that of the CG LNA with the shunt-peaking inductor, because L_p can be stacked on L_s , i.e., a stacked transformer, which occupies the area of one inductor.

4.3.2 Noise Cancellation

The transformer partly cancels the output noise originating from the CG transistor M_1 and load resistor R_L , thereby improving the LNA noise performance. The small-signal circuit of the proposed LNA are shown in Fig. 4.4, where the voltage supply terminal (V_{DD}) is connected to the AC ground; the noise of the signal source resistance R_s , M_1 , and R_L are represented by the noise current sources i_{ns} , i_{nd} , and noise voltage source v_{nR_L} , respectively; M, given by $k\sqrt{L_pL_s}$, is the mutual inductance of the transformer and k the magnetic coupling factor; C_p represents the sum of the gate-source capacitance of M_1 and the parasitic capacitances of the input pad and L_p ; Z_L is the load impedance considering the right hand side of output node A. The mechanisms for the noise cancellation are conceptually illustrated in Figs. 4.5(a) and (b). The transformer detects noise currents flowing through the primary (or secondary) inductor L_p , inducing voltages correlated with the currents to the secondary (or primary) inductor L_s .



Figure 4.4: Small-signal equivalent circuit with noise sources.

Transistor Noise Cancellation

The output noise voltage generated by i_{nd} is partly canceled by the induced noise voltage originating from i_{nd} flowing through L_p , as shown in Fig. 4.5(a). The noise current i_{nd} first flowing through L_s and R_L generates a noise voltage $v_{n1} = -i_{nd}(R_L + sL_s)$, and then a noise voltage $v'_{n1} = -i_{nd} \cdot sM$ is induced to L_p . Next, i_{nd} flows through L_p , producing a noise voltage $v_{n2} = i_{nd} \cdot sL_p$, which is canceled by v'_{n1} . Here, the transformer induces a noise voltage $v'_{n2} = i_{nd} \cdot sM$ to L_s . The induced noise voltage v'_{n2} is correlated and in antiphase with v_{n1} and hence the total output noise voltage is reduced: $-i_{nd}(R_L + sL_s - sM)$. The expression of the output noise voltage (at node A) including the effect of Z_L can be derived from Fig. 4.4:

$$v_{out,i_{nd}} = -Z_L \frac{n^2 k^2 - nk + \left(\frac{1}{R_s} + sC_p + \frac{1}{sL_p}\right) (R_L + s(1 - k^2)L_s)}{\left(\frac{1}{R_s} + Y_{IN}\right) (Z_L + R_L + s(1 - k^2)L_s)},$$
(4.6)

where $n = \sqrt{L_s/L_p}$ is the turn ratio of the transformer; $Y_{IN} = i_{in}/v_{in}$, described in the next section, is the input admittance of the LNA, and i_{in} and v_{in} are the input current and voltage, respectively, as shown in Fig. 4.4. The term of $s(1 - k^2)L_s$ in the numerator of Eq. (4.6) shows that the transistor noise is partly canceled by the transformer.

Load Resistor Noise Cancellation

The CG transistor M_1 drains a part of the output noise current originating from v_{nR_L} , reducing the output noise voltage, as shown in Fig. 4.5(b). The noise current due to v_{nR_L} , which is given by $v_{nR_L}/(sL_s + R_L + Z_L)$, first flows through L_s and then the transformer induces a noise voltage



Figure 4.5: Mechanisms for noise cancellation of (a) i_{nd} and (b) v_{nR_L} .

 $v'_{nR_L} = -sMv_{nR_L}/(sL_s + R_L + Z_L)$ to L_p . The transistor M_1 detects a gate-source voltage $v_{gs} = v'_{nR_L}/sL_p(1/R_s + g_m + sC_p + 1/sL_p)$, and drains noise current of $g_m v_{gs}$ accordingly. This results in a reduction of the output noise current originating from v_{nR_L} . Considering the noise current due to M_1 , we can obtain the output noise current flowing Z_L :

$$i_{out,v_{nR_L}} = \frac{\left(\frac{1}{R_s} + (1 - nk)g_m + sC_p + \frac{1}{sL_p}\right)v_{nR_L}}{\left(\frac{1}{R_s} + Y_{IN}\right)\left(Z_L + R_L + s(1 - k^2)L_s\right)},\tag{4.7}$$

and the output noise voltage (at node A), $v_{out,v_{nR_L}}$, is given by $Z_L i_{out,v_{nR_L}}$. The term $-nkg_m$ in Eq. (4.7) originates from the noise cancellation.

Verification

The effectiveness of the transformer noise cancellation is verified through simulation. Figure 4.6 shows the simulated NF and NF_{min} of the proposed LNAs with and without the noise cancellation (i.e., k = 0, 1.0), where 90 nm CMOS process parameters are used and R_p represents the parasitic resistance of L_p . The LNA with k = 0 corresponds to a CG LNA with a load resistor and shunt-peaking inductor. The NF of the LNA with k = 1.0 is up to 2.2 dB lower than that of the LNA with k = 0. Figure 4.7 shows the simulated noise factors contributed from M_1 , R_L , and R_p (F_{M_1} , F_{R_L} , and F_{R_p} , respectively) with and without the noise cancellation. The transformer reduces F_{M_1} by up to 35 % and F_{R_L} by 65 %. The contribution from R_p also slightly decreases and hence R_p contributes little to the overall NF (i.e., 0.1 dB in Fig. 4.6). The noise contributions from M_1 and R_L change with the turn ratio. A noise optimization procedure will be presented in the next section.

4.4 Circuit Analysis

The transformer improves the LNA noise performance at the cost of the input and gain bandwidths. The output series inductor L_1 extends both the bandwidths. In this section, the noise, input admittance, gain, stability, and group delay of the LNA are analyzed, and noise optimization and impedance matching procedures are presented.

4.4.1 Noise

The amount of noise cancellation is mainly determined by the turn ratio of the transformer. From the small-signal equivalent circuit shown in Fig. 4.4, the output noise voltage due to R_s is given by

$$v_{out,i_{ns}} = Z_L \frac{(g_m R_L + nk + s(1 - k^2)L_s g_m) i_{ns}}{\left(\frac{1}{R_s} + Y_{IN}\right) (Z_L + R_L + s(1 - k^2)L_s)}.$$
(4.8)

Using Eqs. (4.6)–(4.8), we obtain the noise factor of the proposed LNA:

$$F \approx 1 + F_{M_1} + F_{R_L},$$
 (4.9)

$$F_{M_1} = \left| \frac{v_{out, i_{nd}}}{v_{out, i_{ns}}} \right|^2 = \left| \frac{n^2 - n + \left(\frac{1}{R_s} + j\omega C_p + \frac{1}{j\omega L_p}\right) R_L}{g_m R_L + n} \right|^2 \gamma g_{d0} R_s,$$
(4.10)

$$F_{R_L} = \left| \frac{v_{out, v_{nR_L}}}{v_{out, i_{ns}}} \right|^2 = \left| \frac{(1-n) g_m R_L + \left(\frac{1}{R_s} + j\omega C_p + \frac{1}{j\omega L_p}\right) R_L}{g_m R_L + n} \right|^2 \frac{R_s}{R_L},$$
(4.11)

where g_{d0} is the zero-bias drain conductance of M_1 . The value of γ in a fabricated 90 nm MOS-FET is approximately two, shown by the measured and simulated NF of the LNA, as will be



Figure 4.6: Simulated NF and NF_{\min} of the LNAs with and without noise cancellation (k = 0, 1.0)



Figure 4.7: Simulated noise contributions from M_1 , R_L , and R_p to the LNAs with and without noise cancellation (k = 0, 1.0).

shown in Section 4.6. For simplicity, the magnetic coupling factor k is assumed to be one. The parasitic resistance of L_p , the parasitic capacitance between L_p and L_s , and the induced-gate noise current of M_1 are ignored, because they do not have a significant effect on the overall NF. The transconductance g_m and load resistance R_L cannot be optimized for noise, because they are determined from input impedance matching conditions, as will be shown in the following subsection.

The optimum *n* for the noise performance can be obtained from Eqs. (4.9)–(4.11). Setting the derivatives of Eqs. (4.10) and (4.11) with respect to *n* to zero (i.e., $\partial F_{M_1}/\partial n = 0$ and $\partial F_{R_L}/\partial n = 0$ for $\omega = 1/\sqrt{L_p C_p}$), we can obtain

$$n_{opt,i_{nd}} = -g_m R_L + \sqrt{(g_m R_L)^2 + g_m R_L + \frac{R_L}{R_s}},$$
(4.12)

$$n_{opt,v_{nR_L}} = 1 + \frac{1}{g_m R_s},\tag{4.13}$$

for which a minimum F_{M_1} and F_{R_L} are achieved, respectively. Similarly, the optimum n for F, n_{opt} , can be obtained from $\partial F/\partial n = 0$:

$$\left(n_{opt}^2 - n_{opt} + \frac{R_L}{R_s} \right) \left(n_{opt}^2 + (2n_{opt} - 1)g_m R_L - \frac{R_L}{R_s} \right) \gamma g_{d0} - \left((1 - n_{opt})g_m + \frac{1}{R_s} \right) \left(g_m^2 R_L + g_m + \frac{1}{R_s} \right) R_L = 0.$$
(4.14)

Figures 4.8(a) and (b) show the calculated F, F_{M_1} , and F_{R_L} ($R_L = 50 \Omega$) versus n and NF with R_L as a parameter at $\omega = 1/\sqrt{L_p C_p}$, respectively. For $R_L = 50 \Omega$, minimum F_{M_1} , F_{R_L} , and F are achieved for n of 0.68, 1.66, and 1.0, given by Eqs. (4.12)–(4.14), respectively. Figure 4.8(b) shows that the calculated NF ($R_L = 150 \Omega$) for n = 1.0 is consistent with the simulated NF (k = 1.0) at 7.2 GHz, shown in Fig. 4.6, although R_p is ignored in Eqs. (4.9)–(4.11). Moreover, the NF becomes a minimum around one even with varying R_L from 50 to 200 Ω . A large n makes the LNA unstable, as will be explained in Section 4.4.4, and leads to an increase in the parasitic capacitance of L_s , causing poor high-frequency performance. The optimum n is thus determined to be one.

4.4.2 Input Impedance Matching

In the proposed topology with input and shunt-peaking inductors coupled, the output load affects the LNA input impedance through the coupling. The output series inductor L_1 contributes to wideband input impedance matching. From Fig. 4.4, the input admittance of the proposed LNA, Y_{IN} , is given by

$$Y_{IN}(j\omega) = g_m + j\omega C_p + \frac{1}{j\omega L_p} + Y_T.$$
(4.15)



Figure 4.8: Calculated (a) F, F_{M_1} , and F_{R_L} ($R_L = 50 \Omega$) versus n and (b) NF with R_L as a parameter.

The first three terms in Eq. (4.15) represent the input admittance of the CG LNA. The last term Y_T is generated by coupling L_p and L_s , and is given by

$$Y_T(j\omega) = \frac{nk(nk - g_m Z_L)}{R_L + Z_L + j\omega n^2 L_p (1 - k^2)}.$$
(4.16)

When L_1 is connected in series with the output, Z_L is expressed as

$$Z_L(j\omega) = \frac{1}{j\omega C_1} / \left(j\omega L_1 + \frac{1}{j\omega C_2} \right), \qquad (4.17)$$

where C_1 represents the sum of the gate-drain capacitance of M_1 and the parasitic capacitance of L_1 ; C_2 , which is typically larger than C_1 , represents the sum of the input capacitance of the following stage and the parasitic capacitance of L_1 . Equations (4.15)–(4.17) show that Y_{IN} is a function of Y_T , whose frequency behavior significantly depends on that of Z_L . From Eqs. (4.15)– (4.17), the calculated frequency behavior of Y_T is shown in Fig. 4.9(a) (solid line), and that of Y_T for $Z_L(j\omega) = 1/j\omega(C_1 + C_2)$ is also shown for comparison (dashed line). The π network consisting of C_1 , L_1 , and C_2 acts as a short or an open [27] (i.e., $Z_L = 0$ or ∞), providing a maximum and minimum $\text{Re}[Y_T(j\omega)]$ and $\text{Im}[Y_T(j\omega)]$:

$$\operatorname{Re}[Y_T(j\omega)]_{\max} \approx \frac{n^2 k^2}{R_L},\tag{4.18}$$

$$\operatorname{Re}[Y_T(j\omega)]_{\min} \approx -nkg_m,$$
(4.19)

$$\operatorname{Im}[Y_T(j\omega)]_{\max} \approx \frac{nk}{2} \left(g_m + \frac{nk}{R_L} \right), \tag{4.20}$$

$$\operatorname{Im}[Y_T(j\omega)]_{\min} \approx -\frac{nk}{2} \left(\frac{g_m \frac{L_1 C_2}{C_1 + C_2}}{\frac{L_1 C_2}{C_1 + C_2} + n^2 (1 - k^2) L_p} + \frac{nk}{R_L} \right),$$
(4.21)

at the following frequencies:

$$\omega_1 = \frac{1}{\sqrt{L_1 C_2}},\tag{4.22}$$

$$\omega_2 = \frac{1}{\sqrt{L_1 \frac{C_1 C_2}{C_1 + C_2}}},\tag{4.23}$$

$$\omega_3 \approx \frac{1}{R_L(C_1 + C_2)},\tag{4.24}$$

$$\omega_4 \approx \frac{R_L}{\frac{L_1 C_2}{C_1 + C_2} + n^2 (1 - k^2) L_p},\tag{4.25}$$

respectively. The above equations and approximations are derived from the following conditions: $Z_L = 0$ and $j\omega n^2(1-k^2)L_p$ is ignored against R_L in Eqs. (4.18) and (4.22); $Z_L = \infty$ in Eqs. (4.19) and (4.23); $Z_L = 1/j\omega(C_1 + C_2)$ and $\omega^2 n^2(1-k^2)L_p(C_1 + C_2) \ll 1$ in Eqs. (4.20)



Figure 4.9: Calculated real and imaginary parts of (a) Y_T and (b) Y_{IN} , and (c) S_{11} of the LNAs with and without L_1 .

and (4.24); $Z_L = j\omega L_1 C_2/(C_1 + C_2)$ in Eqs. (4.21) and (4.25). A negative $\text{Re}[Y_T(j\omega)]$ shown in Eq. (4.19) originates from the positive feedback provided by the transformer. The calculated Y_{IN} is also shown in Fig. 4.9(b). The real part of Y_{IN} , $\text{Re}[Y_{IN}(j\omega)]$, is simply shifted by g_m from $\text{Re}[Y_T(j\omega)]$, and the imaginary part of Y_{IN} , $\text{Im}[Y_{IN}(j\omega)]$, becomes zero at resonance frequencies. The first resonance frequency ω_0 is derived from the following equation:

$$\operatorname{Im}[Y_{IN}(j\omega_0)] \approx j\omega_0 C_p + \frac{1}{j\omega_0 L_p} + j\omega_0 \frac{nk(C_1 + C_2)(nk + g_m R_L)}{1 + \omega_0^2 R_L^2 (C_1 + C_2)^2} = 0,$$
(4.26)

where the last term in Im[$Y_{IN}(j\omega_0)$] is Im[$Y_T(j\omega_0)$] for $Z_L = 1/j\omega_0(C_1 + C_2)$ and $\omega_0^2 n^2(1 - k^2)L_p(C_1 + C_2) \ll 1$.

Input impedance matching conditions are derived from $Y_{IN}(j\omega_0)$ and $Y_{IN}(j\omega_1)$. At ω_0 , the real part of $Y_{IN}(j\omega)$ is approximated as

$$\operatorname{Re}[Y_{IN}(j\omega_0)] \approx g_m \left(1 - \frac{nk}{1 + \omega_0^2 R_L^2 (C_1 + C_2)^2} \right),$$
(4.27)

where $\omega_0^2 n k R_L (C_1 + C_2)^2 / g_m \ll 1$. At ω_1 , Im $[Y_{IN}(j\omega)]$ is negligible against Re $[Y_{IN}(j\omega)]$:

$$Y_{IN}(j\omega_1) \approx \operatorname{Re}[Y_{IN}(j\omega_1)] = g_m + \frac{n^2 k^2}{R_L}.$$
(4.28)

For input impedance matching ($|S_{11}| < -10 \text{ dB}$), Y_{IN} must satisfy the following condition:

$$|S_{11}| = \left|\frac{1 - R_s Y_{IN}}{1 + R_s Y_{IN}}\right| < 0.316.$$
(4.29)

When $\text{Im}[Y_{IN}(j\omega)] = 0$, Eq. (4.29) can be simplified to

$$10 \text{ mS} < \text{Re}[Y_{IN}(j\omega)] < 38 \text{ mS}.$$
 (4.30)

Substituting Eqs. (4.27) and (4.28) into Eq. (4.30), we can derive the following impedance matching conditions:

$$10 \text{ mS} < g_m \left(1 - \frac{nk}{1 + \omega_0^2 R_L^2 (C_1 + C_2)^2} \right), \tag{4.31}$$

$$g_m + \frac{n^2 k^2}{R_L} < 38 \text{ mS.}$$
(4.32)

Equations (4.31) and (4.32) determine the lower and upper limits to g_m and the lower limit to R_L . An impedance matching procedure for the proposed LNA is as follows:

- 1. Select g_m and R_L to satisfy Eqs. (4.31) and (4.32)
- 2. Select L_p such that ω_0 equals the lower edge of the desired input band

3. Select L_1 such that ω_4 equals the upper edge of the desired input band

Figure 4.9(c) shows the calculated S_{11} of the LNAs with and without L_1 , where ω_0 and ω_4 are set to approximately 3.1 GHz and 10.6 GHz, respectively. A transconductance of 30 mS and load resistance of 150 Ω allow $|S_{11}| < -10$ dB from ω_0 to ω_1 . Around ω_4 , the π network including L_1 decreases Re[$Y_{IN}(j\omega)$] and Im[$Y_{IN}(j\omega)$]:

$$\operatorname{Re}[Y_{IN}(j\omega_4)] = g_m - \frac{nk}{2R_L} \left(\frac{\omega_4 L_1 C_2 g_m}{C_1 + C_2} - nk\right),$$
(4.33)

$$\operatorname{Im}[Y_{IN}(j\omega_4)] = \omega_4 C_p - \frac{1}{\omega_4 L_p} - \frac{nk}{2R_L} \left(\frac{\omega_4 L_1 C_2 g_m}{C_1 + C_2} + nk\right),$$
(4.34)

providing $|S_{11}| < -10$ dB. Consequently, the input impedance matching is achieved from ω_0 to ω_4 .

4.4.3 Gain

The transformer provides the positive feedback from node A to the input, as shown in Section 4.3.1. The transformer positive feedback reduces the gain (S_{21}) bandwidth of the LNA. The S_{21} of the LNA with output impedance matching is given by

$$S_{21} = \frac{v_{out}}{v_s/2} = \frac{2v_{in}}{v_s} \frac{v_{out}}{v_{in}} = \frac{2}{1 + R_s Y_{IN}} A_v,$$
(4.35)

where v_s is the signal voltage and A_v , defined by v_{out}/v_{in} , is the voltage gain from the input to the output of the LNA, as shown in Fig. 4.4. Equation (4.35) shows that the frequency response of A_v is shaped by that of Y_{IN} (i.e., S_{11}), which results in that of S_{21} .

The frequency response of A_v of the proposed LNA is similar to that of a CG LNA with a load resistor and output series inductor. The output network combined a shunt-peaking inductor with an output series inductor gives a larger bandwidth than the counterpart with either inductor, as explained in [23, 27]. However, the shunt-peaking inductor L_s in the proposed LNA does not increase the bandwidth. The A_v of the LNA is given by

$$A_{v}(s) = \frac{g_{m}R_{L}\left(1 + \frac{nk}{g_{m}R_{L}} + \frac{1-k^{2}}{m_{1}}\frac{s}{\omega_{c}}\right)}{1 + \frac{s}{\omega_{c}} + \left(\frac{1-k^{2}}{m_{1}} + \frac{1-k_{c}}{m_{2}}\right)\frac{s^{2}}{\omega_{c}^{2}} + \frac{k_{c}(1-k_{c})}{m_{2}}\frac{s^{3}}{\omega_{c}^{3}} + \frac{1-k^{2}}{m_{1}}\frac{k_{c}(1-k_{c})}{m_{2}}\frac{s^{4}}{\omega_{c}^{4}}},$$
(4.36)

where $\omega_c = 1/R_L(C_1 + C_2)$, $k_c = C_1/(C_1 + C_2)$, $m_1 = R_L^2(C_1 + C_2)/L_s$, and $m_2 = R_L^2(C_1 + C_2)/L_1$ [23]. Substituting k = 0 into Eq. (4.36) gives the A_v of the CG LNA with both the shuntpeaking and output series inductors. Equation (4.36) shows that all m_1 are divided by a factor of $(1 - k^2)$, i.e., L_s is multiplied by a factor of $(1 - k^2)$. This means that the effective L_s in the proposed LNA becomes small, compared with the shunt-peaking inductor in the CG LNA, and then contributes less to bandwidth extension. The calculated A_v with k as a parameter is shown in Fig. 4.10, where $f_c = 4.2$ GHz, $k_c = 0.4$, $m_1 = 1.6$, and $m_2 = 2.25$ originate from $C_1 = 100$ fF, $C_2 = 150$ fF, $L_s = 3.5$ nH, $L_1 = 2.5$ nH, and $R_L = 150 \Omega$. A very large peak (ripple) is found when k = 0, because L_s is larger than L_1 , i.e., $m_1 < m_2$ [23, 27]. Figure 4.10 shows that both the peak and bandwidth decrease as k increases. Consequently, the bandwidth of the proposed LNA ($k \simeq 0.9$) closely equals that of the CG LNA with only the output series inductor. A flat voltage gain of the CG LNA across the entire UWB frequency band can be obtained by selecting an appropriate value of m_2 (approximately 2), as discussed in [23].

An S_{21} variation of the proposed LNA mainly originates from the characteristic of Y_{IN} (S_{11}). As shown in Fig 4.9(c), the input impedance matching condition improves around ω_0 and ω_4 , but deteriorates around ω_1 . This means that an input signal of ω_1 is less transferred to the input of the LNA, compared to that of ω_0 or ω_4 , which results in a reduction in the magnitude of S_{21} around ω_1 . The difference between $S_{21}(j\omega_0)$ and $S_{21}(j\omega_1)$ can be approximated from Eq. (4.35):

$$\Delta S_{21} = \frac{S_{21}(j\omega_0)}{S_{21}(j\omega_1)} = \frac{1 + R_s Y_{IN}(j\omega_1)}{1 + R_s Y_{IN}(j\omega_0)} \approx \frac{1 + R_s \cdot \operatorname{Re}[Y_{IN}(j\omega_1)]}{1 + R_s \cdot \operatorname{Re}[Y_{IN}(j\omega_0)]},$$
(4.37)

where $A_v(j\omega_0) = A_v(j\omega_1)$ is assumed, the real parts of $Y_{IN}(j\omega_0)$ and $Y_{IN}(j\omega_1)$ are given by Eqs. (4.27) and (4.28), respectively, and the imaginary parts of $Y_{IN}(j\omega_0)$ and $Y_{IN}(j\omega_1)$ can be neglected, as shown in the previous subsection. Figure 4.11 shows the calculated A_v and S_{21} of the proposed LNA with k = 0.9. Substituting the parameters shown in Fig. 4.11 into Eq. (4.37) gives $\Delta S_{21} = -3.6$ dB, while an S_{21} variation of -4.7 dB is seen in Fig. 4.11, and then -1.1 dB originates from the difference of A_v . The difference of S_{21} can be reduced by decreasing R_L , as shown by Eqs. (4.27), (4.28), and (4.37). Moreover, using a common-source (CS) amplifier with a gain peak around ω_1 as the second stage, we can obtain a flat gain.

4.4.4 Stability

The proposed LNA becomes potentially unstable, due to the transformer positive feedback. For unconditionally stable, the LNA must meet the following conditions, as shown in Section 1.2.2:

$$\operatorname{Re}[Z_{IN}] > 0, \tag{4.38}$$

$$\operatorname{Re}[Z_{OUT}] > 0. \tag{4.39}$$

In what follows, to simplify the expression of the output impedance of the LNA, we will verify whether the LNA without the output π network (C_1 , L_1 , and C_2) satisfies the above conditions or not.

First, the real part of the input admittance of the LNA can be derived from Eqs. (4.15)–(4.16) for $Z_L = \infty$:

$$\operatorname{Re}[Y_{IN}(j\omega)] = g_m \left(1 - nk\right). \tag{4.40}$$

In the case of the proposed LNA, n is selected to be one, shown in Section 4.4.1, and k of the on-chip transformer is less than one [28]: nk < 1. The requirement of Eq. (4.38) is thus satisfied.



Figure 4.10: Calculated A_v of the proposed LNA with k as a parameter.



Figure 4.11: Calculated A_v and S_{21} of the proposed LNA with k = 0.9.



Figure 4.12: Simulated K and B_1 of the proposed LNA.

Next, the output impedance considering the left hand side of output node A (Fig. 4.4) is given by

$$Z_{OUT,A}(j\omega) = \frac{\left(\frac{1}{R_s} + g_m + j\omega C_p\right) (R_L + j\omega(1 - k^2)L_s) + n^2 + \frac{R_L}{j\omega L_p}}{\frac{1}{R_s} + g_m(1 - nk) + j\omega C_p + \frac{1}{j\omega L_p}}.$$
(4.41)

Equation (4.41) indicates that the real part of $Z_{OUT,A}$ becomes a maximum around $\omega = 1/\sqrt{L_p C_p}$ and can be approximated by

$$\operatorname{Re}[Z_{OUT,A}(j\omega)] \approx R_L,$$
(4.42)

$$\operatorname{Re}[Z_{OUT,A}(j\omega)] \approx R_L + \frac{(1-k^2)L_s}{C_p} \left(\frac{1}{R_s} + g_m\right), \qquad (4.43)$$

at low and high frequencies (i.e., $\omega \ll 1/\sqrt{L_pC_p}$ and $\omega \gg 1/\sqrt{L_pC_p}$), respectively. The requirement of Eq. (4.39) is therefore satisfied.

The stability is also ensured through simulation. Figure 4.12 shows the simulated K and B_1 of the proposed LNA, which are given by [29]

$$K = \frac{1 - |S_{11}|^2 - |S_{22}|^2 + |\Delta|^2}{2|S_{12}S_{21}|},$$
(4.44)

$$B_1 = 1 + |S_{11}|^2 - |S_{22}|^2 - |\Delta|^2,$$
(4.45)

respectively, where $\Delta = S_{11}S_{22} - S_{12}S_{21}$. The necessary and sufficient conditions for unconditional stability are K > 1 and $B_1 > 0$ [29]. The simulations show that the LNA satisfies these conditions across the entire UWB frequency band.

4.4.5 Group Delay

A group delay variation is important for DS-UWB or pulse-based UWB systems. The group delay is the derivative of the phase of the signal transfer function (S_{21}) , and hence any resonance in the signal path contributes to the variation [30]. The critical resonances in the proposed LNA originate from the combinations of L_p (transformer) and C_p at the input, and L_1 , C_1 , and C_2 at the output, and these resonance frequencies, ω_0 and ω_2 , are given by Eqs. (4.26) and (4.23), respectively. Pushing the resonance frequencies out of the desired frequency band (i.e., increasing L_p or decreasing L_1) allows a small group delay variation. Figure 4.13 shows that the simulated group delays of the proposed LNA with L_p and L_1 as a parameter. The group delay (for $L_p = 3.0$ nH in Fig. 4.13(a)) dramatically changes around 3 GHz (ω_0) and 11 GHz (ω_2). The simulations also show that the variation can be reduced by increasing L_p or decreasing L_1 .

4.5 Design

By using a 90 nm CMOS process and device parameters, the proposed LNA is designed to satisfy the following typical specifications of the UWB LNA: $|S_{11}| < -10$ dB, NF < 4 dB, and $|S_{21}| > 10$ dB across the entire UWB frequency band (3.1–10.6 GHz). Current consumption is set to 2.5 mA at a 1.0 V supply.

4.5.1 Input Transistor and Load Resistor

The transconductance g_m and load resistance R_L are determined by the input impedance matching conditions, given by Eqs. (4.31) and (4.32), and the desired gain. A transconductance of 30 mS and load resistance of 145 Ω provide both $|S_{11}| < -10$ dB and $A_v \simeq 14$ dB² in the lower UWB band (3.1–5 GHz). The load resistance includes the parasitic resistance of L_s . A bias current of 2.5 mA and g_m of 30 mS result in a gate width of $4 \times 10 \ \mu m$ (10 gate fingers, each with a unit of of 4 μm width) and gate length of 100 nm.

4.5.2 Transformer

The transformer adopts a stacked configuration in which L_p is stacked on L_s . This configuration provides the largest coupling factor and a small area [28]. The large parasitic resistance of L_s , due to the lower thin metal layer, is not problematic, because it can be absorbed into R_L .

The parasitic capacitance between L_p and L_s , C_c , has a relatively small effect on the LNA performance. This capacitance significantly affects the frequency response of a noninverting transformer [28]. Although the proposed LNA employs the noninverting stacked transformer, the signal current injected into L_s by M_1 reduces the effect of C_c . Figure 4.14 shows the simulated S_{11} and NF of the LNA including C_c . In the lower UWB band, C_c slightly increases the magnitude of the S_{11} and has little impact on the NF; in the higher, a large C_c decreases the

²At low frequencies, $A_v \approx g_m R_L + nk = 5.25$.



Figure 4.13: Simulated group delays with (a) L_p and (b) L_1 as a parameter.



Figure 4.14: Simulated S_{11} and NF of the LNA with C_c as a parameter.

input bandwidth and increases the NF by up to 0.20 dB. In the simulations, for wideband input impedance matching, C_c must be less than 300 fF, which can be realized even with the stacked transformer.

The transformer is designed to achieve $|S_{11}| < -10$ dB (of the LNA) in the lower UWB band and NF < 4.0 dB across the entire UWB band. Figure 4.15 shows the top view and cross section of the designed transformer. Selecting L_p such that ω_0 equals approximately 3.1 GHz allows the LNA to achieve $|S_{11}| < -10$ dB in the lower band. A wide metal for realizing L_p reduces the parasitic resistance; however, it leads to a large chip area and large parasitic capacitance. For L_p , we adopt a 3.5-turn square inductor with an outer diameter of 165 μ m, metal width of 3 μ m, and metal spacing of 2 μ m. To achieve a turn ratio of one, L_s is designed as follows: an outer diameter is 165 μ m, a metal width 2 μ m, and a metal spacing 3 μ m. The metal thicknesses of L_p (top pad metal) and L_s (metal 6) are 1.9 μ m and 0.9 μ m, respectively. The parasitic capacitance (3 μ m), which results in $C_c \simeq 240$ fF. Three-dimensional (3-D) electromagnetic (EM) simulations by Ansoft HFSS showed $L_p = L_s = 4.0$ nH and k = 0.9.

4.5.3 Output Series Inductor

The output series inductor L_1 is designed to set ω_4 to the upper edge of the desired input band (10.6 GHz). We use a relatively low Q inductor to reduce the chip area and parasitic capacitance, which reduces the gain bandwidth of the LNA. The outer diameter of L_1 is 140 μ m, the metal width 3 μ m, the metal spacing 2 μ m, and the metal thickness 1.9 μ m (top pad metal). EM



Figure 4.15: (a) Top view and (b) cross section a-a' of the designed transformer.

simulations showed that the inductance and maximum Q were 3.2 nH and 6.0 (at 5.0 GHz), respectively.

4.6 Experimental Results and Discussion

The designed LNA shown in Fig. 4.16 was fabricated in a 90 nm digital CMOS process without metal-insulator-metal (MIM) capacitors. A DC-blocking capacitor of 400 fF consisted of vertical parallel plates [31], due to low input and output parasitic capacitances (~ 25 fF). A micrograph of the fabricated LNA is shown in Fig. 4.17. The active chip area excluding pads was 0.48×0.25 mm². The input and output pads was not electrostatic-discharge (ESD) protected. Metal fills consisting of metal 1–6 layers were placed both inside and outside the fabricated transformer and



Figure 4.16: Complete schematic of the designed transformer noise-canceling LNA



Figure 4.17: Micrograph of the fabricated LNA.

inductor to meet metal density rules in the CMOS process. They were 1.5 μ m by 1.5 μ m squares with a spacing of 0.2 μ m. The average horizontal distance between the metal fills and traces of the inductors was 15 μ m. For measurements, a unity-gain CS amplifier with a 50 Ω output resistor was used as a buffer. The S-parameters, noise, and linearity of the LNA were measured using on-wafer RF probes. The power consumption of the LNA and buffer were 2.5 mW and 4.0 mW, respectively, at a supply voltage of 1.0 V.

4.6.1 S-parameters

Figure 4.18 shows the measured and simulated S_{11} and S_{21} of the LNA. The S-parameters was measured using an Agilent Technologies HP8722ES network analyzer. The LNA achieved $|S_{11}| < -10$ dB and $|S_{21}| > 9.3$ dB across 3.1–10.6 GHz. The discrepancy between the measurements and simulations at frequencies above 4 GHz is mainly attributed to insufficient accuracy in the simulations of the transformer and inductor used. The HFSS simulation models of the transformer and inductor included no metal fills to solve convergence problems and reduce the memory requirement. The metal fills increase the parasitic capacitances and resistances of the transformer and inductor [32–34], which results in the discrepancy.

Figure 4.19 shows the measured and simulated S_{12} of the LNA with the buffer. The difference between the measured and simulated S_{12} is due to substrate effects. The LNA achieved $|S_{12}| < -34$ dB across 3.1–10.6 GHz. The measured S_{12} of the stand-alone buffer (not shown) was less than -24 dB over the same frequency range. Thus, the S_{12} of the LNA without the buffer was less than -10 dB. The poor reverse isolation performance is due to the transformer, and an additional stage may be required to improve the isolation performance.

Figure 4.20 shows the measured and simulated group delays. The group delay variation increased around the edge of the UWB frequency band, as analyzed in Section 4.4.5. A group delay variation of approximately 60 ps was achieved for the entire band.

4.6.2 NF

Figure 4.21 shows the measured and simulated NF of the LNA. The NF was measured using an Agilent Technologies HP8970B noise figure meter. Note that these results included the noise of the output buffer, which increased the overall NF by 0.8 dB for an LNA gain of 10 dB in simulation. The LNA with the buffer achieved an NF of 3.8–4.4 dB across the entire UWB band. This means that the proposed LNA with an additional CS amplifier like the buffer can achieve NF < 4.4 dB. The difference between the measurements and simulations can be explained by the extra input-referred noise of the buffer, caused by the lower measured gain than the simulated one.

4.6.3 Linearity

Figure 4.22 shows the output power of the fundamental tone and third-order intermodulation (IM_3) products for two tones (3.000 GHz and 3.001 GHz), measured using an Agilent Tech-



Figure 4.18: Measured and simulated S_{11} and S_{21} of the LNA.



Figure 4.19: Measured and simulated S_{12} of the LNA.



Figure 4.20: Measured and simulated group delays of the LNA.



Figure 4.21: Measured and simulated NF of the LNA.



Figure 4.22: Measured IIP_3 of the LNA at 3 GHz.



Figure 4.23: Measured IIP_3 and IIP_2 of the LNA.

nologies E4448A spectrum analyzer. The two tones were generated by Agilent Technologies HP8671B and E4438C signal generators. The measured IIP_3 and 1-dB compression point were approximately -9.3 dBm and -20 dBm, respectively. Figure 4.23 shows IIP_3 and IIP_2 measured by applying two tones with 1-MHz spacing. The measured frequency range of 3–6 GHz was limited by the signal generator. An $IIP_3 > -9.3$ dBm and $IIP_2 > -6.3$ dBm were obtained in the frequency range.

4.6.4 Comparison

Table 4.1 shows a summary of the LNA performance and a comparison with previously reported 3.1–10.6 GHz CMOS LNAs. The proposed LNA achieved input impedance matching and comparable noise performance across the entire UWB band with the lowest reported power consumption and supply voltage. The LNA also consumed the smallest chip area among the wideband LNAs employing inductors [10, 11, 16, 23].

An additional amplifier stage can allow the proposed LNA to achieve a more and flatter gain across 3.1-10.6 GHz. A relatively low gain of the implemented LNA (>9.3 dB) leads to an increase in the overall NF of the receiver. For instance, the NF specification for RF receivers of the MB-OFDM UWB system is less than 6.6 dB [35, 36]. A receiver employing the proposed LNA may have difficulty in satisfying such an NF specification. A CS amplifier with a load inductor, shown in Fig. 4.24, improves the LNA gain, alleviating this problem. The CS amplifier is designed to have a gain peak around 8.0 GHz and a power consumption of 2.0 mW. The 2.6-nH inductor consists of stacked square spiral inductors implemented by the top pad metal and metal 6 layers, and occupies only $55 \times 55 \ \mu m^2$. Figure 4.25 shows the simulated S_{21} and NF of the LNAs with and without the CS amplifier. The inductor was designed by using the EM simulator. The LNA with the CS amplifier achieved more and flatter gain ($|S_{21}| > 20$ dB) and the same noise performance as the LNA without the CS amplifier (NF < 4.3 dB) across 3.1–10.6 GHz. The group delay variation (not shown) was reduced to approximately 20 ps. Considering the measurements of the fabricated LNA, we conclude that the proposed LNA with the CS amplifier can achieve $|S_{21}| > 20$ dB and NF < 4.4 dB across 3.1–10.6 GHz with an additional power consumption of 2.0 mW and chip area of $55 \times 55 \ \mu m^2$.

4.7 Conclusion

We have demonstrated a transformer noise-canceling UWB CMOS LNA with an output series inductor. The transformer partly cancels the noise of the common-gate transistor and load resistor, thereby improving the LNA noise performance. The output series inductor improves both the gain and input bandwidths. Circuit analysis showed that the best turn ratio for the noise performance is one and input impedance matching depends not only on the common-gate transistor but also on the load resistor. The LNA designed for UWB applications was fabricated in a 90 nm digital CMOS process. The fabricated LNA occupied 0.12 mm², and achieved $|S_{11}| < -10$ dB, NF < 4.4 dB, and $|S_{21}| > 9.3$ dB across 3.1–10.6 GHz, while consuming 2.5 mW from a 1.0 V

CMOS LNAs.
wideband
nparison of
ce and con
l performan
: Measured
Table 4.1

Reference	CMOS Technology	BW* [GHz]	NF [dB]	S_{21} [dB]	IIP_3 [dBm]	Supply [V]	Power [†] [mW]	Area [mm ²]	Topology
This work	90 nm	2.8-13.2	$3.8-4.4^{\ddagger}$	9.3–13.1 [‡]	-9.3	1.0	2.5	0.12	Transformer noise-canceling
[4]	90 nm	0.5-6.2	1.9–2.6	25^{\diamond}	-16	2.7	35.2	0.025	Resistive feedback CS
[9]	90 nm	0.5 - 5.0	2.3-2.6	21–22	-8.8	1.8	12	0.012	Resistive feedback CS
[7]	90 nm	9-0	3.4-4.3	12.5–15.3	N/A	1.0	3.4	0.0017	Common-drain feedback CS
[8]	130 nm	1-7.4	>2.4	15-17	-4.1	1.4	25	0.019	Common-drain feedback CS*
[10]	130 nm	3.1-10.6	2.1–2.9	13.7–16.5	-8.5	1.2	9.0	0.40	Reactive feedback CS
[11]	180 nm	0-14.1	$4.5-5.1^{\ddagger}$	12.0–13.7 [‡]	-6.2	1.8	20	0.50	Noise-canceling CG using CS
[13]	65 nm	0.2–6.2	2.8-4.2	9.9–15.6°	0	1.2	14	0.009	Noise-canceling CG using CS
[14]	130 nm	3.7-8.8	3.6-4.5	8.1–11	-7.2	1.5	19	0.05	Noise-canceling CS
[16]	180 nm	1-11	2.9–3.0	8–9	-3.55	1.8	21.6	0.20	Distributed
[23]	180 nm	2.6-10.7	4.4–5.3 [‡]	6.0–8.5 [‡]	7.4	1.8	4.5	0.40	Capacitor cross-coupled CG*

*Input bandwidth

[†]Without buffers [‡]3.1–10.6 GHz frequency range [◊]Voltage gain *Input-output differential topology

88

CHAPTER 4. TRANSFORMER NOISE-CANCELING UWB CMOS LNA



Figure 4.24: Common-source amplifier with a load inductor.



Figure 4.25: Simulated S_{21} and NF of the LNAs with and without the common-source amplifier.

supply. The proposed topology is the most suitable for low-power and low-voltage UWB CMOS LNAs.

Bibliography

- [1] MultiBand OFDM Alliance SIG. (2004, Sep.) Multiband OFDM physical layer proposal for IEEE 802.15 task group 3a. [Online]. Available: http://www.wimedia.org
- [2] R. Fisher and R. Kohno and H. Ogawa and H. Zhang and K. Takizawa and M. McLaughlin and M. Welborn. (2005, Sep.) DS-UWB physical layer submission to 802.15 task group 3a. [Online]. Available: http://www.decawave.com
- [3] A. Bevilacqua and A. M. Niknejad, "An ultrawideband CMOS low-noise amplifier for 3.1– 10.6 GHz wireless receivers," *IEEE J. Solid-State Circuits*, vol. 39, no. 12, pp. 2259–2268, Dec. 2004.
- [4] J.-H. C. Zhan and S. S. Taylor, "A 5 GHz resistive-feedback CMOS LNA for low-cost multi-standard applications," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, San Francisco, CA, Feb. 2006, pp. 200–201.
- [5] B. G. Perumana, J.-H. C. Zhan, S. S. Taylor, and J. Laskar, "A 12 mW, 7.5 GHz bandwidth, inductor-less CMOS LNA for low-power, low-cost, multi-standard receivers," in *IEEE Radio Frequency Integrated Circuits Symp. Dig. Papers*, Honolulu, HI, Jun. 2007, pp. 57–60.
- [6] B. G. Perumana, J.-H. C. Zhan, S. S. Taylor, B. R. Carlton, and J. Laskar, "Resistive-feedback CMOS low-noise amplifiers for multiband applications," *IEEE Trans. Microw. Theory Tech.*, vol. 56, no. 5, pp. 1218–1225, May 2008.
- [7] J. Borremans, P. Wambacq, and D. Linten, "An ESD-protected DC-to-6 GHz 9.7 mW LNA in 90 nm digital CMOS," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, San Francisco, CA, Feb. 2007, pp. 422–423.
- [8] R. Ramzan, S. Andersson, J. Dabrowski, and C. Svensson, "A 1.4 V 25 mW inductorless wideband LNA in 0.13 μm CMOS," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, San Francisco, CA, Feb. 2007, pp. 424–425.
- [9] M. T. Reiha, J. R. Long, and J. J. Pekarik, "A 1.2 V reactive-feedback 3.1–10.6 GHz ultrawideband low-noise amplifier in 0.13 μm CMOS," in *IEEE Radio Frequency Integrated Circuits Symp. Dig. Papers*, San Francisco, CA, Jun. 2006, pp. 41–44.
- [10] M. T. Reiha and J. R. Long, "A 1.2 V reactive-feedback 3.1–10.6 GHz low-noise amplifier in 0.13 μm CMOS," *IEEE J. Solid-State Circuits*, vol. 42, no. 5, pp. 1023–1033, May 2007.
- [11] C.-F. Liao and S.-I. Liu, "A broadband noise-canceling CMOS LNA for 3.1–10.6-GHz UWB receivers," *IEEE J. Solid-State Circuits*, vol. 42, no. 2, pp. 329–339, Feb. 2007.

- [12] S. Chehrazi, A. Mirzaei, R. Bagheri, and A. A. Abidi, "A 6.5 GHz wideband CMOS low noise amplifier for muti-band use," in *Proc. IEEE Custom Integrated Circuits Conf.*, San Francisco, CA, Sep. 2005, pp. 801–804.
- [13] S. C. Blaakmeer, E. A. Klumperink, B. Nauta, and D. M. Leenaertsr, "An inductorless wideband balun-LNA in 65nm CMOS with balanced output," in *Proc. IEEE European Solid-State Circuits Conf.*, Munich, Germany, Sep. 2007, pp. 364–367.
- [14] Q. Li and Y. P. Zhang, "A 1.5-V 2–9.6-GHz inductorless low-noise amplifier in 0.13-μm CMOS," *IEEE Trans. Microw. Theory Tech.*, vol. 55, no. 10, pp. 2015–2023, Oct. 2007.
- [15] F. Zhang and P. R. Kinget, "Low-power programmable gain CMOS distributed LNA," *IEEE J. Solid-State Circuits*, vol. 41, no. 6, pp. 1333–1343, Jun. 2006.
- [16] P. Heydari, "Design and analysis of a performance-optimized CMOS UWB distributed LNA," *IEEE J. Solid-State Circuits*, vol. 42, no. 9, pp. 1892–1905, Sep. 2007.
- [17] T. Kihara, T. Matsuoka, and K. Taniguchi, "A 1.0 V, 2.5 mW, transformer noise-canceling UWB CMOS LNA," in *IEEE Radio Frequency Integrated Circuits Symp. Dig. Papers*, Atlanta, GA, Jun. 2008, pp. 493–496.
- [18] D. J. Allstot, X. Li, and S. Shekhar, "Design considerations for CMOS low-noise amplifiers," in *IEEE Radio Frequency Integrated Circuits Symp. Dig. Papers*, Fort Worth, TX, Jun. 2004, pp. 97–100.
- [19] P. P. Jindal, "Compact noise models for MOSFETs," *IEEE Trans. Electron Devices*, vol. 53, no. 9, pp. 2051–2061, Sep. 2006.
- [20] Y. Kiyota, C.-H. Chen, T. Kubodera, A. Nakamura, K. Takeshita, and M. J. Deen, "A new approach of high frequency noise modeling for 70-nm NMOS transistors by accurate noise source extraction," in *IEEE Radio Frequency Integrated Circuits Symp. Dig. Papers*, Honolulu, HI, Jun. 2007, pp. 635–638.
- [21] J. Jussila and P. Sivonen, "A 1.2-V highly linear balanced noise-cancelling LNA in 0.13-μm CMOS," *IEEE J. Solid-State Circuits*, vol. 43, no. 3, pp. 579–587, Mar. 2008.
- [22] W. Zhuo, S. Embabi, J. P. de Gyvez, and E. Sanchez-Sinencio, "Using capacitive crosscoupling technique in RF low noise amplifiers and down-conversion mixer design," in *Proc. IEEE European Solid-State Circuits Conf.*, Stockholm, Sweden, Jun. 2000, pp. 116–119.
- [23] S. Shekhar, J. S. Walling, and D. J. Allstot, "Bandwidth extension techniques for CMOS amplifiers," *IEEE J. Solid-State Circuits*, vol. 41, no. 11, pp. 2424–2439, Nov. 2006.
- [24] X. Li, S. Shekhar, and D. J. Allstot, "Gm-boosted common-gate LNA and differential colpitts VCO/QVCO in 0.18-μm CMOS," *IEEE J. Solid-State Circuits*, vol. 40, no. 12, pp. 2609–2619, Dec. 2005.

- [25] F. Bruccoleri, E. A. M. Klumperink, and B. Nauta, "Wide-band CMOS low-noise amplifier exploiting thermal noise canceling," *IEEE J. Solid-State Circuits*, vol. 39, no. 2, pp. 275– 282, Feb. 2004.
- [26] A. Liscidini, C. Ghezzi, E. Depaoli, G. Albasini, I. Bietti, and R. Castello, "Common gate transformer feedback LNA in a high IIP3 current mode RF CMOS front-end," in *Proc. IEEE Custom Integrated Circuits Conf.*, San Francisco, CA, Jul. 2006, pp. 25–28.
- [27] S. Galal and B. Razavi, "40-Gb/s amplifier and ESD protection circuit in 0.18-μm CMOS technology," *IEEE J. Solid-State Circuits*, vol. 39, no. 12, pp. 2389–2396, Dec. 2004.
- [28] J. R. Long, "Monolithic transformers for silicon RF IC design," *IEEE J. Solid-State Circuits*, vol. 35, no. 9, pp. 1368–1381, Sep. 2000.
- [29] G. Gonzalez, *Microwave Transistor Amplifiers*, 2nd ed. Upper Saddle River, NJ: Prentice Hall, 1997.
- [30] Y. Park, C.-H. Lee, J. D. Cressler, and J. Laskar, "The analysis of UWB SiGe HBT LNA for its noise, linearity, and minimum group delay variation," *IEEE Trans. Microw. Theory Tech.*, vol. 54, no. 4, pp. 1687–1697, Apr. 2006.
- [31] R. Aparicio and A. Hajimiri, "Capacity limits and matching properties of integrated capacitors," *IEEE J. Solid-State Circuits*, vol. 37, no. 3, pp. 384–393, Mar. 2002.
- [32] J.-H. Chang, Y.-S. Youn, H.-K. Yu, and C.-K. Kim, "Effects of dummy patterns and substrate on spiral inductors forsub-micron RF ICs," in *IEEE Radio Frequency Integrated Circuits Symp. Dig. Papers*, Seattle, WA, Jun. 2002, pp. 419–422.
- [33] F. Zhang and P. R. Kinget, "Design of components and circuits underneath integrated inductors," *IEEE J. Solid-State Circuits*, vol. 41, no. 10, pp. 2265–2271, Oct. 2006.
- [34] L. Nan, K. Mouthaan, Y.-Z. Xiong, J. Shi, S. C. Rustagi, and B.-L. Ooi, "Experimental characterization of the effect of metal dummy fills on spiral inductors," in *IEEE Radio Frequency Integrated Circuits Symp. Dig. Papers*, Honolulu, HI, Jun. 2007, pp. 307–310.
- [35] A. Batra, J. Balakrishnan, G. R. Aiello, J. R. Foerster, and A. Dabak, "Design of a multiband OFDM system for realistic UWB channel environments," *IEEE Trans. Microw. Theory Tech.*, vol. 52, no. 9, pp. 2123–2138, Sep. 2004.
- [36] R. Roovers, D. M. W. Leenaerts, J. Bergervoet, K. S. Harish, R. C. H. van de Beek, G. van der Weide, H. Waite, Y. Zhang, S. Aggarwal, and C. Razzell, "An interferencerobust receiver for ultra-wideband radio in SiGe BiCMOS technology," *IEEE J. Solid-State Circuits*, vol. 40, no. 12, pp. 2563–2572, Dec. 2005.

Chapter 5

Conclusion

In this dissertation, I have proposed and demonstrated low-voltage and small-area CMOS LNAs for narrowband and wideband applications.

Chapter 2 has demonstrated a 1.0 V, 0.25 mm^2 two-stage CMOS LNA with inductive source degeneration for 5 GHz applications. The presented two-stage topology that consists of common-source and common-gate stages is more suitable for low-voltage operation than a conventional cascode topology. The complete analytical expressions of the LNA performance were derived from the small-signal equivalent circuits that include an input parasitic capacitance and the Miller effect due to the gate-drain capacitance of the common-source transistor. These derived expressions showed that a higher V_{od1} results in a lower NF and IIP_3 while a higher V_{od2} leads to a higher NF and IIP_3 . The proposed design methodology based on these expressions allows us to efficiently design a two-stage LNA that satisfies desired gain, NF, and IIP_3 . The measurements were consistent with the calculations obtained from the analytical expressions.

Chapter 3 has demonstrated a 0.5 V, 0.21 mm² transformer folded-cascode CMOS LNA for 5 GHz applications. The internal and load inductors in a conventional folded-cascode LNA were magnetically coupled to reduced the chip area. The effects of the magnetic coupling between these inductors were analyzed. More magnetic coupling leads to a decrease in the resonance frequency of the input matching network, the peak frequency and magnitude of the gain, and the noise figure. The proposed partially-coupled transformer reduces the chip area, while having a small effect on the LNA performance. The folded-cascode LNA employing the transformer fabricated in a 90 nm CMOS process achieved the performance comparable to the conventional folded-cascode LNA, while consuming three fourths of the chip area of the conventional LNA. The fabricated LNA also achieved the best FoM with the smallest chip area among previously reported 0.4–0.6 V, 1.0 mW, 5 GHz CMOS LNAs. It has been demonstrated that the transformer folded-cascode LNA can replace conventional low-voltage CMOS LNAs.

Chapter 4 has demonstrated a 1.0 V, 0.12 mm² transformer noise-canceling CMOS LNA for fullband UWB (3.1–10.6 GHz) applications. The transformer noise cancellation scheme and output series inductor was incorporated into a conventional common-gate LNA. The transformer consisting of the input and shut-peaking inductors partly cancels the noise of the common-gate transistor and load resistor, thereby improving the LNA noise performance without increased
power consumption and chip area. The output series inductor improves both the gain and input bandwidths. Circuit analysis showed that the best turns ratio for the noise performance is one and input impedance matching depends not only on the common-gate transistor but also on the load resistor. The LNA fabricated in a 90 nm digital CMOS process achieved an S_{11} of less than -10 dB, NF of less than 4.4 dB, and S_{21} of more than 9.3 dB with the smallest supply voltage, power consumption (2.5 mW), and chip area, among previously reported 3.1–10.6 GHz CMOS LNAs. The proposed circuit topology is the most suitable for low-voltage, low-cost, and low-power UWB CMOS LNAs.

In summary, two circuit techniques allow the low-voltage and small-area design and implementation of CMOS LNAs. The first one is to use no cascode transistor, which alleviates the Miller effect but consumes voltage headroom. As alternative approaches, a common-gate stage is connected to a common-source LNA with inductive source degeneration as shown in Chapters 2 and 3; a common-gate topology, which needs no cascode transistor, is adopted for wideband applications as shown in Chapter 4. The second one is to magnetically couple two inductors to form a transformer, which consumes smaller chip area than two inductors, resulting in smaller area LNAs. In addition, the transformer consisting of the inductors connected to the source and drain terminals of the transistor in a common-gate topology partly cancels the noise produced by the transistor, improving the noise performance of the topology. This transformer noise cancellation scheme can be applied to all LNAs based on the common-gate topology. I conclude that the presented circuit techniques and LNAs contribute to low-voltage and low-cost CMOS RF front-ends.

Appendix A

NF Derivations

A.1 Two-Stage LNA

The noise of the two-stage LNA originates from M_1 , M_2 , R_I , and R_L . All output noise currents due to these elements flow into the equivalent resistance of the load LC tank, R_L . The LNA noise factor is given by

$$F_{LNA} = \frac{\overline{|i_{o,s,eq}|^2 + |i_{o,M_1}|^2 + |i_{o,M_2}|^2 + |i_{o,R_I}|^2 + |i_{o,R_L}|^2}}{|i_{o,s,eq}|^2}$$

= 1 + F_{M1} + F_{M2} + F_{RI} + F_{RL}, (A.1)

where $i_{o,s,eq}$, i_{o,M_1} , i_{o,M_2} , i_{o,R_I} , and i_{o,R_L} are the output noise currents due to R_{eq} , M_1 , M_2 , R_I , and R_L , respectively.

The output noise current originating from the equivalent signal source R_{eq} , $i_{o,s,eq}$, can be derived from the noise equivalent circuit of the input stage, shown in Fig. A.1. The noise current produced by R_{eq} is given by

$$\overline{|i_{ns,eq}|^2} = \frac{4k_B T \Delta f}{R_{eq}},\tag{A.2}$$

where k_B is Boltzmann's constant, T the absolute temperature, and Δf the noise bandwidth. The transfer function from $i_{ns,eq}$ to $i_{o,s,eq}$ is derived from Fig. A.1:

$$H_{ns,eq}(j\omega_0) \approx \frac{g_{m1}R_{eq}}{j\omega_0\alpha_M C_{gs1}(R_{eq} + R_{in})(1 + 1/g_{m2}R_I)},$$
(A.3)

where $\omega_0 L_{eq} = -X_{in}$ is assumed and $r_{nqs,eff} \ll R_{eq}$. The output noise current originating form R_{eq} is therefore given by

$$\overline{|i_{o,s,eq}|^2} = |H_{ns,eq}(j\omega_0)|^2 \overline{|i_{ns,eq}|^2} = \frac{4k_B T R_{eq} \omega_{T1}^2 \Delta f}{\omega_0^2 \alpha_M^2 (R_{eq} + R_{in})^2 (1 + 1/g_{m2} R_I)^2}.$$
 (A.4)



Figure A.1: Noise equivalent circuit of the input stage

A.1.1 *F*_{*M*₁}

The main noise sources in a MOSFET are the drain noise current i_{nd} and induced-gate noise current i_{ng} , expressed as [1,2]

$$\overline{|i_{nd}|^2} = 4k_B T \gamma g_{d0} \Delta f, \tag{A.5}$$

$$\overline{|i_{ng}|^2} = 4k_B T \delta \frac{(\omega C_{gs})^2}{\kappa g_{d0}} \Delta f, \qquad (A.6)$$

respectively, where g_{d0} is the zero-bias drain conductance of a MOSFET; γ , δ , and κ represent the noise parameters. The induced-gate noise current correlates to the drain noise current, and the correlation coefficient is given by [1]

$$c = \frac{\overline{i_{ng} \cdot i_{nd}^*}}{\sqrt{|i_{ng}|^2} \sqrt{|i_{nd}|^2}}.$$
(A.7)

Using this coefficient, we can express the induced-gate noise as [3]

$$\frac{|i_{ng}|^2}{|i_{ngc}|^2} = \overline{|i_{ngc}|^2} + \overline{|i_{ngu}|^2} = \overline{|i_{ng}|^2}|c|^2 + \overline{|i_{ng}|^2}(1 - |c|^2),$$
(A.8)

where i_{ngc} and i_{ngu} are the correlated and uncorrelated components, respectively. The output noise current originating from M_1 is therefore expressed as

$$\begin{aligned} \overline{|i_{o,M_1}|^2} &= \overline{|i_{o,nd1} + i_{o,ng1}|^2} \\ &= \overline{|i_{o,nd1} + i_{o,ngc1}|^2} + \overline{|i_{o,ngu1}|^2} \\ &= \overline{|i_{o,nd1}|^2} + \overline{i_{o,ngc1} \cdot i_{o,nd1}^*} + \overline{i_{o,nd1} \cdot i_{o,ngc1}^*} + \overline{|i_{o,ngc1}|^2} + \overline{|i_{o,ngu1}|^2}, \end{aligned}$$
(A.9)

where $i_{o,nd1}$, $i_{o,ngc1}$ and $i_{o,ngu1}$ are the output noise currents due to i_{nd1} , i_{ngc1} , and i_{ngu1} , respectively. From Fig. A.1, the transfer function from i_{nd1} to $i_{o,nd1}$ is given by

$$H_{nd1}(j\omega_0) = \frac{(R_{eq} + j\omega_0 L_{eq})(1 + \alpha_{gd1}) + j\omega_0 L_s + \frac{1}{j\omega_0 C_{gs1}}}{R_{eq} + j\omega_0 L_{eq} + \left(\omega_{T1}L_s + j\omega_0 L_s + \frac{1}{j\omega_0 C_{gs1}}\right)/\alpha_M} \cdot \frac{1}{\alpha_M(1 + 1/g_{m2}R_I)} \approx \frac{R_{eq}(1 + \alpha_{gd1})}{\alpha_M(R_{eq} + R_{in})(1 + 1/g_{m2}R_I)},$$
(A.10)

where $r_{nqs,eff} \ll R_{eq}$ and $j\omega_0(L_{eq} + L_s/\alpha_M) + 1/j\omega_0\alpha_M C_{gs1} \simeq 0$ under input impedance matching condition. The transfer function from i_{ng1} to $i_{o,ng1}$ is also given by

$$H_{ng1}(j\omega_0) = -\frac{\frac{g_{m1}}{j\omega_0 C_{gs1}} \left(R_s + j\omega_0 L_{eq} + j\omega_0 L_s\right) - \alpha_{gd1} \left(R_s + j\omega_0 L_{eq}\right)}{R_s + j\omega_0 L_{eq} + \left(\omega_{T1}L_s + j\omega_0 L_s + \frac{1}{j\omega_0 C_{gs1}}\right)/\alpha_M} \cdot \frac{1}{\alpha_M (1 + 1/g_{m2}R_I)} \approx -\frac{g_{m1} \left(R_{eq} + j/\omega_0 C_{gs1}\right)}{j\omega_0 \alpha_M C_{gs1} \left(R_{eq} + R_{in}\right)(1 + 1/g_{m2}R_I)},$$
(A.11)

where $\alpha_M \ll g_{m1}/\omega_0 C_{gs1}$. Using Eqs. (A.5), (A.6), (A.10), and (A.11), we have

$$\overline{|i_{o,nd1}|^2} = |H_{nd1}(j\omega_0)|^2 \overline{|i_{nd1}|^2} = \frac{4k_B T \gamma_1 g_{m1} R_{eq}^2 (1 + \alpha_{gd1})^2 \Delta f}{\alpha_1 \alpha_M^2 (R_{eq} + R_{in})^2 (1 + 1/g_{m2} R_I)^2},$$
(A.12)

 $\overline{i_{o,ngc1} \cdot i_{o,nd1}^*} + \overline{i_{o,nd1} \cdot i_{o,ngc1}^*} = H_{ngc1}(j\omega_0)i_{ngc1} \cdot H_{nd1}^*(j\omega_0)i_{nd1}^* + H_{nd1}(j\omega_0)i_{nd1} \cdot H_{ngc1}^*(j\omega_0)i_{ngc1}^*$

$$= -2|c| \sqrt{\frac{\delta_1}{\kappa_1 \gamma_1}} \frac{4k_B T \gamma_1 g_{m1} R_{eq}^2 \Delta f}{\alpha_M (R_{eq} + R_{in})^2 (1 + 1/g_{m2} R_I)^2},$$
(A.13)

$$\overline{|i_{o,ngc1}|^2 + |i_{o,ngu1}|^2} = |H_{ng1}(j\omega_0)|^2 \overline{|i_{ng1}|^2} = \frac{4k_B T \alpha_1 \delta_1 g_{m1} (R_{eq}^2 + 1/\omega_0^2 C_{gs1}^2) \Delta f}{\kappa_1 \alpha_M^2 (R_{eq} + R_{in})^2 (1 + 1/g_{m2} R_I)^2}.$$
(A.14)

Substituting Eqs. (A.12)–(A.14) into Eq. (A.9) gives the output noise current originating from M_1 as

$$\overline{|i_{o,M_1}|^2} = \frac{4k_B T R_{eq} \Delta f}{\alpha_M^2 (R_{eq} + R_{in})^2 (1 + 1/g_{m2} R_I)^2} \left(\frac{\gamma_1 \chi_1}{\alpha_1} g_{m1} R_{eq} + \frac{\alpha_1 \delta_1 g_{m1}}{\kappa_1 \omega_0^2 C_{gs1}^2 R_{eq}}\right),$$
(A.15)

where χ_1 is given by Eq. (2.40). Dividing Eq. (A.15) by Eq. (A.4), we obtain F_{M_1} (Eq. (2.36)).

A.1.2 F_{M_2}

The noise contribution from M_2 , F_{M_2} , can be derived in the same way as the derivation of F_{M_1} . The noise equivalent circuit of the common-gate stage, shown in Fig. A.2, gives the transfer



Figure A.2: Noise equivalent circuit of the common-gate stage

functions from the drain and induced-gate noise currents to the output currents:

$$H_{nd2}(j\omega_0) = \frac{1}{1 + g_{m2}R_I},\tag{A.16}$$

$$H_{ng2}(j\omega_0) = \frac{1}{1 + 1/g_{m2}R_I},\tag{A.17}$$

respectively. Using Eqs. (A.5), (A.6), (A.16), and (A.17), we have

$$\overline{|i_{o,nd2}|^2} = \frac{4k_B T \gamma_2 g_{m2}}{\alpha_2 (1 + g_{m2} R_I)^2},$$
(A.18)

$$\overline{i_{o,ngc2} \cdot i_{o,nd1}^*} + \overline{i_{o,nd2} \cdot i_{o,ngc2}^*} = 0, \tag{A.19}$$

$$\overline{|i_{o,ngc2}|^2} + \overline{|i_{o,ngu2}|^2} = \frac{4k_B T \delta_2 \alpha_2 (\omega_0 C_{gs2})^2}{\kappa_2 g_{m2} (1 + 1/g_{m2} R_I)^2},$$
(A.20)

where $i_{o,nd2}$, $i_{o,ngc2}$ and $i_{o,ngu2}$ are the output noise currents due to i_{nd2} , i_{ngc2} , and i_{ngu2} , respectively. Using Eqs. (A.18)–(A.20), we obtain

$$\overline{|i_{o,M_2}|^2} = \overline{|i_{o,nd_2}|^2} + \overline{i_{o,ngc2} \cdot i_{o,nd_2}^*} + \overline{i_{o,nd2} \cdot i_{o,ngc2}^*} + \overline{|i_{o,ngc2}|^2} + \overline{|i_{o,ngc2}|^2} + \overline{|i_{o,ngu2}|^2} = 4k_B T \frac{\gamma_2}{\alpha_2} \chi_2 \frac{g_{m2}}{(1+1/g_{m2}R_I)^2} \Delta f,$$
(A.21)

where χ_2 is given by Eq. (2.41). Dividing Eq. (A.21) by Eq. (A.4) gives F_{M_2} (Eq. (2.37)).

A.1.3 F_{R_I} and F_{R_L}

The noise currents due to the internal and load LC tanks are given by

$$\overline{|i_{nR_I}|^2} = \frac{4k_B T \Delta f}{R_I},\tag{A.22}$$

$$\overline{|i_{nR_L}|^2} = \frac{4k_B T \Delta f}{R_L},\tag{A.23}$$

respectively. From Fig. A.2, the output noise currents originating from the internal and load LC tanks are derived as

$$\overline{|i_{o,R_I}|^2} = \frac{4k_B T \Delta f}{R_I (1 + 1/g_{m2} R_I)^2},$$
(A.24)

$$\overline{|i_{o,R_L}|^2} = \overline{|i_{nR_L}|^2} = \frac{4k_B T \Delta f}{R_L},\tag{A.25}$$

respectively. Dividing Eqs. (A.24) and (A.25) by Eq. (A.4) gives F_{R_I} (Eq. (2.38)) and F_{R_L} (Eq. 2.39)), respectively.

A.2 Transformer Folded-Cascode LNA

The noise of M_1 , M_2 , R_I , and R_L contribute to the overall LNA noise. The LNA noise factor is given by

$$F = \frac{\overline{|v_{o,R_s}|^2} + \overline{|v_{o,M_1}|^2} + \overline{|v_{o,M_2}|^2} + \overline{|v_{o,R_I}|^2} + \overline{|v_{o,R_L}|^2}}{\overline{|v_{o,R_s}|^2}},$$

= 1 + F_{M1} + F_{M2} + F_{LI} + F_{LL}, (A.26)

where v_{o,R_s} , v_{o,M_1} , v_{o,M_2} , v_{o,R_I} , and v_{o,R_L} are the output noise voltages originating from R_s , M_1 , M_2 , R_I , and R_L , respectively.

The output noise voltage originating from R_s , v_{o,R_s} , can be derived from the noise equivalent circuit of the input stage, shown in Fig. A.3. The signal source noise current is expressed as

$$\overline{|i_{ns}|^2} = \frac{4k_B T \Delta f}{R_s},\tag{A.27}$$

where k_B is Boltzmann's constant, T the absolute temperature, and Δf the noise bandwidth. The transfer function from i_{ns} to the noise current at node I, $i_{I,Rs}$, is given by

$$H_{ns}(j\omega_0) = \frac{g_{m1}R_s}{j\omega_0\alpha_M C_{gs1} \left(R_s + \omega_T L_s/\alpha_M\right)},\tag{A.28}$$

where $\text{Im}[Z_{in}(j\omega_0)]=0$ and α_M is approximated as $\text{Re}[\alpha_M]$ for input impedance matching as shown in Section 3.1. Using Eq. (A.28), we have

$$\overline{|i_{I,Rs}|^2} = |H_{ns}(j\omega_0)|^2 \overline{|i_{ns}|^2} = \frac{4k_B T R_s \omega_{T1}^2 \Delta f}{\omega_0^2 |\alpha_M|^2 (R_s + \omega_T L_s / |\alpha_M|)^2}.$$
(A.29)

The common-gate stage converts $i_{I,Rs}$ to the output voltage:

$$\overline{|v_{o,Rs}|^2} = |Z_T|^2 \,\overline{|i_{I,Rs}|^2},\tag{A.30}$$

where Z_T is the transimpedance of the common-gate stage and is given by Eq. (3.14).



Figure A.3: Noise equivalent circuit of the input stage.

A.2.1 F_{M_1}

The noise current of M_1 is also converted by the common-gate stage. The noise contribution from M_1 can be expressed as

$$F_{M_1} = \frac{\overline{|v_{o,M_1}|^2}}{\overline{|v_{o,R_s}|^2}} = \frac{|Z_T|^2 \overline{|i_{I,M_1}|^2}}{|Z_T|^2 \overline{|i_{I,R_s}|^2}} = \frac{\overline{|i_{I,M_1}|^2}}{\overline{|i_{I,R_s}|^2}},$$
(A.31)

where i_{I,M_1} is the noise current at node I as shown in Fig. A.3.

The main noise sources in a MOSFET are the drain noise current i_{nd} and induced-gate noise current i_{ng} , expressed as Eqs. (A.5) and (A.6), respectively. Considering the correlation between i_{nd} and i_{ng} , we can express the noise current due to M_1 at node I as

$$\overline{|i_{I,M_{1}}|^{2}} = \overline{|i_{I,nd1} + i_{I,ng1}|^{2}}
= \overline{|i_{I,nd1} + i_{I,ngc1}|^{2}} + \overline{|i_{I,ngu1}|^{2}}
= \overline{|i_{I,nd1}|^{2}} + \overline{i_{I,ngc1} \cdot i_{I,nd1}^{*}} + \overline{i_{I,nd1} \cdot i_{I,ngc1}^{*}} + \overline{|i_{I,ngc1}|^{2}} + \overline{|i_{I,ngu1}|^{2}},$$
(A.32)

where $i_{I,nd1}$, $i_{I,ngc1}$, and $i_{I,ngu1}$ are the noise currents due to i_{nd1} , i_{ngc1} , and i_{ngu1} at node I, respectively. From Fig. A.3, the transfer function from i_{nd1} to $i_{I,nd1}$ is approximated as

$$H_{nd1}(j\omega_0) = \frac{(R_s + j\omega_0 L_g) (1 + \alpha_{gd1}) + j\omega_0 L_s + \frac{1}{j\omega_0 C_{gs1}}}{R_s + j\omega_0 L_g + \left(\omega_{T1}L_s + j\omega_0 L_s + \frac{1}{j\omega_0 C_{gs1}}\right) / \alpha_M} \cdot \frac{1}{\alpha_M} \approx \frac{R_s (1 + \alpha_{gd1})}{\alpha_M (R_s + \omega_{T1}L_s / \alpha_M)},$$
(A.33)

where $j\omega_0(L_g + L_s/\alpha_M) + 1/j\omega_0\alpha_M C_{gs1} \simeq 0$ under input impedance matching condition. The

transfer function from i_{ng1} to $i_{I,ng1}$ is also approximated as

$$H_{ng1}(j\omega_0) = -\frac{\frac{g_{m1}}{j\omega_0 C_{gs1}} \left(R_s + j\omega_0 L_g + j\omega_0 L_s\right) - \alpha_{gd1} \left(R_s + j\omega_0 L_g\right)}{R_s + j\omega_0 L_g + \left(\omega_{T1} L_s + j\omega_0 L_s + \frac{1}{j\omega_0 C_{gs1}}\right) / \alpha_M} \cdot \frac{1}{\alpha_M}$$

$$\approx -\frac{g_{m1} \left(R_s + j\frac{1}{\omega_0 C_{gs1}}\right)}{j\omega_0 \alpha_M C_{gs1} \left(R_s + \omega_{T1} L_s / \alpha_M\right)},$$
(A.34)

where $\alpha_M \ll g_{m1}/\omega_0 C_{gs1}$ and $j\omega_0 (L_g + L_s/\alpha_M) + 1/j\omega_0 \alpha_M C_{gs1} \simeq 0$. Using Eqs. (A.5), (A.6), (A.33), and (A.34), we have

$$|i_{I,nd1}|^{2} = |H_{nd1}(j\omega_{0})|^{2} |i_{nd1}|^{2}$$

= $\frac{4k_{B}T\gamma_{1}g_{m1}R_{s}^{2}(1+\alpha_{gd1})^{2}\Delta f}{\alpha_{1}|\alpha_{M}|^{2}(R_{s}+\omega_{T1}L_{s}/|\alpha_{M}|)^{2}},$ (A.35)

$$\overline{i_{I,ngc1} \cdot i_{I,nd1}^*} + \overline{i_{I,nd1} \cdot i_{I,ngc1}^*} = H_{ngc1}(j\omega_0)i_{ngc1} \cdot H_{nd1}^*(j\omega_0)i_{nd1}^* + H_{nd1}(j\omega_0)i_{nd1} \cdot H_{ngc1}^*(j\omega_0)i_{ngc1}^*$$

$$= -2|c|\sqrt{\frac{\delta_1}{\kappa_1\gamma_1}}\frac{4k_B T \gamma_1 g_{m1} R_s^2 \left(1 + \alpha_{gd1}\right) \Delta f}{|\alpha_M|^2 (R_s + \omega_{T1} L_s / |\alpha_M|)^2},$$
(A.36)

$$\frac{|i_{I,ngc1}|^2}{|i_{I,ngu1}|^2} = |H_{ng1}(j\omega_0)|^2 \overline{|i_{ng1}|^2} = \frac{4k_B T \alpha_1 \delta_1 g_{m1} \left(R_s^2 + \frac{1}{\omega_0^2 C_{gs1}^2}\right) \Delta f}{\kappa_1 |\alpha_M|^2 (R_s + \omega_{T1} L_s / |\alpha_M|)^2}.$$
(A.37)

Substituting Eqs. (A.35)–(A.37) into Eq. (A.32) gives the noise current of M_1 at node I:

$$\overline{|i_{I,M_1}|^2} = \frac{4k_B T R_s \Delta f}{|\alpha_M|^2 (R_s + \omega_{T1} L_s / |\alpha_M|)^2} \left(\frac{\gamma_1 \chi_1}{\alpha_1} g_{m1} R_s + \frac{\alpha_1 \delta_1 g_{m1}}{\kappa_1 \omega_0^2 C_{gs1}^2 R_s}\right),$$
(A.38)

where χ_1 is given by Eq. (3.22). Substituting Eqs. (A.29) and (A.38) into Eq. (A.31) gives F_{M_1} (Eq. (3.21)).

A.2.2 F_{M_2}

In the common-gate topology, the gate-induced noise current of the MOSFET can be ignored against the drain noise current:

$$\overline{|v_{o,M_2}|^2} \approx \overline{|v_{o,nd2}|^2},\tag{A.39}$$

where $v_{o,nd2}$ is the output voltage originating from the drain noise current of M_2 , i_{nd2} , and is derived from the noise equivalent circuit of the common-gate stage, shown in Fig. A.4:

$$\overline{|v_{o,nd2}|^2} = \left|\frac{N_{nd2}}{Y_0 + Y_I}\right|^2 \overline{|i_{nd2}|^2} = \left|\frac{N_{nd2}}{Y_0 + Y_I}\right|^2 \cdot 4k_B T \frac{\gamma_2}{\alpha_2} g_{m2} \Delta f,$$
(A.40)



Figure A.4: Noise equivalent circuit of the common-gate stage.

$$N_{nd2} = -\left(j\omega L_L + R_L - \frac{(j\omega M)^2}{j\omega L_I + R_I}\right)(Y_0 + Y_{L_IC_I}) + \frac{j\omega M}{j\omega L_I + R_I}\left(1 - \frac{j\omega M}{j\omega L_I + R_I}\right) \approx -j\omega n^2(1 - k^2)L_I \cdot (Y_0 + Y_{L_IC_I}) + nk(1 - nk),$$
(A.41)

where R_I and R_L are ignored for simplicity; Y_0 represents the output admittance of the input stage at node I; $Y_{L_IC_I} = j\omega C_I + 1/(j\omega L_I + R_I)$, as shown in Section 3.3. Rewriting Eq. (A.30) in terms of $Y_0 + Y_I$, we have

$$\overline{|v_{o,Rs}|^{2}} = \left|\frac{N_{ns}}{Y_{0} + Y_{I}}\right|^{2} \left|\frac{Y_{0} + Y_{I}}{Y_{I}}\right|^{2} \overline{|i_{I,Rs}|^{2}},$$

$$N_{ns} = \frac{j\omega M}{j\omega L_{I} + R_{I}} + g_{m2} \left(j\omega L_{L} + R_{L} - \frac{(j\omega M)^{2}}{j\omega L_{I} + R_{I}}\right)$$

$$\approx nk + g_{m2} \cdot j\omega n^{2}(1 - k^{2})L_{I},$$
(A.42)
(A.43)

where R_I and R_L are ignored for simplicity. Dividing Eq. (A.40) by Eq. (A.42) with $L_I = 1/\omega_0^2(1+k)C_L$ and n = 1, we obtain F_{M_2} (Eq. (3.23)).

A.2.3 F_{L_I} and F_{L_L}

The noise voltages of the parasitic resistances of L_I and L_L are given by

$$\overline{|v_{nR_I}|^2} = 4k_B T R_I \Delta f, \tag{A.44}$$

$$\overline{|v_{nR_L}|^2} = 4k_B T R_L \Delta f, \tag{A.45}$$

respectively. The output noise voltages due to v_{nR_I} and v_{nR_L} can be expressed from Fig. A.4:

$$\overline{|v_{o,R_L}|^2} = \left|\frac{N_{nR_L}}{Y_0 + Y_I}\right|^2 \overline{|v_{nR_L}|^2},\tag{A.46}$$

$$N_{nR_{L}} = g_{m2} \left(1 - \frac{j\omega M}{j\omega L_{I} + R_{I}} \right) + Y_{0} + Y_{L_{I}C_{I}}$$

$$\approx g_{m2}(1 - nk) + Y_{0} + Y_{L_{I}C_{I}}$$
(A.47)

and

$$\overline{|v_{o,R_I}|^2} = \left|\frac{N_{nR_I}}{Y_0 + Y_I}\right|^2 \overline{|v_{nR_I}|^2}$$

$$N_{nR_I} = g_{m2} \left(\frac{j\omega L_L + R_L}{j\omega L_I + R_I} - \frac{j\omega M}{j\omega L_I + R_I}\right) - \frac{j\omega M}{j\omega L_I + R_I} \left(Y_0 + j\omega C_I\right)$$

$$\approx n(n-k)g_{m2} - nk \left(Y_0 + j\omega C_I\right),$$
(A.48)
(A.49)

respectively, where R_I and R_L are ignored for simplicity. Dividing Eqs. (A.46) and (A.48) by Eq. (A.42) with $L_I = 1/\omega_0^2(1+k)C_L$ and n = 1, we derive F_{L_I} (Eq. (3.24)) and F_{L_L} (Eq. (3.25)), respectively.

Bibliography

- [1] A. van der Ziel, *Noise in Solid State Devices and Circuits*. Toronto: John Wiley & Sons, 1986.
- [2] D. K. Shaeffer and T. H. Lee, "A 1.5-V, 1.5-GHz CMOS low noise amplifier," *IEEE J. Solid-State Circuits*, vol. 32, no. 5, pp. 745–759, May 1997.
- [3] T. H. Lee, *The Design of CMOS Radio-Frequency Integrated Circuits*, 2nd ed. Cambridge: Cambridge University Press, 2004.

Appendix B

Frequency Responses of Y_I and $1/Y_I$

B.1 Y_I

The frequency response of Y_I of the transformer folded-cascode CMOS LNA is shown in Fig. B.1(a). At low frequencies $(1 \gg -\omega^2(1 - k^2)L_LC_L + j\omega R_LC_L)$, Eq. (3.3) can be approximated as

$$\operatorname{Re}[Y_I] \approx g_{m2}(1 - nk),\tag{B.1}$$

$$\operatorname{Im}[Y_I] \approx \omega \left(C_I + n^2 k^2 C_L \right) - \frac{1}{\omega L_I},\tag{B.2}$$

which shows that $Im[Y_I]$ becomes zero around

$$w_{0,Y_{I}} = \frac{1}{\sqrt{L_{I}\left(C_{I} + n^{2}k^{2}C_{L}\right)}}.$$
(B.3)

At $\omega_{1,Y_{I}} = 1/\sqrt{(1-k^2)L_LC_L}$, Re[Y_I] exceeds g_{m2} and Im[Y_I] becomes a maximum:

$$\operatorname{Re}[Y_I] \approx g_{m2} + \frac{n^2 k^2}{R_L},\tag{B.4}$$

$$\operatorname{Im}[Y_I] \approx \omega_{1,Y_I} C_I - \frac{1}{\omega_{1,Y_I} L_I} + \frac{nkg_{m2}}{\omega_{1,Y_I} R_L C_L},\tag{B.5}$$

respectively, where $1 - \omega_{1,Y_{I}}^{2}(1-k^{2})L_{L}C_{L} \simeq 0$. Above $\omega_{1,Y_{I}}$, Re[Y_I] and Im[Y_I] approach gradually g_{m2} and $\omega C_{I} - 1/\omega L_{I}$, respectively.

B.2 $1/Y_I$

The reverse of Y_I is expressed as

$$\frac{1}{Y_I} = \frac{sC_L + \frac{1}{s(1-k^2)L_L + R_L}}{D},$$
(B.6)



Figure B.1: Calculated (a) Y_I and (b) $1/Y_I$ with k as a parameter.

where D is given by Eq. (3.15). Figure B.1(b) shows the calculated frequency response of $1/Y_I$ with k as a parameter. At low frequencies ($\omega \simeq 0$), $1/Y_I$ is approximated as

$$\operatorname{Re}\left[\frac{1}{Y_{I}}\right] \approx \frac{R_{I}}{1 + R_{I}g_{m2}(1 - nk)},\tag{B.7}$$

$$\operatorname{Im}\left[\frac{1}{Y_I}\right] \approx 0. \tag{B.8}$$

Equation (3.15) indicates that $\text{Re}[1/Y_I]$ and $\text{Im}[1/Y_I]$ have peaks around ω_p , given by Eq. (3.18). Above ω_p , $\text{Re}[1/Y_I]$ and $\text{Im}[1/Y_I]$ approach gradually $1/g_{m2}$ and $1/(\omega C_I - 1/\omega L_I)$, respectively.

Achievements

Papers

- 1. <u>木原 崇雄</u>, 金 奎哲, 清水 由幸, 村上 豊生, 上田 啓介, 後藤 克, 松岡 俊匡, 谷口 研二, "雑 音と線形性を考慮した CMOS 低雑音増幅器の設計手法に関する検討," 電子情報通信 学会論文誌 C, vol. J89-C, no. 2, pp. 72–75, February 2006.
- Guechol Kim, Yoshiyuki Shimizu, Bunsei Murakami, Masaru Goto, Keisuke Ueda, <u>Takao Kihara</u>, Toshimasa Matsuoka, and Kenji Taniguchi, "Accurate Small-Signal Modeling of FD-SOI MOSFETs," *IEICE Trans. Electron*, vol. E89-C, no. 4, pp. 517–519, April 2006.
- Guechol Kim, Bunsei Murakami, Masaru Goto, <u>Takao Kihara</u>, Keiji Nakamura, Yoshiyuki Shimizu, Toshimasa Matsuoka, and Kenji Taniguchi, "Small-Signal and Noise Model of FD-SOI MOS Devices for Low Noise Amplifier," *Jpn. J. Appl. Phys.*, *PT.1*, vol. 45, no. 9A, pp. 6872–6877, September 2006.
- <u>Takao Kihara</u>, Guechol Kim, Masaru Goto, Keiji Nakamura, Yoshiyuki Shimizu, Toshimasa Matsuoka, and Kenji Taniguchi "Analytical Expression Based Design of a Low-Voltage FD-SOI CMOS Low-Noise Amplifier," *IEICE Trans. Fundamentals of Electronics, Communications and Computer Sciences*, vol. E90-A, no. 2, pp. 317–325, February 2007.
- 5. <u>Takao Kihara</u>, Hae-Ju Park, Isao Takobe, Fumiaki Yamashita, Toshimasa Matsuoka, and Kenji Taniguchi, "A 0.5 V Area-Efficient Transformer Folded-Cascode CMOS Low-Noise Amplifier," *IEICE Trans. Electronics*, vol. E92-C, no. 4, April 2009, to be published.
- 6. <u>Takao Kihara</u>, Toshimasa Matsuoka, and Kenji Taniguchi, "A Transformer Noise-Canceling Ultra-Wideband CMOS Low-Noise Amplifier," submitted to *IEEE Trans. Microw. Theory Tech*.

International Conferences

1. Guechol Kim, Bunsei Murakami, Yoshiyuki Shimizu, Masaru Goto, Sungwoo Cha, Keisuke Ueda, <u>Takao Kihara</u>, Yoshihiro Utsurogi, Toshimasa Matsuoka, and Kenji Taniguchi, "Sma-

ll-Signal RF Modeling of FD-SOI MOSFETs Including NQS Effect," in *Proc. IEEE Int. Meeting for Future of Electron Devices, Kansai (IMFEDK)*, Kyoto, Japan, April 2005, pp. 61–62.

- Guechol Kim, Bunsei Murakami, Masaru Goto, <u>Takao Kihara</u>, Keiji Nakamura, Yoshiyuki Shimizu, Toshimasa Matsuoka, and Kenji Taniguchi, "A Simple Noise Model of FD-SOI MOS Devices for Low Noise Amplifier," in *Proc. IEEE Int. Meeting for Future of Electron Devices, Kansai (IMFEDK)*, Kyoto, Japan, April 2006, pp. 59–60.
- 3. <u>Takao Kihara</u>, Hae-Ju Park, Isao Takobe, Fumiaki Yamashita, Toshimasa Matsuoka, and Kenji Taniguchi, "A 0.5 V CMOS Low-Noise Amplifier," in *Proc. IEEE Int. Meeting for Future of Electron Devices, Kansai (IMFEDK)*, Osaka, Japan, May 2008, pp. 91–92.
- <u>Takao Kihara</u>, Hae-Ju Park, Isao Takobe, Fumiaki Yamashita, Toshimasa Matsuoka, and Kenji Taniguchi, "A 0.5 V Area-Efficient Transformer Folded-Cascode Low-Noise Amplifier in 90 nm CMOS," in *Proc. IEEE Int. Conf. on Integrated Circuit Design and Technology (ICICDT)*, Grenoble, France, June 2008, pp. 21–24.
- <u>Takao Kihara</u>, Toshimasa Matsuoka, and Kenji Taniguchi, "A 1.0 V, 2.5 mW, Transformer Noise-Canceling UWB CMOS LNA," in *IEEE Radio Frequency Integrated Circuits (RFIC) Symp. Dig. Papers*, Atlanta, GA, June 2008, pp. 493–496.

Domestic Conferences

- 木原 崇雄, 金 奎哲, 清水 由幸, 村上 豊生, 後藤 克, 洞木 吉博, 上田 啓介, 車 承佑, 松岡 俊匡, 谷口 研二, "CMOS 低雑音増幅器の設計手法に関する検討," 電子情報通信学会 シリコンアナログ RF 研究会 (2005 年 5 月).
- 2. 後藤 克, 金 奎哲, 清水 由幸, 村上 豊生, 上田 啓介, 木原 崇雄, 車 承佑, 松岡 俊匡, 谷口 研二, "高周波における MOSFET の雑音特性のモデリング," 電子情報通信学会 シリコ ンアナログ RF 研究会 (2005 年 5 月).
- 村上 豊生, 金 奎哲, 木原 崇雄, 後藤 克, 松岡 俊匡, 谷口 研二, "FD-SOI MOSFET の高 周波小信号特性の高精度モデリング," 電子情報通信学会 シリコン材料・デバイス研 究会 (2005 年 12 月).
- 中村 圭志,金 奎哲,木原 崇雄,村上 豊生,後藤 克,上田 啓介,松岡 俊匡,谷口 研二, "FD-SOI MOSFET の小信号マクロモデルを用いた RF 低雑音増幅器の設計,"電子情 報通信学会 シリコンアナログ RF 研究会 (2006 年 2 月).
- 5. 木原 崇雄, 金 奎哲, 後藤 克, 中村 圭志, 清水 由幸, 松岡 俊匡, 谷口 研二, "FD-SOI CMOS デバイスを用いた低電圧動作低雑音増幅器の設計," 電気学会 電子回路研究会, ECT-06-54, pp. 11–16 (2006 年 6 月).

- 6. 後藤 克, 木原 崇雄, 金 奎哲, 松岡 俊匡, 谷口 研二, "可変バイアス回路を用いた擬似差動 LNA の検討," 電子情報通信学会 2006 年ソサイエティ大会, C-12-24 (2006 年 9 月).
- 7. 陳 炳宏,<u>木原 崇雄</u>,松岡 俊匡,谷口 研二,"CMOS 電圧制御発振器の位相雑音の解 析モデルの検討,"電子情報通信学会 シリコンアナログ RF 研究会 (2006 年 11 月).
- 8. <u>木原 崇雄</u>, 後藤 克, 松岡 俊匡, 谷口 研二, "トランスを用いた CMOS UWB LNA の設計," 電子情報通信学会 シリコンアナログ RF 研究会 (2007 年 3 月).
- 10. 田古部 勲, 木原 崇雄, 松岡 俊匡, 谷口 研二, "トランスを用いた雑音除去 UWB 差動 CMOS LNA," 電子情報通信学会 2008 年ソサイエティ大会, C-12-43 (2008 年 9 月).
- 11. 山下 文明, 木原 崇雄, 田古部 勲, 朴 海柱, 松岡 俊匡, 谷口 研二, "0.5 V 動作バイアス 制御 CMOS 型 VCO," 電子情報通信学会 2008 年ソサイエティ大会, C-12-53 (2008 年 9月).

Awards

- 平成18年度電気学会優秀論文発表賞, "FD-SOI CMOS デバイスを用いた低電圧動作低雑音増幅器の設計" (2007年1月).
- 2. Global COE EDIS2008 student paper award, "Transformer Noise-Canceling UWB CMOS LNA," Osaka, Japan, January 2008.
- 3. 平成 19 年度 JSPS「魅力ある大学院教育」イニシアティブ「先端通信エキスパート養成プログラム」個人提案型研究優秀研究賞, "トランス雑音除去 CMOS UWB 低雑音 増幅器" (2008 年 3 月).
- 第 10 回 LSI IP アワード 研究助成賞, "トランス雑音除去 UWB CMOS 低雑音増幅器" (2008 年 4 月).
- 5. IEEE ICICDT2008 student paper award (3rd prize), "A 0.5 V Area-Efficient Transformer Folded-Cascode Low-Noise Amplifier in 90 nm CMOS," Grenoble, France, June 2008.